The benefits of using silicon nutrient on strawberries including reducing epidemics of strawberry powdery mildew (*Podosphaera aphanis*)

Carmilla Ifeoma Asiana

Department of Biological and Environmental Sciences, School of Life and Medical Sciences, University of Hertfordshire,

Hatfield, AL10 9AB, UK

A thesis submitted to the University of Hertfordshire in partial fulfilment of the requirements for the degree of Doctor of Philosophy

October 2021

Abstract

Powdery mildew of strawberry plants caused by the fungus *Podosphaera aphanis* is a significant fungal disease of protected strawberry crops in the UK, causing yield losses between 20 to 70% of crop potential. At 20% losses, this can contribute to an industry volume of 23,100 tonnes, estimated at a market value of £56.8 million. Although growers frequently limit the spread of strawberry powdery mildew by a weekly to fortnightly application of fungicides (April to October), it has become more prominent in recent years. This research aimed to investigate four key areas. Firstly, the effects of the silicon delivered through a fertigation system on the development of strawberry powdery mildew (*Podosphaera aphanis*) disease levels in different strawberry plant cultivars. Secondly, examine the amounts and pattern of distribution of silicon in leaves, leaf petioles and roots of strawberry plants growing in glasshouse and field experiments. Thirdly, evaluate °Brix levels of silicon-treated strawberry plant fruits and leaf petioles with the untreated control, and lastly, measure growth parameters of strawberry plants in the absence and presence of silicon in a glasshouse hydroponic experiment.

In this thesis, silicon fertigation field experiments were set up on a commercial strawberry farm to evaluate the effects of silicon in reducing levels of disease of *Podosphaera aphanis* throughout the growing season. Results from this study (chapter three) revealed that the application of silicon as a nutrient reduced levels of strawberry powdery mildew in 2016. The lowest disease levels (P<0.05) occurred in Malling Centenary strawberry crops that received silicon twice-a-week with fungicides (AUDPC, 410) and without fungicides (AUDPC, 375) compared with the untreated control (AUDPC, 3423). Results from this experiment showed that the addition of silicon delayed the rise in disease levels by 29 days in the silicon twice-a-week treatment with and without fungicides compared with the untreated control. Disease level assessments carried out in 2017 and 2018 field experiments using the cultivar Amesti showed low levels of disease were only found in the untreated control plot compared to all other treatments in 2016.

A silicon deposition experiment was conducted on strawberry plants in a glasshouse (chapter four) in 2017. The results revealed that high amounts of silicon were deposited in the upper and lower cuticle, epidermis, palisade layer, and vein of the leaf (fluorescence intensity, 7.2cps) of silicon-treated plants compared with the untreated control (fluorescence intensity, 2.2cps) (P<0.05). In the leaf petiole, more silicon was found in the upper and lower cuticle, epidermis and xylem (fluorescence intensity, 7.7cps) of silicon-treated plants compared with the untreated control (fluorescence intensity, 1.9cps) (P<0.05). In the roots, more silicon was found deposited mainly in the xylem (fluorescence intensity, 11.6cps) of silicon-treated plants compared with the untreated (fluorescence intensity, 1.2cps) (P<0.05). Results from a fertigation field experiment in 2017 also found that more silicon was laid down regularly in the upper and lower cuticle, epidermis and palisade layer of the leaves (fluorescence intensity,

19.4cps) of silicon-treated plants compared with the untreated (fluorescence intensity, 7.9cps) (P<0.05). In the leaf petiole, more silicon was found in the xylem (fluorescence intensity, 16.7cps) of silicon-treated plants compared with the untreated (fluorescence intensity, 10.2cps) (P<0.05). In the roots, silicon was found in the xylem (fluorescence intensity, 8.4cps) of silicon-treated plants compared with the untreated (fluorescence intensity, 6.0cps) (P<0.05). The 2018 deposition field experiment showed that more silicon was laid down in the upper and lower cuticle, epidermis, and palisade layer of the leaves (fluorescence intensity, 2.2cps) (P<0.05). In the leaf petiole, silicon-treated plants compared with the untreated untreated (fluorescence intensity, 2.2cps) (P<0.05). In the leaf petiole, silicon-treated and untreated plants (fluorescence intensity, 3.72cps) of both silicon-treated and untreated plants (fluorescence intensity, 1.98cps) (P>0.05). In the roots, silicon was also mainly found in the xylem (4.97cps) of silicon-treated and untreated plants (1.76cps) (P>0.05). The hypothesis for chapter four is that the silicon can enhance the passive defence pathway of strawberry plants and is absorbed regularly in this manner.

Chapter five assessed strawberry plants grown hydroponically in Hoagland's solution to measure growth parameters between silicon-treated and untreated plants. This experiment revealed that the plants treated with silicon had significantly increased (P<0.05) numbers of leaves, runners and fruits compared with the untreated control. No significant differences (P>0.05) were found in the experiment's chlorophyll contents of strawberry leaves. These results suggested that silicon improved the quality of strawberry plants (treated with silicon) in a hydroponic glasshouse experiment by enhancing these growth parameters. This thesis demonstrates that the use of silicon via fertigation not only reduces the severity of strawberry powdery mildew (Podosphaeraaphanis) and but has some additional benefits in strawberry production. Therefore, it is recommended that growers incorporate silicon nutrient to manage strawberry production, including strawberry powdery mildew disease control.

Acknowledgements

Two roads diverged in a yellow wood, And sorry I could not travel both And be one traveller, long I stood And looked down one as far as I could To where it bent in the undergrowth.

Then took the other, as just as fair, And having perhaps the better claim, Because it was grassy and wanted wear. Though as for that, the passing there Had worn them really about the same,

And both that morning equally lay In leaves, no step had trodden black. Oh, I kept the first for another day. Yet knowing how way leads on to way, I doubted if I should ever come back.

I shall be telling this with a sigh Somewhere ages and ages hence: Two roads diverged in a wood, and I— I took the one less travelled by, And that has made all the difference. (Robert Frost)

Supervisory team: My appreciation and heartfelt thanks to my supervisors, Dr Avice M. Hall and Dr Keith Davies for their thorough support and guidance.

Family: To my loved ones, which I could not mention all here, my most sincere and honest gratitude for standing by me wholeheartedly each day passing. You are heroes, and this is a dedication to all of you.

Abstract	i
Acknowledgements	iii
Contents	iv
List of figures	xiii
List of tables	xviii
Conference communications	XX

TABLE OF CONTENTS

1.	CHA	APTER 1 – GENERAL INTRODUCTION	.1
	1.1.	Structure of the thesis	.1
	1.2.	Health and nutrition benefits of strawberries	.2
	1.3.	Strawberry production worldwide	.3
	1.4.	British growing systems	.8
	1.5.	Fertigation systems	.9
	1.6.	Coir bags on raised soil beds under polythene tunnels	11
	1.7. 1.7.1 1.7.2	Fleece and mulch 1 1. Fleece 1 2. Mulch 1	1 1 2
	1.8.	Tabletops1	14
	1.9.	Strawberry plant	15
	1.10.	Strawberry plant breeding	15
	1.11.	Harvest and shelf life	19
	1.12.	Pest and diseases of strawberries	20
	1.13. 1.13 1.13 1.13 1.13	Erysiphales (Powdery mildews) 2 9.1. Introduction 2 9.2. General classification of Erysiphales 2 9.3. Infection 2 9.4. The general life cycle of Erysiphales 2	22 22 23 23 24
	1.14. 1.14 1.14	Strawberry powdery mildew (Podosphaera aphanis) 2 1. The life cycle of Podosphaera aphanis 2 2. Infection and symptoms 2	25 25 26
	1.15. 1.15	Epidemiology and the environment 2 5.1. Epidemiology and spread	27 27
	1.16. 1.16 1.16 1.16	Plant defence mechanisms 2 5.1. Constitutive defence 2 5.2. Active defence 2 5.3. Silicon (Si) 2	29 29 33 33
	1.16. 1.16 1.16 1.16	Plant defence mechanisms 0.1. Constitutive defence 0.2. Active defence 0.3. Silicon (Si)	

1.17. The effects of silicon on plant diseases	34
2. CHAPTER 2 - GENERAL MATERIAL AND METHODS FOR ALL	
CHAPTERS	38
2.1. Farm description	38
2.1.1. Description of field experiments in 2016, 2017 and 2018	40
2.1.2. Effects of the use of silicon on disease levels of powdery mildew on	
strawberries: 2016 Ladybird field and 2017 to 2018 Amelia field	40
2.2 Leaf sample collection storage and assessments	43
2.2.1 Leaf sampling for disease assessments	
	4.4
2.3. Silicon extraction	44
2.4. Silicon content in fertigation water 2016, 2017 and 2018	46
2.4.1. Sampling	46
2.4.2. Measurement	48
2.5. Laboratory experiment methods	49
2.5.1. °Brix Measurements	50
2.6 Glasshouse experiment methods (not and compost)	53
2.7. Data analysis	57
2.7.1. Comparisons of disease levels between treatment	58
3. CHAPTER 3 - THE USE OF SILICON NUTRIENTS IN REDUCING	
DISEASE LEVELS OF STRAWBERRY POWDERY MILDEW	59
3.1. Introduction	59
3.2. Bioavailable silicon in plants	59
3.2.1. Rationale	60
3.2.2. Aim	60
3.2.3. Hypothesis	60
3.2.4. Objectives	60
3.3 Material and methods	61
3 3 1 Cultivars	61
3.3.2. Treatments	61
3.3.3. Leaf sampling	62
3.3.4. Disease assessments: Strawberry powdery mildew (<i>Podosphaera</i>	
aphanis) on the strawberry leaf surface from 2016 to 2018	63
3.3.5. Water sampling	63
3.3.6. Silicon extraction	63
3.4 Results	63
3.4.1 Disease levels of strawherry nowdery mildew in the silicon fertigation	05 1
field experiment 2016 (Malling Centenary)	- 64
2.5 gill a statistic statistic statistics of the	
3.5. Silicon content in strawberry leaves from the silicon fertigation field	60
experiment 2016 (Malling Centenary)	68

 3.6. Levels of disease in untreated (uncontrolled) strawberry plants in the silicon fertigation field experiment 2017 (Amesti)	1 2 4
 3.7. Levels of disease in untreated (uncontrolled) strawberry plants in the silicon fertigation field experiment 2018 (Amesti)	1 7 8
 3.8. Discussion	2 v
fertigation field experiment 2016 to 2018	3
3.9. Conclusion	4
4. CHAPTER 4 – SILICON DEPOSITION IN STRAWBERRY PLANTS	5
4.1 Introduction 8	5
4.1.1. Silicon deposition in plants	5
4.2 Patienale aim chiestives and hypothesis of silicon deposition in	
4.2. Rationale, and, objectives and hypothesis of sincon deposition in strawberry plants	6
4.2.1 Rationale 80	6
4.2.2. Aim	6
4.2.3. Objectives	6
4.2.4. Hypothesis	5
4.3 Material and methods 80	6
4.3.1 Silicon deposition	6
4 3.2 Plant sectioning 8'	7
4.3.3. LysoTracker Yellow HCK-123 fluorescence dve staining	7
4.3.4. GXML3201 LED fluorescence microscope	7
4.3.5. Silicon accumulation (fluorescence intensity) measurement	8
4.3.6. Treatments	9
4.3.7. Cultivars	9
4.4. Results	9
4.4.1. Pattern and deposition of silicon in the leaves, leaf petioles and roots	
from a glasshouse experiment 2017 (quantification of fluorescence intensity)) ; 4

4.4 sili	4.3. Pattern and deposition of silicon in leaves, petioles and roots from the icon fertigation field experiment 2018 (quantification of fluorescence	00
inte	ensity)	98
4.5.	Discussion	102
4.5 ex ₁	5.1. Silicon deposition in strawberry plants from a silicon fertigation field periment in 2017 and 2018 and a glasshouse experiment (pot compost) in 2 102	017
4.6.	Conclusion	105
5. CH PARAN	IAPTER 5 - MEASUREMENTS OF STRAWBERRY PLANT GROWTH METERS AND QUALITY ABOUT SILICON	106
5.1. conce	Measurements of strawberry plant growth parameters and quality erning silicon	106
5.2.	Rationale	108
5.3.	Aim	108
5.5.	Objectives	100
5.5	Hymothesis	100
5.5.	Material and matheda	100
3.0.	Material and methods	108
5.7.	Hydroponics (methods)	109
5.7	7.1. Number of Leaves	114
5.7 5.7	7.2. Number of runners	114
5.7	7.5. Flowering	114
5.7	7.5. Briv measurements	114
5.7	7.5. Drix measurements	115
5.7	7.7. Fresh weight biomass	117
5.8.	Results	118
5.8	8.1. Mean values of 10 strawberry fruit Brix per treatment from the silico	n
fer	tigation field experiment 2017.	118
5.8	3.2. Means of 10 strawberry leaf petioles °Brix per treatment from the silic	on
fer	tigation field experiment 2017	121
5.8	3.3. Effects of silicon nutrient on average values of leaf number (per plant	and
trea	atment) in the hydroponic experiment	124
5.8	8.4. Effects of silicon nutrient on average values of the number of runners	
(pe	er plant and treatment) in the hydroponic experiment	124
5.8	3.5. Effects of silicon nutrient on the number of flowers (per plant and	
trea	atment) in the hydroponic experiment	125
5.8	3.6. Effect of silicon nutrient on the average values of the number of fruits	in
the	e hydroponic experiment	126
5.8	3.7. Effects of silicon nutrient on the °Brix levels of fruit (per treatment) in	1
the	e hydroponic experiment	126
5.8	5.8. Effects of silicon nutrient on average values of chlorophyll levels (per	10.5
trea	atment) in leaves of plants grown hydroponically	126

5.8.9. Effects of silicon nutrient on the fresh weights (biomass) of whole strawberry plants grown hydroponically
5.8.10. Observations on plants grown hydroponically without silicon and
plants grown with silicon130
5.9. Discussion
5.9.1. Effects of silicon nutrient on the °Brix levels of fruits and leaf petioles in
the silicon fertigation field experiment 2017 and fruits of the Hydroponic
glasshouse experiment
5.9.2. Effects of silicon nutrient on the average values of the number of leaves
5.0.3 Effects of silicon nutrient on the average values of the number of runners
produced in the hydroponic experiment
5.9.4. Effects of silicon nutrient on flowering in the hydroponic experiment.132
5.9.5. Effects of silicon nutrient on the number of fruits produced in the
hydroponic experiment
5.9.6. Effects of silicon nutrient on chlorophyll contents in leaves of plants
grown hydroponically
5.9.7. Effects of silicon nutrient on the fresh weights (biomass) of plants grown
5.9.8 Observations on the health of plants grown hydroponically with and
without silicon
5 10 Complexity 125
5.10. Conclusion
6. CHAPTER 6 - OVERALL DISCUSSION AND CONCLUSION
6.1. Effects of silicon on powdery mildew disease <i>Podosphaera aphanis</i> on Malling Centenary and Amesti Cultivar
6.2. Silicon deposition in strawberry plants cultivar (passive defence pathway) 137
6.3. Benefits of silicon nutrients use on strawberries
6.4. Conclusion
6.5. Future work140
REFERENCES141

APPENDICES

Appendix trial 2	1 silicon fertigation field trial 2016 (ladybird field) silicon fertigation f 2016 (ladybird field)	ield 157
Appendix	2 silicon fertigation field trial 2017 (amelia field)	158
Appendix	3 silicon fertigation field trial 2018 amelia field	159
Appendix	4 strawberry powdery mildew disease assessment key by jin (2015)	160
Appendix	5 safety data sheet	161
Appendix	6 safety data sheet cont'd	162

Appendix 7 maps of maltmas farmmap a - rural land register map of maltmas farm
Appendix 8 map b - distribution of strawberry field in maltmas farm
Appendix 9 list of fungicides for the silicon fertigation field trial 2016165
Appendix 10 fungicides sprays used for the silicon fertigation field trial 2017 cont'd
Appendix 11 fungicides sprays used for the silicon fertigation field trial 2018 cont'd
Appendix 12 fungicides sprays used for the silicon fertigation field trial 2018 cont'd
Appendix 13 calculation of silicon nutrients (sirius) application rate in the 2016 to 2018 silicon fertigation field trials at maltmas farm
Appendix 14 chapter 3 - results workings for disease levels of strawberry powdery mildew in the silicon fertigation field experiment 2016 (malling centenary)170
Appendix 15 raw data for silicon extraction from strawberry leaves in the silicon fertigation field trial 2016 (chapter 3)
Appendix 16 raw data for silicon extraction from strawberry leaves in the silicon fertigation field trial 2016 cont'd (chapter 3)
Appendix 17 chapter 3 results workings for silicon extraction between treatment in the 2016 field experiment
Appendix 18 correlation analysis between silicon extracted and levels of disease in the 2016 field experiment
Appendix 19 raw data for silicon extraction from strawberry leaves in the silicon fertigation field trial 2017 (chapter 3)
Appendix 20 chapter 3 results workings for silicon extraction in the silicon fertigation field experiment 2017
Appendix 21 raw data for silicon extraction from strawberry leaves in silicon fertigation field trial 2018 (chapter 3)
Appendix 22 chapter 3 results workings for silicon extraction in the silicon fertigation field experiment 2018
Appendix 23 protocol for lysotracker yellow hck-123 fluorescence dye staining (chapter 4)
Appendix 24 silicon fluorescence intensity quantification using imagej (chapter 4)
Appendix 25 cross-sections (replicates) of strawberry leaves in the glasshouse experiment 2017 (x40 and x400 magnifications) chapter 4190
Appendix 26 cross-sections (replicates) of strawberry leaves in the glasshouse experiment 2017 (x40 and x400 magnifications) cont'd, chapter 4191
Appendix 27 : cross-sections (replicates) of strawberry leaves in the glasshouse experiment 2017 (x40 and x400 magnifications) cont'd, chapter 4192

ix

Appendix 28 cross-sections (replicates) of strawberry leaf petioles in the glasshouse experiments 2017 (x40 and x400 magnifications) chapter 4......193 Appendix 29 cross-sections (replicates) of strawberry leaf petioles in the glasshouse experiments 2017 (x40 and x400 magnifications) chapter 4......194 Appendix 30 cross-sections (replicates) of strawberry leaf petioles in the glasshouse experiments 2017 (x40 and x400 magnifications) cont'd, chapter 4......195 Appendix 31 cross-sections (replicases) of strawberry roots in the glasshouse Appendix 32 cross-sections (replicases) of strawberry roots in the glasshouse experiments 2017 (x40 and x400 magnifications) cont'd, chapter 4.....197 Appendix 33 cross-sections (replicases) of strawberry roots in the glasshouse experiments 2017 (x40 and x400 magnifications) cont'd, chapter 4......198 Appendix 34 cross-sections (replicates) of strawberry leaves in the silicon fertigation Appendix 35 cross-sections (replicates) of strawberry leaves in the silicon fertigation field 2017 (x40 and x400 magnifications) cont'd chapter 4.....200 Appendix 36 : cross-sections (replicates) of strawberry leaves in the silicon fertigation field 2017 (x40 and x400 magnifications)......201 Appendix 37 cross-sections (replicates) of strawberry leaf petioles in the silicon field experiments 2017 (x40 and x400 magnifications) chapter 4......202 Appendix 38 cross-sections (replicates) of strawberry leaf petioles in the field experiment 2017 (x40 and x400 magnifications) cont'd, chapter 4203 Appendix 39 cross-sections (replicates) of strawberry leaf petioles in the silicon field experiment 2017 (x40 and x400 magnifications) cont'd, chapter 4204 Appendix 40 cross-sections (replicates) of strawberry roots in the field experiment 2017 (x40 and x400 magnifications) chapter 4.....205 Appendix 41 cross-sections (replicates) of strawberry roots in the field experiment 2017 (x40 and x400 magnifications) cont'd, chapter 4206 Appendix 42 cross-sections (replicates) of strawberry roots in the field experiment 2017 (x40 and x400 magnifications) cont'd, chapter 4207 Appendix 43 cross-sections (replicates) of strawberry leaves in the silicon fertigation field 2018 (x40 and x400 magnifications) chapter 4208 Appendix 44 : cross-sections (replicates) of strawberry leaves in the silicon fertigation field 2018 (x40 and x400 magnifications) cont'd, chapter 4......209 Appendix 45 cross-sections (replicates) of strawberry leaves in the silicon fertigation field 2018 (x40 and x400 magnifications) cont'd, chapter 4......210 Appendix 46 cross-sections (replicates) of strawberry leaves in the silicon fertigation field 2018 (x40 and x400 magnifications) cont'd......211 Appendix 47 cross-sections (replicates) of strawberry leaf petioles in the silicon fertigation field 2018 (x40 and x400 magnifications) chapter 4......212

Appendix 48 cross-sections (replicates) of strawberry leaf petioles in the silico fertigation field 2018 (x40 and x400 magnifications) cont'd chapter 421
Appendix 49 cross-sections (replicates) of strawberry leaf petioles in the silico fertigation field 2018 (x40 and x400 magnifications) cont'd chapter 421
Appendix 50 cross-sections (replicates) of roots in the silicon fertigation field 201 (x40 and x400 magnifications) chapter 4
Appendix 51 cross-sections (replicates) of roots in the silicon fertigation field 201 (x40 and x400 magnifications) cont'd chapter 4
Appendix 52 fluorescence images of untreated and silicon-treated flower stalks ar strawberry fruit, from 2018 silicon field, chapter 421
Appendix 53 fluorescence images of untreated and silicon-treated only achenes strawberry plants, from 2018 silicon field, chapter 4
Appendix 54 (imagej data sheet) raw fluorescence intensities from chapter 421
Appendix 55 (imagej data sheet) raw fluorescence intensities from chapter 422
Appendix 56 (imagej data sheet) raw fluorescence intensities; chapter 422
Appendix 57 (imagej data sheet) raw fluorescence intensities; chapter 422
Appendix 58 imagej data sheet) raw fluorescence intensities; chapter 422
Appendix 59 (imagej data sheet) raw fluorescence intensities; chapter 4
Appendix 60 chapter 4 - results workings means of 10 fluorescence intensitie (integrated density) in the leaf, leaf petiole and root from a glasshouse experime 2017
Appendix 61 chapter 4 - results workings means of 10 fluorescence intensitie (integrated density) in the leaf, leaf petiole and root from a field experiment 201
Appendix 62 chapter 4 - results workings means of 10 fluorescence intensitie (integrated density) in the leaf, leaf petiole and root from a field experiment 201
Appendix 63 hoagland's solution recipe (chapter 5)
Appendix 64 chapter 5 - results workings - means of 10 strawberry fruits ^o brix leve from the silicon field experiment 201723
Appendix 65 chapter 5 - results workings - means of 10 strawberry leaf petioles ^o br levels from the silicon field experiment 201724
Appendix 66 chapter 5 - results workings - means number of leaves from the hydroponic experiment 2018
Appendix 67 chapter 5 - results workings - means number of runners from the hydroponic experiment 2018
Appendix 68 mean numbers of fruits per plant tubs counted at two sample dates25
Appendix 69 chapter 5 - results workings - mean number of strawberry fruits from the hydroponic experiment 2018
 Appendix 66 chapter 5 - results workings - means number of leaves from the hydroponic experiment 2018

Appendix 7 silicon-t	1 mean of 10 leaves, per plant (chlorophyll content) from untreated an reated plants	1d 57
Appendix 72 the leave	2 chapter 5 - results workings - means number of chlorophyll contents es from the hydroponic experiment 2018	of 58
Appendix 73 strawber	B fresh weights biomass (no fruits were included) of 10 individual who rry plants (untreated and silicon-treated)	le 59
Appendix 74	coshh forms (field experiments)20	50
Appendix 75	coshh forms (laboratory and glasshouse experiments)	56
Appendix 76	5 list of posters	75

LIST OF FIGURES

CHAPTER 1 GENERAL INTRODUCTION

Figure 1.1. (a) worldwide production and yield of strawberries (b) top 10 strawberry- producing countries source: faostat, 2020
Figure 1.2. Production and yield quantities of strawberries. (a) production and yield quantities of strawberries in china (b) production and yield quantities of strawberries in the united states of america (c) production and yield quantities of strawberries in spain (d) production and yield quantities of strawberries in mexico. The blue line indicates the area of harvested strawberries, and the red line is produced from 1994 to 2018. Source: (faostat, 2020)
Figure 1.3. Production and yield quantities of strawberries in the united kingdom. Source: faostat (2020)7
Figure 1.4. Strawberry plants are grown in coir bags. Each bag is placed on raised beds
Figure 1.5. Availability of nutrients at varying ph values. Source: almanac (2020). The image shows ph ranges of crop nutrients that are under the category of acidic to alkaline
Figure 1.6. Raised soil bed strawberries11
Figure 1.7. Fleece-covered strawberry beds under a polythene tunnel12
Figure 1. 8. Fleece-covered strawberries in spring (before the polythene is put on). Fleece is put over the crop for frost protection
Figure 1 9. Tabletop strawberries. Source: (palmers, 2018). The photo shows strawberries growing on tabletops on a field. During the fruiting season, fruits will hang down from the plants, as shown in figure 1.9, allowing easy picking compared to the crops growing on the ground14
Figure 1.10. Anatomy of a strawberry plant. Source: strawberryplants (2020). The fully established (mature) strawberry plant consists of woody/ fibrous roots (with
secondary and primary roots), a crown from, which the leaves (with 3 leaflets) arise, flowers (not shown in the picture) and runners, also known as daughter plants
 secondary and primary roots), a crown from, which the leaves (with 3 leaflets) arise, flowers (not shown in the picture) and runners, also known as daughter plants. Figure 1.11. Anatomy of the strawberry fruit. (a) strawberry plants in halves (b) structure of the strawberry fruit and (c) tertiary and secondary fruits of the strawberry plant.
 secondary and primary roots), a crown from, which the leaves (with 3 leaflets) arise, flowers (not shown in the picture) and runners, also known as daughter plants
 secondary and primary roots), a crown from, which the leaves (with 3 leaflets) arise, flowers (not shown in the picture) and runners, also known as daughter plants
 secondary and primary roots), a crown from, which the leaves (with 3 leaflets) arise, flowers (not shown in the picture) and runners, also known as daughter plants. Figure 1.11. Anatomy of the strawberry fruit. (a) strawberry plants in halves (b) structure of the strawberry fruit and (c) tertiary and secondary fruits of the strawberry plant. Figure 1.12. Strawberry plant flowers in a pot glasshouse experiment. (photo taken when automatic lights in the glasshouse were switched on). Figure 1.13. Malling centenary strawberry runners attached to mother plants (pot compost) in uh glasshouse experiment. Figure 1.14. The appearance of a strawberry plant grown in hydroponics hoagland's solution.

Figure 1.16. Diseases of strawberry plants. (a) <i>botrytis cinerea</i> (fruit rot) (b) <i>phytophthora fragariae</i> (red core) (c) <i>phytophthora cactorum</i> (crown rot) (d) <i>verticillium dahlia</i> (verticillium wilt). Source: alchetron (2020); pnwhandbooks.org (2020)
Figure 1.17. Phylogeny of tribes and genera of <i>erysiphales</i> . Adapted from braun, 2011 and meeboon and takamatsu, 2017
Figure 1.18. The life cycle of strawberry powdery mildew (podosphaera aphanis)24
Figure 1.19. Symptoms of <i>podosphaera aphanis</i> on strawberries. (a) leaf cupping (b) leaf blotching (c) mycelium on leaves (d) infected flower (e) mycelium on unripe fruit, and (f) mycelium on ripe fruit. Source: (hall, jin and dodgson, 2016)27
Figure 1.20. Fungicide crop sprayer on ladybird field
Figure 1.21. The cross-section of a strawberry leaf (uv light)
Figure 1.22. Plant defences. Source: binder and parniske (2018). The figure shows that plant defence mechanism are divided into two main pathways. The passive or the constitutive and the active or the induced defence mechanism
Figure 1.23. Silicon deposition in the epidermal cell of leaves of rice plants. Source: debona et al. (2017)

CHAPTER 2 MATERIAL AND METHODS

Figure 2.1. (a) map location of maltmas farm,	, wisbech, cambridgeshire. Source: google
maps (2018). (b) map location of maltma	as farm, wisbech, cambridgeshire. Source:
google maps (2018)	

- Figure 2. 2. Heat sealed with the plastic film machine; e) sealed strawberry punnets are separated; f) labels with information on strawberry variety, classification and producer are requested by supermarkets; g) strawberry punnets stored in cold room waiting to be transported to supermarkets. Source: liu (2016)......40
- Figure 2.3. The ladybird (a) and amelia silicon field (b) experiment plan 2016 to 2018. Ladybird tunnel (180 metres) contains 6 treatments, while the amelia tunnel (180 metres) contains 4 treatments. Silicon is applied through the fertigation tubes (drippers), and fungicides are applied through a crop sprayer (figure 1.20).42

- Figure 2.8. (a) garlic crusher (left) and (b) refractometer (right)......51

Figure 2.10. A and b: location map of bayfordbury source: google maps (2020)......53

CHAPTER 3 THE USE OF SILICON IN REDUCING DISEASE LEVELS OF strawberry powdery mildew

Figure 3.1. Disease levels in the 2016 silicon fertigation field experiment
Figure 3.2. Flooded ladybird in 201665
Figure 3.3. Silicon extraction from strawberry leaves (μ g/mg) in the fertigation field in 2016 (at the end of the growing season in august)
Figure 3.4. Disease levels in the 2017 silicon fertigation field experiment73
Figure 3.5. Silicon extraction from strawberry leaves (μ g/mg) in the fertigation field in 2017 (at the end of the growing season in august)
Figure 3.6. Disease levels in 2018 silicon fertigation field experiment77
Figure 3.7. Silicon extraction from strawberry leaves ($\mu g/mg$) in the silicon fertigation

CHAPTER 4 SILICON DEPOSITION IN STRAWBERRY PLANTS

Figure 4.1. Gxml3201 led fluorescence microscope with a cross-section sample at x 4 magnification	0
Figure 4.2. Silicon deposition in glasshouse experiment 20179	0
Figure 4.3. Silicon deposition in silicon field experiment 2017. Silicon deposits in gree fluorescence	n 5
Figure 4.4. Figure silicon deposition in silicon field experiment 20189	9

CHAPTER 5 MEASUREMENT OF STRAWBERRY PLANT GROWTH PARAMETERS AND QUALITY IN RELATION TO SILICON

Figure 5. 1. A design plan showing the treatment and plants used for the hydroponic experiment in 2018. T = treated - with silicon. U = untreated - without silicon, total number of plants used in experiment = 10 silicon-treated and 10 untreated = 20.

- Figure 5. 5. Fruiting in strawberry plants in the hydroponic experiment 2018115
- Figure 5. 6. The chlorophyll meter spad-502plus. Lightweight and water-resistant. 116
- Figure 5. 7. A harvested strawberry plant from the hydroponic experiment at 22 weeks. End of experiment......117
- Figure 5. 8. Means of 10 strawberry fruits °brix levels from the silicon field experiment 2017. The different bars are average values of strawberry fruits sampled from 20 july to 1 august 2017. Light green is untreated, the blue is the fungicides-only treatment, the yellow is silicon applied twice weekly + fungicides, and the dark green is silicon applied twice weekly without fungicides. Means from 10 strawberries per treatment were used to create the graph shown. The paired sample for means was used to analyse the data collected as it was normally distributed.

- Figure 5. 12. Flowering to fruiting period in the hydroponic experiment 2018......126

LIST OF TABLES

CHAPTER 1 GENERAL INTRODUCTION

Table 1. 1 world strawberry-producing countries	4
Table 1.2. Characteristics of june bearers and ever bearer strawberries	6
Table 1.3. Source of uk strawberries and country of strawberry imports	7
Table 1.4. Types of strawberry production in the uk	8
Table 1.5. Strawberry plant anatomy and morphology	17
Table 1.6. Conditions for strawberry powdery mildew conidia.	
Table 1.7. Effects of silicon on disease severity	
Table 1.8. Effects of silicon on disease severity continued	

CHAPTER 2 MATERIAL AND METHODS

Table 2.1. A summary of all field experiments carried out at maltmas farm,	wisbech,
between 2016 and 2018	41
T 11 2 2 C1 1	57
Table 2.2. Glasshouse experiments in 2017 and 2018	

CHAPTER 3 THE USE OF SILICON IN REDUCING DISEASE LEVELS OF STRAWBERRY POWDERY MILDEW

Table 3.1. Treatments used in the silicon fertigation field experiments from 2016 to 2018
Table 3.2. Summary of findings for disease levels of strawberry powdery mildew in the silicon fertigation field experiment 2016 (malling centenary)
Table 3.3. Summary of silicon extraction from strawberry leaves in the 2016 field experiment.
Table 3.4. Silicon concentration (mg/ml) in fertigation field water 201671
Table 3.5. Summary table of silicon extraction from strawberry leaves in the silicon fertigation field experiment 2017
Table 3.6. Silicon content (mg/ml) in fertigation field water 201776
Table 3.7. Summary table of silicon extraction from strawberry leaves in the silicon fertigation field experiment 2018
Table 3.8a. Silicon content (mg/ml) in fertigation field water

CHAPTER 4 SILICON DEPOSITION IN STRAWBERRY PLANTS

Table 4.1. Readings from fluorescence intensities (integrated density) in the leaf, leaf petiole and root from a glasshouse experiment 2017. Values are the mean of 3 readings per leaf, leaf petioles and roots
Table 4.2. Summary of 10 fluorescence intensities (integrated density) in the leaf, leafpetiole and root from a glasshouse experiment 2017
Table 4.3. Fluorescence intensities (integrated density) in the leaf, leaf petiole and root from a field experiment 2017
Table 4.4. Summary of 10 fluorescence intensities (integrated density) in the leaf, leafpetiole and root from a field experiment in 2017
Table 4.5. Readings from fluorescence intensities (integrated density) in the leaves, leaf petioles, and roots from the silicon fertigation field experiment 2018101
Table 4.6. Summary of 10 fluorescence intensities (integrated density) in the leaf, leaf petiole and root from a field experiment in 2018

CHAPTER 5 MEASUREMENT OF STRAWBERRY PLANT GROWTH PARAMETERS AND QUALITY IN RELATION TO SILICON

Table 5. 1 benefits of using silicon on non-accumulators of silicon	.07
Table 5. 2 nutrient components for hoagland's solution1	.12
Table 5. 3 summary of °brix levels from strawberry fruits from the 2017 fi experiment	eld 19
Table 5. 4 summary of obrix levels from strawberry fruits from the 2017 fi experiment (continued)	eld 20
Table 5. 5 summary of °brix levels from strawberry leaf petioles from the 2017 fi experiment	eld 22
Table 5. 6 summary of °brix levels from strawberry leaf petioles from the 2017 fi experiment	eld 23
Table 5. 7 summary of findings in the hydroponic experiment 2018	.28

CONFERENCE COMMUNICATIONS

BSPP presidential conference-meeting - Poster presentation Title: Reducing disease levels of strawberry powdery mildew using a silicon nutrient to give reduced susceptibility to *Podosphaera aphanis*) – St Hugh's College, Oxford, UK. 11-13th September 2016. Asiana, I., Hall A.M., Davies K.

Crop protection in Southern–Britain - Poster presentation, Title: Reducing disease levels of strawberry powdery mildew using a silicon nutrient to give reduced susceptibility to *Podosphaera aphanis* – Peterborough, UK, 15 -16th February 2017. Asiana, I., Hall A.M., Davies K.

Lunch time seminar – Oral presentation; Title: Reducing disease levels of strawberry powdery mildew using a silicon nutrient to give reduced susceptibility to *Podosphaera aphanis*. Asiana, I., Hall A.M., Davies K. 12th June 2017, University of Hertfordshire, college lane, UK.

BSPP presidential conference-meeting - Poster presentation, Title: Reducing disease levels of strawberry powdery mildew using a silicon nutrient to give reduced susceptibility to *Podosphaera aphanis*) – The University of Nottingham, UK 11- 13th September 2017. Asiana, I., Hall A.M., Davies K.

Lunch time seminar – Oral presentation; Title: Reducing disease levels of strawberry powdery mildew using a silicon nutrient to give reduced susceptibility to *Podosphaera aphanis*. Asiana, I., Hall A.M., Davies K. 12th January 2018, University of Hertfordshire, college lane, UK.

LMS Research Day- LMS Research Day; Oral presentation; Title: <u>Silicon deposition in</u> <u>strawberry plants.</u> Asiana, I., Hall A.M., Davies K. 16th April 2018, University of Hertfordshire, college lane.

ICPP International Congress of Plant Pathology conference meeting; Poster presentation; <u>Silicon deposition in strawberry plants;</u> Asiana, I., Hall A.M., Davies K, Boston Massachusetts, The United States 29 July – 3rd August 2018.

Berry Gardens – **Research** – **Poster Competition** (Annual Technical Conference) – Prize winner of poster competition 24th November 2018 – <u>Title: The deposition of silicon linked to</u> the reduction in susceptibility to strawberry powdery mildew; Asiana, I., Hall A.M., Davies K.

BSPP presidential conference meeting- Poster presentation, Title: <u>Are strawberries ever</u> <u>deficient in silicon?</u>; University of Warwick 10th – 11th December 2018. Asiana, I., Hall A.M., Davies K.

BSPP presidential conference meeting: Oral presentation; Title: <u>How does a silicon nutrient</u> <u>enhance the passive defence pathway of strawberries to improve disease control?</u>; Asiana, I., Hall A.M., Davies K. University of West England, Bristol, UK. 2nd – 3rd September 2019. PH Gregory competition.

Lunch time seminar – Oral presentation; Title: The benefits of using silicon nutrient on strawberries, including reducing disease levels of powdery mildew (*Podosphaera aphanis*). Asiana, I., Hall A.M., Davies K. November 11th 2019.

Lunch time seminar_- Oral presentation; <u>Title: How does a silicon nutrient enhance the</u> passive defence pathway of strawberries to improve disease control. Asiana, I., Hall A.M., Davies K University of Hertfordshire, College lane. November 14th 2019.

CHAPTER 1 – GENERAL INTRODUCTION

1.1. Structure of the thesis

Chapter one provides general information about the thesis, focusing on worldwide strawberry production. It covered the health and nutrition benefits of strawberries (section 1.2), a general introduction to strawberry production worldwide (section 1.3), British growing systems (section 1.4), the strawberry plant (section 1.9) and strawberry plant breeding (section 1.10). This chapter introduces *Erysiphales* (section 1.13) and strawberry powdery mildew Podosphaera aphanis (Wallr.). Chapter two provides the general material and methods, including farm description (section 2.1), laboratory (section 2.5) and glasshouse experiment methods (section 2.6). Chapter three provides an introduction to the chapter (section 3.1), rationale, aims, objectives and hypothesis (section 3.2.1 to 3.2.4), material and methods used in the chapter (section 3.3), results (section 3.4), discussion (section 3.8) and conclusion (section 3.9). Chapter four provides an introduction to the chapter (section 4.1), rationale, aim, objectives and hypothesis of the chapter (section 4.2.1 to 4.2.4), material and methods (section 4.3), results (section 4.4), discussion (section 4.5) and conclusion (section 4.6). Chapter five provides the introduction to the chapter (section 5.1), rationale, aim, objectives and hypothesis of the chapter (section 5.2 to 5.5), material and methods (section 5.6), results (section 5.8), discussion (section 5.9) and conclusion (section 5.10). Finally, chapter six provides the overall discussion and conclusion of chapters in the whole thesis divided into sections (6.1 to 6.3), conclusion (section 6.4) and future work (section 6.5).

Previously, Jin (2015) conducted experiments to determine the efficacy of silicon nutrient in reducing strawberry powdery mildew. In the study, silicon was applied through foliar sprays and root treatments via a fertigation system once-a-week only. The work reported by Jin (2015) also explored silicon concentration in fertigation water in a 1-year experiment and localization in a glasshouse experiment alone.

The work conducted in commercial field tunnels in this thesis compared the use of silicon in reducing levels of disease of strawberry powdery mildew for 3 consecutive years. Firstly, in 2016, a field experiment was conducted to evaluate the differences in disease reduction between strawberry crops applied with both silicon once-a-week and twice-a-week. Following the 2016 field experiment, silicon was applied twice-a-week only in the 2017 and 2018 field experiments as it showed a higher reduction in disease levels than the once-a-week silicon application in 2016. The silicon contents in strawberry leaves and fertigation water were assessed for 3 years (2016 to 2018). Furthermore, a deposition experiment was set up for the quantification analysis of silicon deposits in plant tissues (leaves, petioles and roots) in a glasshouse and two field experiments (2017 and 2018). Lastly, strawberry plant growth parameters were quantified between silicon-treated and untreated strawberry plants in a hydroponic glasshouse experiment in 2018.

1.2. Health and nutrition benefits of strawberries

Strawberries are beneficial to human health and nutrition. Researchers have found that their vitamin C content produces 30% of strawberries' total antioxidant capacity (TAC) (Li et al.,2019). Vitamin C is essential to a healthy immune system as it stimulates the white blood cells, which form a defence against infections. According to Danielsen (2018), research on the antioxidant content of strawberries has provided evidence about their ability to lower the risk of cardiovascular disease. Recent studies have shown that improved blood sugar regulation is a health benefit associated with strawberry consumption, mainly after meals (Brown-Paul, 2016).

Researchers have discovered an enhanced regulation of insulin and blood sugar levels in connection with strawberry intake (Foito et al., 2018). Alongside vitamin C, strawberries provide other vital antioxidants and anti-inflammatory nutrients. Strawberries are an excellent source of manganese (29%), a mineral that plays a key antioxidant role as a cofactor for the enzymes superoxide dismutase (Kumar et al., 2019).

Dai (2018) reported that strawberries are not a high-fat food but contain seeds and serve as a good source of omega-3 fatty acid alpha-linolenic acid. Strawberries also offer significant quantities of other nutrients such as folate (vitamin B9), vitamin B6, calcium, iron, copper, phosphorus, potassium, traces of selenium and dietary fibre. Joint inflammation, such as arthritis and gout, can be controlled by the regular consumption of strawberries (Legard, 2018).

The presence of nutrients such as vitamin C, folate, anthocyanins, ellagic acid and ellagitannins in strawberries helps prevent the growth and progression of cancer cells drastically and reduces the chances of tumour formation (Turgut and Cakmakci, 2018). A study has shown that strawberries produce fat-burning and metabolic hormones such as adiponectin and leptin, aiding weight loss. Calvano et al. (2019) pointed out that strawberries' potassium and magnesium are nutrients that effectively control blood pressure.

Strawberries contain a large amount of dietary fibre, which is essential in promoting healthy digestion (Morris and Sistrunk, 2018). Strawberries have high levels of phenolic antioxidants, with levels 2 to 11 times higher than other fruits (Zitouni et al., 2020). The health benefits of strawberries arise from consuming 1 to 2 servings a day (50 grams per serving) and, when consumed daily, contribute to supporting health (Strawberryplants, 2020).

1.3. Strawberry production worldwide

Strawberries (*Fragaria ananassa*) are a popular fruit in many parts of the world. They are widely grown and globally cultivated for their fruits and are best known for their distinctive succulent nature, bright red colour, texture and aroma (Cockerton et al., 2018). Strawberries are either fresh or used to prepare various foods such as jams, juices, milkshakes, ice cream and pies. According to Simpson (2018), the garden strawberry was first propagated during the 1750s in Brittany, France, through crossbreeding *Fragaria virginiana* from the eastern regions of North America and *Fragaria chiloensis* from Chile, which was brought to Europe in 1714. The figures below include 1.1 (a) Worldwide production and yield of strawberries and (b) Top 10 strawberry-producing countries.



Figure 1.1. (a) Worldwide production and yield of strawberries (b) Top 10 strawberry-producing countries Source: FAOSTAT, 2020.

In figure 1.1a, the blue dotted line on the graph indicates the area harvested strawberries, and the red dotted line indicates production in tonnes. In figure 1.1(b), the bar chart represents the production of strawberries by countries in tonnes. The worldwide production of strawberries is estimated at around 8.3 million tonnes per annum (figure 1.1a and b) (FAOSTAT, 2020); From the graphs displayed, China is the world's leading producer of strawberries with 2,964,263 tonnes of yearly production (figure 1.2a and table 1.1) (FAOSTAT, 2020). The second largest producer of strawberries is the United States, with 1,296,272 tonnes yearly production of the world production of strawberries (figure 1.2b) (FAOSTAT, 2020). Spain is third in world production with 366,161 tonnes yearly (figure 1.2c) after the United States, followed

by Mexico with 658,436 tonnes produced yearly (1.2d) (FAOSTAT, 2020). The production, harvest and yield of strawberries produced are summarised in table 1.1. Below (figure 1.2 a to d) are the 4 top strawberry-producing countries in the world according to FAOSTAT (2020). In each of the graphs, the blue dotted lines indicate the area of harvested strawberries, and the red dotted lines indicate the production of strawberries in tonnes between 1994 to 2018.

Table 1.1 below is a list of 11 strawberry-producing countries. The 4 top strawberry producers are also shown in the table below. China is the largest producer. The United States is the second largest producer, Spain is the third, and Mexico is the fourth largest producer.

Country	Production	Production per	Harvested	Yield (Kg/Ha)
	tonnes	Capita (kg)	Area (Ha)	
China	2,964,263	2.7	111,132	26,869
United States of America	1,296,272	4.3	190,919	66,876
Spain	366,161	7.8	7,032	47,648
Mexico	658,436	3.8	13,850	42,219
Turkey	440,968	5.1	16,102	26,904
South Korea	213,054	3.8	5,658	30,906
Japan	163,486	1.3	5,259	29,432
Poland	196,972	5.1	50,600	3,893
Russian	195,578	1.3	29,520	47,833
Federation				
Egypt	362,639	4.8	8,880	46,566
Germany	143,221	1.7	14,299	10,016

Table 1. 1 World strawberry-producing countries

Adapted from FAOSTAT (2020) and Atlasbig (2018).



Figure 1.2. Production and yield quantities of strawberries. (a) Production and yield quantities of strawberries in China (b) Production and yield quantities of strawberries in the United States of America (c) Production and yield quantities of strawberries in Spain (d) Production and yield quantities of strawberries in Mexico. The blue line indicates the area of harvested strawberries, and the red line is produced from 1994 to 2018. Source: (FAOSTAT, 2020).

Over the last 20 years, the UK season for strawberry production has increased from more than six weeks to more than seven months (Orde and Sideman, 2019). This has been achieved through polythene tunnels, increased Everbearing varieties (table 1.2), advanced growing systems using coir on raised soil beds, and table-top with automatic fertigation systems (Forcelini & Peres, 2018). Strawberry cultivation in the UK expanded very quickly towards the end of the 19th century and, by 1924, after a short drop in production during the First World War, reached its peak at 13,000Ha (Calleja, Ilbery and Mills, 2012).

The strawberry plant can be classified as a June bearer or an Ever bearer (day-neutral). Everbearing strawberries, also known as perpetual fruiting types, will produce crops multiple times during the growing season (Simpson, 2018). Junebearing strawberries usually produce a harvest in June, which can be further classified into early season, midseason and late session. (Kumar, 2018). The differences and characteristics of these two types are explained further in table 1.2 below. Growers grow the June bearers as an early crop, and the Ever bearers are a continuous crop. Production output has increased by introducing new varieties and growing systems, improving crop productivity and fruit quality (Simpson, 2018). Table 1.2 below is a short list showing the characteristics of June bearers and Ever bearers of strawberries.

June bearers (day length)	Ever bearers (day-neutral)
Develop flower buds in late summer	Planted in February/March (covered with
	fleece until April) and produce branch crowns
	and buds throughout the season. However, a
	higher temperature may inhibit the initiation.
Begin fruiting in June, lasting up to 3-	Begin fruiting in mid-May or June, lasting for
4 weeks.	10-12 weeks.
	Harvest is twice a year: One in spring and one
Harvest is once a year	in late summer or early autumn.
They produce large fruits and heavy	They produce smaller fruits and less heavy
cropping over two to three weeks.	cropping and produce fewer runners.
Produce more runners.	

Table 1.2. Characteristics of June bearers and Ever bearer strawberries.

According to the FAOSTAT (2020), The UK produces 131,639 tonnes of strawberries yearly (figure 1.3). The UK produces approximately 80% of the UK demand for strawberries compared to only 20% 20 years ago (Anderson, Rogers and Hoover, 2019). Farminguk (2019) reported that the value of the berry industry had pushed over \pounds 1.4 billion since it now has a quarter of the market share of all fruit grown in the UK.

The graph below (figure 1.3) shows the trend of strawberries produced (tonnes) and harvested (hectares) between 1994 to 2018 in the UK alone. In 1994, there was a decline in the areas harvested. However, production is showing a gradual increase over time.



Figure 1.3. Production and yield quantities of strawberries in the United Kingdom. Source: FAOSTAT (2020).

The UK strawberry season begins from April to May and up till October however, in the colder temperatures (winters), it grows and harvests strawberries in polythene tunnels and glasshouses and sources strawberries from countries listed in table 1.3 below. The UK production of strawberries is from April to October, Spain produces strawberries from January to April, Morocco produces strawberries at its peak season from December to April, Egypt produces strawberries from November to April, and Israel produces strawberries from November to February.

Table 1.3.	Source of UK	strawberries	and country	of strawberry	[,] imports.
			•		1

Country	Jan	Feb	March	April	May	June	July	August	Sep	Oct	Nov	Dec
UK				Ó	Ó	Ó	Ó	Ó	۲	Ó		
Spain	٢	٢	٢	٢								
Morocco	٢	٢	٢	Ó								٢
Egypt	۲	Ó	Ó	Ó							Ó	۷
Israel	Ó	Ő									۷	Ő

Source: Sampson and Kirk (2016).

There is a variation in cultivation systems for strawberry production in the UK (Whitehouse et al., 2016). They include the growth medium (soil, coir, substrate), crop variety (June bearer, Ever bearer), planting time (spring planted, summer planted), years of cropping (1 to 3) and polythene tunnels and glasshouses (Campos, 2018). Table 1.4 below is a classification of strawberry growing systems used by growers in the UK. British growing systems for strawberry production are classified by: crop protection

(polythene tunnels or glasshouse), growing medium (soil, peat, Hoagland's or coir), variety of crop (June bearers, Ever bearers) and planting time (grown in spring, summer).

Classification	Growing system						
Protection	Protected crops (i.e., polythene tunnels, glasshouse)						
	Unprotected crops (i.e., open fields)						
Growth medium	Soil grown						
	Substrate (i.e., coir) grown crops						
Growth methods	Bags and troughs on raised beds						
	Bags and troughs on tabletops						
Planting time	Spring planted						
	Summer planted						
Crop variety	June bearer						
	Ever bearer						

Table 1.4. Types of strawberry production in the UK

Source: Husaini and Neri (2016)

1.4. British growing systems

Strawberries are grown in a wide range of production systems to produce the objective of producing high yields of quality fruit with sufficient flexibility to meet market demand and labour availability. Twitchen (2018) mentioned that polythene tunnels enable fruit growers to extend the strawberry fruit season from May to mid-autumn. Brown-Paul (2016) suggested that through polythene tunnels, there is guaranteed fruit quality produced by protecting the crops from rain and other environmental conditions. There has been a reduction in crop wastage, improvements in yields and control in labour costs through the use of the polythene tunnels (Allen et al., 2015).

Polythene tunnels (figures 1.6 to 1.8) shield crops from rain, decreasing the need to spray with fungicides to avoid diseases, including *Botrytis*, also known as grey mould (Durner, 2018). Claire et al. (2018) mentioned that the conditions under the polythene tunnel encourage the development of powdery mildew. They also create an environment suitable for natural pest control in which a pest is used to attack another in an enclosed area. For example, these predatory insects control spider mites, aphids, slugs and thrips, reducing reliance on insecticides (McCartney and Lefsrud, 2018).

1.5. Fertigation systems

Fertigation is extensively practised in commercial horticulture by applying nutrients (fertilizer) in a liquid form to strawberries via an irrigation system (Smith, 2017). The fertigation system is generally used on high-value crops, including turf, vegetables, fruit trees and ornamentals (Nestb and Guery, 2017). It allows a timely water supply through drip irrigation, less wasteful than sprinklers with an accurate nutrient (fertilizer) application rate, improving crop nutrient uptake (Durner, 2018). Strawberries are grown in coir (figure 1.4), biodegradable, have an excellent capacity for aeration and water retention and have no fungus (Robinson Boyer et al., 2016).



Figure 1.4. Strawberry plants are grown in coir bags. Each bag is placed on raised beds and provided with drip irrigation systems for providing crops with water and nutrients.

Nitrogen (N_2) is the most commonly used plant nutrient (Hobbie, 2015). Naturally, occurring Nitrogen is a diatomic molecule that most plants cannot consume; therefore, the nitrogen used must be a component of other chemical substances that can be taken up by the plant (Singh et al., 2018). Generally, anhydrous ammonia, ammonium nitrate and urea are used as bioavailable forms of nitrogen (Kennedy, 2016). Other nutrients applied through the fertigation systems include monoammonium phosphate, diammonium phosphate in bioavailable forms and bioavailable sources of silicon (Artyszak, 2018; Gallegos-Cedillo et al., 2018). In fertigation, adjusting the potential of Hydrogen (pH) of irrigation water is essential as it allows optimal uptake of nutrients. Almanac (2020) mentioned that strawberries require a pH between 5.5 and 7.0 (figure 1.5) (slightly acidic to neutral) however, growers recommend a pH between 5.5 and 6.5, ideal for growing strawberries. The pH and EC levels alter the availability of essential nutrients in growing media (Hafiagroup, 2020). These pH levels (5.5 and 6.5) are required in the fertigation streams as very few UK strawberries are grown in soil. Moreover, pH levels above 8 can harm certain strawberry nutrients (such as iron levels in specific cultivars) (Hafiagroup, 2020). The availability of nutrients at varying pH in growing media is shown in figure 1.5.

4.0 pH	4.5 I	5.0	5. I	56	.0 6 I	.5 7 I	.0 7 I	7.5 8 I	3.0 8	3.5 I	9.0 I	9.5 I	10.(I
			Ac	idic					Alk	aline		-	
	I STRON	I IGLY		MEDIUM	SLIGHTLY	VERY SLIGHTLY	VERY SLIGHTLY	SLIGHTLY	MEDIUM		STRONG		
						NITR	DGEN						
						PHOSP	HORUS				-		
						POTA	SSIUM						
						SUL	FUR						
						CAL	l CIUM						
						MAGN	FSILIM						
												T	٦
						IR	ON			<u> </u>	+	+	-
						MANG	ANESE					+	-
						BOR							
													٦
					С	OPPER	AND ZIN	C		-		+	-
						MOLYB	DENUM	1					

Figure 1.5. Availability of nutrients at varying pH values. Source: Almanac (2020). The image shows pH ranges of crop nutrients that are under the category of acidic to alkaline.

Electro-conductivity (EC) is of great importance in measuring the solution's ability to conduct electrical current, which can factor between 0.65ds/m and 0.7ds/m depending on the nutrient formulation (Lyaruu, 2010). The unit of measurement for the electroconductivity of soil is deciSiemens per metre (dS/m) or milliSiemens per centimetre (mS/cm) (Corwin and Yemoto, 2019). The EC measures the quantities of salts available in the soil. Higher levels can negatively impact crop yield, suitability, nutrient availability and activity of soil microorganisms and is an indicator of soil health (Corwin and Scudiero, 2020).

Additional problems associated with high sodium salts are toxicity, poor soil structure and poor infiltration and drainage. Electroconductivity of soil is affected by crop planting, irrigation, land use and application of manure, fertilizer and compost. Irrigation water salinity should also be measured as irrigating in amounts too low or too high in salts enables salts to accumulate in the root zone, thereby increasing EC (Almanac, 2020). An EC reading lesser than 1 dS/m in the soil is considered non-saline and does not affect most crops and soil microbial processes however, an EC reading more significant than 1 dS/m is considered saline and can affect critical microbial processes of soil. These include the production of nitrous and other N oxide gases, nitrogen cycling, respiration and decomposition, nitrogen losses and an increase in the populations of plant-parasitic nematodes (Corwin and Scudiero, 2020).

1.6. Coir bags on raised soil beds under polythene tunnels

Raised soil beds are ideal for strawberries, which benefit from good drainage. Raised beds (figure 1.6) are made of coir bags or troughs on raised soil, which lifts the fruits off the soil and allows easy picking (Goodchild et al., 2018). Raised beds prevent grasses from invading the planting area. Using coir on raised beds eradicates the need for soil sterilization from soil-borne diseases and pests, an essential part of growing crops in soil beds (Nichols, 2013). Fleece is laid over crops in March after planting to boost the flowering, protect crops from harsh climate conditions such as heavy rainfall and frost and remove in April (Hall, Jin and Dodgson, 2016). Raised soil bed strawberries are in figure 1.6.



Figure 1.6. Raised soil bed strawberries

Fleece is usually spread over crops to encourage early flowering and keep crops away from frost. Refer to figure 1.7 to see fleece pulled over crops. (fleece is removed in figure 1.6).

1.7. Fleece and mulch

1.7.1. Fleece

Fleece and other floating films, known collectively as crop cover, are laid over plants hastening their growth and protecting against weather and pests (figure 1.7) (Marshall et al., 2018). Growers use these materials, which provide a small but significant amount of protection from cold or windy weather (Hall and Jin, 2016). Fleece-covered beds are shown in figure 1.7.



Figure 1.7. Fleece-covered strawberry beds under a polythene tunnel.

These fleeces are lightweight unwoven polypropylene fabric that allows filtered light to pass through them. The fleece also filters air movement, reducing wind chill, providing shade, and holding warm air. On fields, the fleece is pulled (like a blanket) over the crops from the beginning to the end of the field and held down with weight bags. Weights (as shown in figure 1.7 above) are placed on the edge of beds to secure the fleece from lifting with high winds. These weights are bags made from heavy-duty polyethene fabric, half filled with gravel, stone, pebbles or sand, with no specific standard measurement.

1.7.2. Mulch

In climates with cold winters, mulch is spread over strawberry plants to protect the plant from cold and extreme temperature fluctuation (Bano and Qureshi, 2016). Generally, mulch inhibits weeds from growing and protects plants from frost (Júnior et al., 2018). Plants show growth when the mulch is removed after the danger of frost is over (Singh et al., 2019). Figure 1.8 below are strawberry beds covered with fleece under a tunnel before the polythene is put over the tunnel hoops.



Figure 1. 8. Fleece-covered strawberries in spring (before the polythene is put on). Fleece is put over the crop for frost protection.

1.8. Tabletops

The use of tabletop growing systems (figure 1.9) adopted by commercial strawberry growers is rising (Daugaard, 2008). The plants are usually grown under polythene tunnels in plastic troughs approximately 1.5 metres above the ground containing compost and watered by trickle irrigation (Downing, 2016).



Figure 1 9. Tabletop strawberries. Source: (Palmers, 2018). The photo shows strawberries growing on tabletops on a field. During the fruiting season, fruits will hang down from the plants, as shown in figure 1.9, allowing easy picking compared to the crops growing on the ground.

Strawberry growers continue to invest in tabletop systems despite their high cost as they reduce picking costs by 30 to 40 per cent. Grimstad and From (2017) mentioned that the tabletop growing system not only creates ease of strawberry picking but makes strawberry picking a more attractive job as labour is increasingly difficult to obtain and manage. Tabletop strawberries decrease the amount of rot and poor harvest decay from *Botrytis* (De Preter, Anthonis and De Baerdemaeker, 2018).

1.9. Strawberry plant

Strawberries are low-growing herbaceous plants from the Rosaceae family with a fibrous root system and a crown, which arise compound leaves typically with three (saw-tooth edges) leaflets (figure 1.10 and table 1.5) (Stewart, 2016). The strawberry plant produces flowers on slender stalks known as the peduncle. As the strawberry plant ages, the root system becomes woody, sending out runners, thus enlarging the plant vegetatively (Visser and Konings, 2018). The strawberry fruit (figure 1.11 and table 1.5) is an accessory fruit consisting of a greatly enlarged flower receptacle with achenes commonly known as strawberry seeds (Nelson et al., 2018). An achene is a separate single-seeded fruit that does not open to release the seed (Ariza et al., 2016). The strawberries, *Fragaria L.*, are native to the temperate regions of the Northern Hemisphere, and cultivated varieties are grown worldwide (Kumar, 2018).

1.10. Strawberry plant breeding

Strawberry breeding was developed in England since the 1800s, and numerous varieties such as 'British Queen' (1840), 'Noble' (1884) and 'Jucunda' (1854) were introduced and were famous for their rich flavour or their resistance to cold and disease (Stewart, 2016). *Fragaria x ananassa* has developed into a large and tasty berry after years of hybridization and dominates modern strawberry cultivation (Jacobs, 1957; Darrow, 1966). The genus *Fragaria* has a primary chromosome number of (x=7) (Ichijima, 1926) and has four main groups, such as the diploids (2n=2x=14) chromosome number 14, which includes the model species for the genus, *F. vesca*, the tetraploids (2n=4x=28), including *Fragaria Orientalis*; the single hexaploid species *Fragaria moschata* (2n=6x=42); and four octoploid species (2n=8x=56): *F. chiloensis*, *F. iturupensis*, *F. virginiana* and the hybrid cultivated strawberry, *F. ananassa* (the *Fragaria ananassa* and *Fragaria vesca* are both diploids (2n=2x=14)) (Aristya et al., 2015).

Crossbreeding techniques have been established since the 20th century and are a traditional method of plant breeding (Folta, 2019). Crossbreeding involves allowing specifically chosen plants to sexually reproduce with other plants (Simpson, 2018). In other words, the crossbreeding techniques involve using plants with favourable characteristics (Schaart et al., 2016). Their offspring are raised, and a decision is made on a selection of plants with the best traits, and the process proceeds to the next generation (Nelson et al., 2018).

In crossbreeding, the pollen of the chosen plant is rubbed over the female receptacle of the other plant to achieve thorough pollination (Hancock, 2018). The strawberry plant is made up of five anatomical structures, which include the leaf (composed of 3 or 5 leaflets), root, crown, stolon and daughter plant (known as the runner) (Garcia, 2016). This structure is shown in figure 1.10 and explained in table 1.5.


Figure 1.10. Anatomy of a strawberry plant. Source: Strawberryplants (2020). The fully established (mature) strawberry plant consists of woody/ fibrous roots (with secondary and primary roots), a crown from, which the leaves (with 3 leaflets) arise, flowers (not shown in the picture) and runners, also known as daughter plants.



Figure 1.11. Anatomy of the strawberry fruit. (a) Strawberry plants in halves (b) Structure of the strawberry fruit and (c) Tertiary and secondary fruits of the strawberry plant.

Although strawberry plants can self-pollinate, the hoverfly (figure 1.11c) is one of several strawberry plant pollinators, including butterflies and wild bees such as bumblebees (Hodgkiss, Brown and Fountain, 2018; Kim et al., 2019). Driscoll's varieties of the strawberry plant became the most popular in Europe (Jublieestrawberries, 2020). Some of Driscoll's unique varieties include Driscolls Jubilee, Driscolls ElizabethTM and Driscolls Zara, which are preferred for their distinctive ruby colour and flavoursome taste (Jublieestrawberries, 2020). The anatomy of a strawberry plant is classified into 6 main parts: the root, crown, leaf, flower, fruit and runner, summarised in table 1.5.

Plant structure	Characteristics			
Crown	Consists of the central crown and branch (figures 1.10 and			
	1.14). The central crown is the plant's body, which holds			
	the leaves, runners, branch crown and inflorescences that			
	arise from it.			
Leaf	In spring, new leaves grow from the crown and replace			
	older leaves. Production of leaves stops when			
	temperatures fall to 0°C in autumn. A well-established leaf			
	canopy can supply the energy to initiate flower buds.			
	(figures 1.10 and 1.14).			
Root	Strawberries are sensitive to deficiency or excess water			
	and high salts in the soil as they have shallow root systems.			
	Most of its roots are produced in spring and autumn.			
	Primary roots conduct water and nutrients to the crown,			
	while feeder roots are generally for water and nutrient			
	absorption. (figures 1.10 and 1.14).			
Runner	Runners form during long days with warm temperatures,			
	beginning in late spring and continuing until autumn.			
	(figure 1.10).			
Flower	The primary flower opens first and yields the largest fruit			
	but is more susceptible to frost than flowers formed later.			
	The secondary flowers normally open 1 to 2 days after the			
	primary flower, followed by tertiary flowers. (figure 1.12).			
Fruit	The strawberry 'fruit' is an enlarged receptacle with seeds			
	(achenes) embedded in the surface. Each achene holds a			
	seed. Primary berries are the largest and first to ripen,			
	followed by secondary and tertiary fruits. (figures 1.11).			
1				

Table 1.5. Strawberry plant anatomy and morphology

Images shown in figures 1. 12, and 1.13 below are from a glasshouse experiment showing flowers of strawberries and strawberry runners during their life cycle. Figures



1.14 and 1.15 show roots from a glasshouse experiment highlighting the parts of plants and the root system.

Figure 1.12. Strawberry plant flowers in a pot glasshouse experiment. (photo taken when automatic lights in the glasshouse were switched on).



Figure 1.13. Malling Centenary strawberry runners attached to mother plants (pot compost) in UH glasshouse experiment.

As shown in figure 1.14. and 1.15, the strawberry plant consists of both secondary (white) and primary (dark) roots. The secondary (young) roots, also known as feeder roots, are of great importance to the plants as they carry absorbed water and nutrients

from the soil or Hoagland's solution into the crown of the plant, as well as the primary (mature) roots, which act as a channel for the transportation of water and nutrients from the crown throughout the plant. The strawberry's primary roots are the main roots of the plant and can survive for multiple years; however, the secondary roots are short-lived and have a lifespan of days to weeks (Darrow, 1966).



Figure 1.14. The appearance of a strawberry plant grown in hydroponics Hoagland's solution.

Figure 1.15. The appearance of strawberry plants 'roots' grown in a hydroponic Hoagland solution

1.11. Harvest and shelf life

Gibberellins are known to enhance the growth of plants and aid in the ripening of fruits (Ogas, 2000). Strawberries usually ripen 28-30 days after full bloom, and not all berries will ripen simultaneously (Poling, 2016). The ideal time of the day to pick (harvest) strawberries is in the early morning hours when the strawberries are still cool before the heat builds up in the fruits altering the quality (Anjom, Vovgioukas and Slaughter, 2018).

Strawberry pickers pick strawberries by grasping the peduncle between the forefinger and thumbnail while pulling and twisting simultaneously to allow the fruit to roll onto the palm of a hand with about one-quarter of the peduncle still attached as they are delicate and can be bruised easily. Unblemished strawberries will last longer and store better than bruised strawberries (Maurstad, 2018). Harvested strawberries are placed in punnets with a capacity of 250 to 450g away from sunlight (Sharma and Singh, 2019). Harvested strawberries are moved as quickly as possible to the cold room to preserve their freshness (Morris and Sistrunk, 2018). The significant fungi causing fruit decay include *Botrytis sp*, *Mucor sp* and *Rhizopus sp* (Siedliska et al., 2018). Strawberries maintain their freshness for 3 to 7 days in refrigeration at 4°C. After that, the quality of the fruit diminishes (Lamb and Chuah, 2018).

1.12. Pest and diseases of strawberries

In the UK, the major diseases are grey moulds such as *Botrytis cinerea* and powdery mildew (Hall, Jin and Dodgson 2016). The fungal infection caused by *Botrytis cinerea's* first symptoms appears as a grey mould at the end of the stalk of strawberries. In strawberries, the area of infection will enlarge, and a sunken brownish grey area will develop in the middle of the mould (Hill, Henshall and Beresford, 2017). *Botrytis cinerea* (figure 1.16a) affects fully ripe strawberries but can also infect green fruits and eventually cover the whole of each affected fruit (Llanos and Apaza, 2018).

Botrytis cinerea fruit rot is considered a significant challenge for strawberry growers, and when conditions favour disease development, losses from fruit rot can exceed 50% (Vorotnikova, VanSickle and Borisova, 2012). In the UK, *Botrytis* disease infection is the second most significant cause of crop losses to the horticulture sector by declining harvest yield and marketability with a cost estimated at £54 million to the industry (Abbey et al., 2019).

Other diseases of strawberries include red core disease, crown rot and *verticillium* wilt. The red core disease is caused by the soil-borne fungus *Phytophthora fragariae* (figure 1.16b). In an average year, it is estimated that the loss to the industry from red core *Phytophthora fragariae* is in the region of £3 to £4 million, and even though preventative measures are applied to most crops, these figures are doubled in bad years (Cross, Fitzgerald and Down, 2005). *Phytophthora fragariae* affects the root system of strawberry plants (Adams, 2019). The first symptoms of the red core disease are usually seen in the late spring during wet conditions when healthy crops resume growth and infected plants remain stunted (Parikka et al., 2016). Plants with the red core disease may die as the season progresses and become subject to increasing stress levels during high temperatures (Fry, 2012).

Crown rot (*Phytophthora cactorum*) (figure 1.16c) of strawberries is a soil-inhabiting pathogen that occurs in strawberries planted in poorly drained, over irrigated soil and during long periods of rain in warmer climates (Toljamo et al., 2016). Early symptoms

of the disease can be found in stunting growth and wilting of young leaves of strawberry plants. The potential loss of plants to crown rot can reach 20 to 30 %, and in 2016, 90,000 tonnes of strawberries were sold in the UK season, with a market value of £386 million. If 25% of plant loss occurs in the UK due to crown rot, the volume of fruit could be reduced by up to 22,500 tonnes, representing a value of £96 million (Xiangming, 2018).

Verticillium wilt in strawberries is caused by *Verticillium dahlia* (figure 1.16d), and symptoms include stunted growth, delayed development and yellowing of lower leaves (Washburn, 2018). As the disease progresses, the older leaves wilt and dry up, while the younger central leaves of the plant remain green until the plant dies and all foliage turns brown (Wu et al., 2019). Depending on the pathogen population in the soil and susceptibility of the cultivar, losses can vary between 5 to 90%. If 25% of plant losses occur in the UK due to *Verticillium* wilt, it will represent lost revenue of £29 million; however, techniques and measures implemented to tackle *Verticillium* wilt could save such potential losses (Xiangming, 2018).

Strawberry powdery mildew (*Podosphaera aphanis*) is a significant cause of crop yield losses (Hall, Jin and Dodgson, 2016). Strawberry powdery mildew can infect all parts of strawberry plants, including leaves, petioles, stolons, flowers and supporting stems (Maas, 1998). Pests may become problematic for strawberry plants during their development. The most common strawberry pest includes spider mites, *Tetranychus urticae*, aphids *Chaetosiphon fragaefoilii*, white fly *Aleyrodidae* and slugs *Gastropoda* (Nellist, 2018). Spider mites *Tetranychus* urticae attack strawberry plants by puncturing the plant cells to feed (Autunović, 2018). The aphids *Chaetosiphon fragaefoilii* are common sap-sucking insects which can cause a distortion in plant growth and transmit plant viruses to strawberries (HE and FU, 2018).

The white fly *Aleyrodidae* typically feeds on the under part of strawberry plant leaves and slugs *Gastropoda*, which attack tender strawberry fruits by forming holes that some insects may use to further damage (Castle, Grass and Westphal, 2019). An invasive pest of strawberries is the spotted wing drosophila SWD (*Drosophila suzukii*), a fruit fly that originated in Japan and has spread worldwide (Hennig and Mazzi, 2018). The spotted wing drosophila spread to the USA, then mainland Europe and then arrived in the United Kingdom in 2012 (Farnsworth et al.,2017). The invasive pest, similar to several species of fruit and vinegar flies, is now found in the UK (Rice et al., 2017). The adult males have a distinctive spot on each wing, while females have a saw-like appendage (ovipositor) used to pierce developing fruits (Hoskins, 2018). The female SWD lays eggs under the surface of the fruit's skin, which hatches into larvae that

contaminate the fruit and feed on the flesh resulting in fruit collapse (Graham and Brennan, 2018).



Figure 1.16. Diseases of strawberry plants. (a) *Botrytis cinerea* (fruit rot) (b) *Phytophthora fragariae* (red core) (c) *Phytophthora cactorum* (crown rot) (d) *Verticillium dahlia* (verticillium wilt). Source: Alchetron (2020); Pnwhandbooks.org (2020).

1.13. *Erysiphales* (Powdery mildews)

1.13.1. Introduction

Braun and Cook (2012) reported that there are over 650 species of *Erysiphales* (powdery mildew), and their order contains 1 family-*Ersiphaeae* and 16 genera. The *Erysiphales* are in the phylum Ascomycota (Takamatsu, 2013). They are obligate biotrophs (which obtain energy and nutrients from living cells) and infect almost every part of the plant (Spanu, 2012).

The term "powdery mildew" describes the appearance of whitish powder on infected plant parts as the conidiospores are produced in abundance on the surface of the host plants (Glawe, 2008). As this group of fungi cannot be grown on culture or artificial media, their taxonomy has been mostly based on morphological characteristics such as

ascocarps and ascospore morphology, conidial germination patterns, mycelial characteristics, appressoria morphology, and host range, among others (Meeboon and Takamatsu, 2015; Meeboon, Hidayat and Takamatsu, 2016). Nowadays, by using molecular techniques such as DNA sequencing and light and scanning electron microscopy, scientists can study the origin, distribution and migration of many species from this group in detail (Meeboon & Takamatsu, 2017). Figure 1.17 represents a phylogeny of tribes and genera of *Erysiphales*. The phylogeny is based on the 28S rDNA data set for 40 taxa of the *Erysiphales*, covering all known tribes and an outgroup taxon (Braun et al., 2006; Meeboon and Takamatsu, 2017).



1.13.2. General classification of *Erysiphales*

Figure 1.17. Phylogeny of tribes and genera of *Erysiphales*. Adapted from Braun, 2011 and Meeboon and Takamatsu, 2017

1.13.3. Infection

The process of infection of powdery mildew is initiated immediately after an ascospore or conidiospores lands on the surface of a host plant (Saharan et al., 2019; Glawe, 2008). The spore germinates and forms a germ tube that elongates at the tip to build an appressorium, from which hypha develops to produce the penetration peg (Glawe, 2006; Kabaktepe et al., 2017; Meeboon, Kokaew and Takamatsu, 2018). The penetration peg is used by the fungus to break the cuticle cell wall of the host and to

penetrate the epidermal cell through both enzymatic and mechanical pressure (Khodaparast, 2016). After penetration of the host plant, the fungus develops a haustorium, which is a feeding structure to supply the fungus with nutrients while the host cell is kept intact (Scholler et al., 2016).

1.13.4. The general life cycle of *Erysiphales*

The general life cycle of *Erysiphales* (powdery mildews) involves two stages asexual reproduction and sexual reproduction stage (Tulek and Canpolat, 2016). The asexual stage typically begins with the conidiospores landing on the host plant's surface, germinating, developing appressoria and penetrating the host cell wall (Braun, 1995). Conidiospores of powdery mildews are single-celled, colourless, and uninucleated and are usually produced in singles or chains (Tulek and Canpolat, 2016; Zhang et al., 2016).

The conidiospore germination can occur in a dry atmosphere as they are fully hydrated but may require relatively high humidity (Glawe, 2008). The conditions suitable are temperatures between 60 to 80°F (15 to 27°C) and moderate to high humidity, respectively (Hall, Jin and Dodgson, 2016). The dispersal of the conidium from the conidiospores (figure 1.18) depends on various factors such as electrostatic charges, wind, mechanical force and leaf shaking (Panstruga and Kuhn, 2019). The life cycle of strawberry powdery mildew is shown in figure 1.18.



Figure 1.18. The life cycle of strawberry powdery mildew (*Podosphaera aphanis*). Source: (Hall, Jin and Dodgson, 2016).

1.14. Strawberry powdery mildew (Podosphaera aphanis)

Podosphaera is a genus of the powdery mildew belonging to the tribe of Cystotheceae of Erysiphaceae (figure 1.17), and some other species of Podosphaera include; Podosphaera macularis, Podosphaera Balsaminae, Podosphaera euphorbiae, Podosphaera *clandestine*, Podosphaera pannosa, Podosphaera spiraeae, Podosphaera fuliginea, Podosphaera clandestine var. cvdoniae, Podosphaera leucotricha, Podosphaera mors-uvae, Podosphaera myrtillina, Podosphaera sp. Podosphaera major and Podosphaera fusca (Takamatsu et al., 2010). Strawberry powdery mildew, Podosphaera aphanis (figure 1.18), is the most important fungal disease of protected strawberry crops, negatively impacting strawberry production worldwide (Hall & Jin, 2016). The disease accounts for up 70% yield loss, which amounts to a market value of £56.8 million (Hall, Jin and Dodgson, 2016).

1.14.1. The life cycle of *Podosphaera aphanis*

The fungus develops from spore germination, mycelium development and spore production, which is influenced by temperature and humidity (table 1.6). The optimum temperatures are between $15 - 30^{\circ}$ C and relative humidity > 60% - 100% (Hall, Jin and Dodgson, 2016). Disease levels are mainly caused by the asexual conidiospores, which are similar genetically. At the same time, the sexual reproduction of *Podosphaera aphanis* leads to the formation of an ascus enclosed in a chasmothecium (figure 1.18) (Jin and Hall, 2012).

The mature chasmothecium contains one ascus and eight ascospores (figure 1.18). The chasmothecium forms multiple visible round black cases (ascocarps, which are genetically dissimilar in ascus and chasmothecia) on the lower surface of the leaf. The chasmothecia are the fungus's long-term survival structures involved in over-wintering the fungus on debris, green leaves and crop trash. The germination and spore production time for *Podosphaera aphanis* is 7-14 days under favourable conditions (Jin and Hall, 2012). The genetically similar asexual conidiospores are responsible for the disease level build-up, and reproduction begins from April to August. Jin and Hall (2012) explained that the chasmothecia (overwintering structures) are formed from August to April. Asalf et al. (2013) stated that the ascospores within the chasmothecium are formed by the sexual reproduction of the fungus.

Previous work has shown that the number of chasmothecia formed is determined by the level of infection during the summer season (Jin and Hall, 2012). They also mentioned that the percentage of mycelium on the leaf in the summer could significantly increase the number of chasmothecia formations during the autumn and winter season. An example is a percentage of mycelium (disease levels) found on leaves of strawberry plants in field experiments, which reached 30%, and levels of chasmothecia found were greater than 50% at the end of the strawberry growing season from September 2011 to 2012 (Jin and Hall, 2012).

1.14.2. Infection and symptoms

1.14.2.1. Infection

Hall, Jin and Dodgson (2016) pointed out that at the start of each season, it is important to consider the newly planted and over-wintered crops. In newly planted crops, the inoculum often develops during the process of propagation with the conidiospores and mycelium existing on plants at the time of delivery and planting (Oliveira, Braga and Rangel., 2018). In the case of over-wintered crops, the fungus survives within the crop through infected plants or as chasmothecia on debris, green leaves and dead leaves.

The fungus grows between the host cells, invading only a few cells to produce nutrientabsorbing structures known as Haustoria (Chowdhury, Coad and Little 2018). This enables the extraction of nutrients from the host plant through the hyphae penetrating the epidermal cells (Ellingham, 2017). After infection occurs, the leaves curl up, known as cupping (figure 1.19a). The leaf's surface is covered by white powdery patches known as the mycelium consisting of conidial chains, clusters of genetically similar asexual conidiospores (figure 1.19b), which spread from the lower to the upper surface of the leaf.

1.14.2.2. Symptoms

The disease displays a series of symptoms. Healthy strawberry leaves are naturally flat, but after infection, they begin to cup (figure 1.19a). However, other cupping causes could be a virus infection or pest infestation (Gaudery et al., 2010). Depending on the variety, blotches begin to form (figure 1.19b). Mycelium is formed (figure 1.19c), and the development of asexual conidiospores produces the powdery effect, which may be seen first on the lower than upper surfaces of the leaves and petiole. However, in some cases, the mycelium becomes visible first on the leaves, petioles and peduncles and spreads early to the fruit (Hall, Jin and Dodgson, 2016).



Figure 1.19. Symptoms of *Podosphaera aphanis* on strawberries. (a) Leaf cupping (b) Leaf blotching (c) Mycelium on leaves (d) Infected flower (e) Mycelium on unripe fruit, and (f) Mycelium on ripe fruit. Source: (Hall, Jin and Dodgson, 2016).

1.15. Epidemiology and the environment

1.15.1. Epidemiology and spread

The rate at which the disease spreads depend on the inoculum's presence and favourable environmental conditions, especially temperature and relative humidity. Table 1.6 summarises the conditions that affect the life cycle of *Podosphaera aphanis*. The

dispersal of the conidiospores depends on various factors such as electrostatic charges, wind, mechanical force, and leaf shaking (Stankevičiené, 2017). Conducive temperatures and relative humidity for powdery mildew development are shown in table 1.6.

Variable		Germination	Infection	Sporulation
Temperature (°C)	Minimum	2 - 5	5	13
	Optimum	15.5 - 30	18 - 30	20
	Maximum	30 - 35	30	35
Relative humidity (%)	Minimum	8 -12	No effect	No effect
	Optimum	97 -100	No effect	No effect
	Maximum	100	No effect	No effect
Presence of free water(immersionhours)		Up to 3	No effect	No effect
Time of day (hrs)	Minimum	No effect	No effect	20.00 - 8.00
	Maximum	No effect	No effect	12.00 - 16.00

Table 1.6. Conditions for strawberry powdery mildew conidia.

Source: (Hall, Jin and Dodgson, 2016)

Powdery mildew disease infection and spread can be managed by the frequent use of fungicide (sprays) application using a crop sprayer (tractor) (figures 1.20a and b). The crop sprayer (tractor) sprays fungicides over each strawberry bed as it moves through the field tunnel. See how fungicides treatments are sprayed over crops in figure 1.20b below.





Figure 1.20. Fungicide crop sprayer on Ladybird field.

1.16. Plant defence mechanisms

1.16.1. Constitutive defence

Plant cell walls are formidable barriers to infection, and most organisms can only cause disease when the barrier (plant cell walls) is breached (Boots and Best, 2018). Most plants invest heavily in thick cell walls and cutin or suberin layers (Fuchs and Krauss, 2018). The suberin is a lipophilic macromolecule found in plants when insulation or protection is required, and suberized cells form epidermis. This tissue envelops

secondary stems as part of the bark and develops as the healing tissue (sealing) after wounding or leaf abscission (Graça, 2015). Figure 1.21 shows the cross-section of a strawberry leaf displaying the components of the plant's defence mechanism. In the case of an invasion of a pathogen, as explained by Debona et al. (2017), the silicon deposited (accumulated) is laid down mainly in the cuticle, epidermis and palisade layer, thereby hindering the appressorium of a fungus breaking through the cuticle wall to form a penetration peg.



Figure 1.21. The cross-section of a strawberry leaf (UV light)

The constitutive defence mechanism is always present in the plant. It includes various preformed barriers such as cell walls, waxy epidermal cuticles and bark, which protect the plant from invasion and give it strength and rigidity (figure 1.21 and 1.22) (Pančić and Kiørboe, 2018). Nollet and Gutierrez-Uribe (2018) reported that compounds (phenols and quinones) were identified that explained the differential resistance of onion cultivars to the pathogen.

He explained that onion cultivars with outer scale leaves that were red or yellow were resistant to smudge caused by *Collectotrichum circinans*, while those with white scale leaves were susceptible. The usefulness of the constitutive defence mechanism is that if the infection is prevented through the constitutive defence pathway, there is no direct damage caused by the pathogen and little risk of damage due to this defence system (Boots and Best, 2018). The characteristics of plant defence mechanisms are described in figure 1.22.



Figure 1.22. Plant defences. Source: Binder and Parniske (2018). The figure shows that plant defence mechanism are divided into two main pathways. The passive or the constitutive and the active or the induced defence mechanism.

Figure 1.23 below explains how a plant's defence mechanism is enhanced, which plays a role in deterring pathogen penetration and where the silicon is deposited in the epidermal cell of leaves of rice plants. This diagram is explained in figure 1.23 a, b, c below.

Debona et al. (2017) showed in figure 1.23 (a) that silicon formed a physical barrier beneath the cuticle and cell wall, which explains the cause of the reduction in rice blast severity. He explained (figure 1.23(b)) that the dis-uniformity in silicon deposition underneath the cuticle wall allowed the penetration peg of the fungus *P. oryzae* of the rice plant. He also showed that the sizes of the rice blast region were larger (figure 1.23 (c)) due to unlimited colonization of the epidermal and mesophyll cells by fungal hyphae.



Figure 1.23. Silicon deposition in the epidermal cell of leaves of rice plants. Source: Debona et al. (2017).

Alhousari and Greger (2018) researched silicon and the mechanism of plant resistance to insect pests on rice plants. The work investigated found that bio-available silicon is deposited as a 2.5µm thick layer beneath the cuticle layer, forming a silicon-cuticle double layer in rice leaf blades. They mentioned that the abrasiveness of silicified wax,

cuticle, hairs and other tissues linked with plant protection, storage, support and strengthening leads to the irreversible wear of mouth parts while insects are feeding, therefore deterring chewing insects (Alhousari and Greger, 2018). Jin (2015) found that adding silicon to strawberries formed greater and thicker wax density on the adaxial leaf surface and reduced the disease levels of strawberry powdery mildew.

Silicon was shown to reduce disease levels of strawberry powdery mildew and increase the length and number of leaf hairs in strawberry plants (Fatema and Hall, 2012). Wang et al. (2017) explained that silicon accumulation in the epidermal tissue and thickening cell walls prevents pathogen penetration. He further explained that silicon-enhanced constitutive defence mechanism is associated with the density of silicified long and short epidermal cells, a thick layer of silica beneath the cuticle and thickness of the cell walls as physical barriers against pathogen penetration. Yang et al. (2017) showed that adding silicon to rice *Oryza sativa* improved the silicification of leaf sheaths that brown plant hoppers feed on, reducing the number of brown plant hoppers. Enhanced constitutive defence is believed to result from increased rigidity and reduced digestibility of plant tissue to pests through the addition and deposition of silicon in plants (Hernán et al., 2019).

1.16.2. Active defence

Defence against infectious disease arises from a complex set of interdependent mechanisms varying from mechanical barriers to the complex array of effectors in the immune system (Veloso and van Kan, 2018). The active (induced) plant defence is produced in reaction to infection, damage or stress caused by herbivores or diseases. It will only shorten the infectious period of the plant (Kupfer and Fessler, 2018). Nearly all living plants can detect invading pathogens and react with inducible defences such as pathogen-degrading enzymes, production of toxic chemicals and deliberate cell suicide (Pančić and Kiørboe, 2018). The active (induced) defence response often waits until a pathogen invasion before producing chemicals or defence-related proteins due to the high energy and nutrient requirements associated with their production and maintenance. Wang et al. (2017) found in a recent study that silicon, concerning the active defence pathway, acts by specifically building up in the area of infection to prevent it from further damage to the plant. Yang et al. (2017) published a study on silicon amendment is involved in the induction of plant defence response to a phloem feeder. It demonstrated that silicon amendment initiates plant resistance to herbivores through the plant's constitutive and active defence mechanism.

1.16.3. Silicon (Si)

In 1824, Silicon (Si) was first isolated and described as the seventh most abundant element in the universe and also known as the second most abundant element on the earth by Swedish chemist Jon Jacob Berzelius (Epstein, 1994). Silicon is in all plants (leaf hairs) and has various benefits. Silicon has an atomic number of 14 and is

generally in the form of ordinary sand; however, it can be found as rock crystals, quartz, amethyst, flint, opal, jasper and agate (Fauteux et al., 2005).

Silicon in a bioavailable form, is absorbed by most plants, conifers, sugar cane, ferns, moss, rice, wheat, potatoes, cassava, soybeans, sugar beet, barley, tomato and corn (Laing et al., 2015). The accumulated concentration can vary from 1% to above 10% (Datnoff and Rodrigues, 2007). Silicon accumulators accumulate high levels (>1.5%) of silicon in their tissue, "intermediate" accumulate moderate levels (0.5-1.5%), and non-accumulators have low levels (<0.5%) of silicon (Alhousari and Greger, 2018). Silicon accumulators are known to be able to take up and deposit silicon. Jin et al. (2012) maintains that although the strawberry plant is considered a non-accumulator of silicon, if the concentration of bioavailable silicon is greater than 1% of dry weight, then the plant is classified as a silicon accumulator. Plants absorb silicon through silicon transporter genes in the roots (Ouellette et al., 2017).

1.17. The effects of silicon on plant diseases

According to Ma and Yamaji (2006), silicon is a plant nutrient not considered essential for the plant. However, it has a positive impact on decreasing disease intensity. Today, bioavailable silicon reduces the powdery mildew of strawberries, apples, roses, cucumbers, wheat, peas, melon, soybean, banana, tomatoes etc. (Rodrigues et al., 2015; Wang et al., 2017). The work conducted in this thesis investigated silicon use in different root applications for three consecutive years from the start to the end of the growing season. Fatema and Hall (2012) identified silicon to have a positive impact in reducing the disease levels of strawberry powdery mildew and increasing the length and number of leaf hairs on the upper and lower surface of the strawberry leaf (Fatema and Hall, 2012).

Silicon acts on the plants by modifying the wax and cuticle and building up the plant's defence mechanism in response to infection (Jin and Hall, 2012). Some other positive effects of silicon include increased °Brix levels, where Jin (2015) showed that silicon nutrients significantly increased °Brix values in strawberry leaves and ripe fruits. Liu (2016) also showed a significant increase in °Brix of leaf petioles with silicon. °Brix is the total amount of sugar content in any aqueous solution determined by the refractometer (chapter 2, section 2.5.1). The average strawberry fruit °Brix reads 10, with 14 being good and 16 excellent (Harrill, 1998). Growers test strawberries for °Brix (sweetness) to ensure that it meets the demands of supermarkets such as Tesco. Silicon application has also been shown to encourage rapid growth and improve the ability of plants to withstand harsh environmental conditions (Pärtel, 2016). Research conducted at the University of Hertfordshire has shown that silicon nutrient applied once and twice a week in a fertigation field trial at Maltmas farm Wisbech significantly reduced the severity of the disease (Liu, 2016). Work at the University also investigated the

utilisation of silicon nutrients with and without the addition of Potassium Carbonate (KHCO₃) in a tank mixture to reduce disease levels (Fatema and Hall, 2012).

A recent study aimed to investigate the level of disease control using Potassium Bicarbonate (K50) and silicon-based wetter, both alone and in combination (Jin and Hall, 2012). The result described the control of the disease level when K50 alone was less significant than the addition of silicon. However, silicon applied alone significantly reduced the number of germinating ascospores and colonies in the trial (Fatema and Hall, 2012). This has shown that silicon can significantly reduce the susceptibility to strawberry plant disease. Jin (2016) reported that a concentration of silicon and silicon nutrient with K50 foliar spray treatment inhibited the disease level build-up of *Podosphaera aphanis* compared to the untreated plants. Liu (2016) found that using a silicon nutrient reduced susceptibility to *Podosphaera aphanis* and two-spotted spider mites. Previous work carried out also showed that silicon nutrients may play a positive role in raising °Brix (measurement of total sugar content in liquids or juices) of strawberry leaf, and petiole, improving pollen viability and influencing the length of flower receptacle and stamens (Liu, 2016).

Tibbits (2018) showed a significant increase in the seed yield of wheat when silicon nutrient was added to a hydroponic solution. The work showed a visual difference in awn straightness when silicon was added. High silicon-treated wheat plants had straight awns, and treatment with lower silicon concentration had twisted awns. Tibbits (2018) showed that the density of trichomes (leaf hairs) on awns increased significantly, and leaves of high silicon-treated wheat plants were stiffer and did not bend easily, suggesting that silicon provided structural support to the leaves of wheat plants. Korkmaz et al. (2018) showed a positive effect of silicon on yield and fruit quality in tomato crops grown in a closed hydroponic system. Diseases known to be effectively reduced (in severity) by silicon are shown in tables 1.7 and 1.8. Most research is mainly shown in silicon accumulators.

Tables 1.7 and 1.8 include the different crops: silicon accumulators, intermediate accumulators and non-accumulators. Tables 1.7 and 1.8 below show that silicon positively affects many diseases, and a fungus causes all diseases mentioned below. It is also shown in tables 1.7 and 1.8 that disease reduction is recorded in both silicon accumulators and non-accumulators plants. Silicon reduces several powdery mildews, and brown rust of wheat is the only disease silicon does not affect (table 1.7).

Crop Species	Disease	Pathogen	Effect of silicon on disease control	Uptake of silicon	
Barley	Powdery mildew	Erysiphe graminis f. sp. hordei	Decreases	Silicon accumulators	
Barley	Black point	Alternaria spp.	Decreases	Silicon accumulators	
Corn	Stalk rot	Pythium aphanidermatum. Fusarium moniliforme	Decreases	Silicon accumulators	
Rice	Brown spot	Chochtiobolus miyabeanus	Decreases	Silicon accumulators	
Rice	Sheath blight	Thanatephorus cucumeris	Decreases	Silicon accumulators	
Rice	Leaf scald	Monographella abescens	Decreases	Silicon accumulators	
Rice	Stem rot	Magnaporthe salvinti	Decreases	Silicon accumulators	
Rice	Grain discolouration	Many fungal species	Decreases	Silicon accumulators	
Sorghum	Anthracnose	Colletotrichum graminicola	Decreases	Silicon accumulators	
Sugarcane	Rust	Puccinea melanocephala	No effect	Silicon accumulators	
Wheat	Powdery mildew	Blumeria gramminis	Decrease	Intermediate accumulators	
Wheat	Brown rust	Puccinia recondita	No effect	Intermediate accumulators	
Wheat	Foot rot	Fusarium spp.	Decreases	Intermediate accumulators	
Wheat	Leaf spot	Phaeosphaeria nodorum	Decreases	Intermediate accumulators	
Pea	Leaf spot	Mycosphaeralla pinodes	Decreases	Silicon accumulators	
Soybean	Stem canker	Diaporthe phaseolorum	Decreases	Silicon accumulators	

Table 1.7. Effects of silicon on disease severity

Crop Species	Disease	Pathogen	Effect on disease control	Uptake of silicon
Strawberry	Powdery mildew Pestalotia leaf spot Anthracnose fruit rot	Podosphaera aphanis Pestalotia longisetula Colletotrichum acutatum	Decreases	Non-accumulator
Cucumber	Powdery mildew Leaf spot Gray mould rot Crown and root rot Fusarium wilt	Sphaerotheca xanthii Corynespora citrllina Botrytis cinerea Pythium ultimum p.aphanidermatum Fusarium oxysporum f.sp. Cucumerinum	erotheca xanthii Decreases nespora citrllina vtis cinerea ium ultimum hanidermatum rium oxysporum f.sp. umerinum	
Coffee	Coffee leaf rust	Hemileia vastatrix	Decreases	Silicon accumulators
Grape	Powdery mildew	Uncinula necator Decreases		Intermediate accumulators
Melon	Powdery mildew Pink rot Fusarium	Podosphaera xanthii Trichothecium roseum Fusarium semitectm	Decreases	Intermediate accumulators
Lettuce	Pythium root rot	Pythium spp. Decreases		Silicon accumulators
Yellow passion fruit	Bacterial spot	Xanthomonas axnopodis pv. passiflorae	Decreases	Silicon accumulators
Tomato	Powdery mildew	Oidium neolicopersici Pseudomonas syringae pv. Tomato	Decreases	Non-accumulators
Pumpkin	Powdery mildew	Sphaerotheca xanthii	Decreases	Intermediate accumulators
Rose	Powdery mildew	Sphaerotheca pannosa	Decreases	Intermediate accumulators
Zucchini squash	Powdery mildew	Podosphaera xanthii	Decreases	Silicon accumulators
Peach	Brown rot	Monilinia fructicola	<i>onilinia fructicola</i> Decreases Internaccun	
Rye	Powdery mildew	Erysiphe graminis	Decreases	Silicon accumulators

Table 1.8. Effects of silicon on disease severity continued

Chapter 2 - General material and methods for all chapters

This chapter covers i) A general introduction to the field experiment site used in this research project – Maltmas Farm, Wisbech; ii) A summary of general methods used for field, laboratory and glasshouse experiments.

2.1. Farm description

Experimental field experiments were established at a commercial strawberry-growing farm in Maltmas, Wisbech, Cambridgeshire, PE14 0HS (figure 2.1). The growers Harriet & Henry Duncalfe manage the County Council farm as a family business with 86 hectares. The farm site is commercially used for soft fruit production; 14 hectares are dedicated to strawberry production, 19 hectares for raspberry production and other arable crops, including wheat, 42 hectares, and oilseed rape, 24 hectares.

At Maltmas farm, strawberries are commercially grown in coir bags on raised beds under polythene tunnels (chapter 1, figure 1.7). The polythene tunnels are designed to protect crops from unfavourable climatic conditions. Fleeces and mulch (chapter 1, figure 1.6) are also used early in the season under the tunnels to stimulate early flowering and prevent plants from being damaged by frost. The farm uses a fertigation system to apply fertilizers, silicon nutrients and water to the crops. The Malling Centenary plants (June bearers), which are more susceptible to strawberry powdery mildew *Podosphaera aphanis*, were planted and established in the Ladybird field tunnel in the 2016 silicon fertigation field experiment (figure 2.3). Amesti plants (Ever bearers) more resistant to strawberry powdery mildew *Podosphaera aphanis* were planted and established in the Amelia field tunnel for the 2017 and 2018 silicon fertigation field experiments (figure 2.3). Figure 2.2 is the strawberry pack house located at Maltmas farm, and the legend below includes the process followed after strawberries are picked straight from the fields to the pack house.



Figure 2.1. (a) Map location of Maltmas farm, Wisbech, Cambridgeshire. Source: Google Maps (2018). (b) Map location of Maltmas farm, Wisbech, Cambridgeshire. Source: Google Maps (2018).



Figure 2. 2. heat sealed with the plastic film machine; e) Sealed strawberry punnets are separated; f) Labels with information on strawberry variety, classification and producer are requested by supermarkets; g) Strawberry punnets stored in cold room waiting to be transported to supermarkets. Source: Liu (2016).

2.1.1. Description of field experiments in 2016, 2017 and 2018

The silicon fertigation field experiments were carried out from 2016 to 2018 (table 2.1). Descriptions of all field experiments are listed below:

• Silicon fertigation field experiments (chapter 3): 2016 Ladybird Field, 2017 and 2018 Amelia Field

The silicon fertigation field experiments were established in the Ladybird tunnel in 2016 and the Amelia field in 2017 and 2018 (figure 2.3). The farm map with fields highlighted is in Appendices 7 and 8.

2.1.2. Effects of the use of silicon on disease levels of powdery mildew on strawberries: 2016 Ladybird field and 2017 to 2018 Amelia field

The silicon fertigation field experiment in the Ladybird field was carried out between 19 April and 9 August 2016, the silicon fertigation field experiment in 2017 was carried out between 9 April and 27 August 2017 and the 2018 silicon fertigation field experiment was carried out between 24 April and 21 September 2018. All silicon fertigation field experiments were set up in polythene tunnels. All polythene tunnels (Ladybird and Amelia field) were 180 metres long. A summary of all field experiments carried out at Maltmas Farm, Wisbech, between 2016 and 2018 is shown in table 2.1.

Table 2.1. A summary of all field experiments carried out at Maltmas Farm, Wisbech, between 2016 and 2018.

Year	Experiment	Cultivation	Area	Strawberry	Summary of
	field	type	/ha	variety	experiment
2016	Ladybird (figure 2.3)	Tunnel	1	Malling Centenary (June bearers) More susceptible to <i>Podosphaera</i> <i>aphanis</i>	 The effect of silicon nutrients on disease levels of strawberry powdery mildew (<i>Podosphaera aphanis</i>)- Disease assessments Silicon contents in leaves and fertigation water Silicon extraction from strawberry leaves
2017	Amelia (figure 2.3)	Tunnel	2	Amesti (Ever bearers) More resistant to <i>Podosphaera</i> <i>aphanis</i>	 The effect of silicon nutrients on disease levels of strawberry powdery mildew (<i>Podosphaera aphanis</i>)- Disease assessments Silicon contents in leaves and fertigation water Silicon deposition in the fertigation field experiment °Brix measurements
2018	Amelia (figure 2.3)	Tunnel	2	Amesti (Ever bearers) More resistant to <i>Podosphaera</i> <i>aphanis</i>	 The effect of silicon nutrients on disease levels of strawberry powdery mildew (<i>Podosphaera aphanis</i>) Disease assessments Silicon contents in leaves and fertigation water Silicon deposition in fertigation field experiments



Figure 2.3. The Ladybird (a) and Amelia silicon field (b) experiment plan 2016 to 2018. Ladybird tunnel (180 metres) contains 6 treatments, while the Amelia tunnel (180 metres) contains 4 treatments. Silicon is applied through the fertigation tubes (drippers), and fungicides are applied through a crop sprayer (figure 1.20).

Treatments used in both field experiments (figure 2.3a and b) were set up in line with the fertigation system to enable commercial purposes such as strawberry picking during harvest season. The silicon nutrient used was 'Sirius,' a bioavailable silicon form containing 1-10% Polyether-modified polysiloxane, 0.5 - 1% Ethyl alcohol (ethanol), 70 - 80% Tetraethyl silicate and 10 - 20% Alkyloxypolyethyleneoxyethanol. For the Sirius composition, see Appendix 5. The fertigation system consists of tubes, which provide water and silicon to the crops. This design plan was established to enable the fertigation system work where some treatments on the field had silicon applied once and twice weekly (in 2016) while in the others, no silicon was applied in half the field and silicon applied in the other half of the field (2017 and 2018). However, limitations to this design containing adjacent treatments are that it can encourage the spread of disease across the block of treatments, thereby allowing contamination through the fields.

• Field experiment methods

This general material and methods chapter includes methods that apply to several chapters. In contrast, specific methods are provided in the appropriate chapters for reducing levels of disease in chapter 3, the silicon deposition experiment in chapter 4 and the strawberry growth parameters experiment in chapter 5. Field experiment methods mainly consist of; field sampling of strawberry leaves (fortnightly) for strawberry powdery mildew disease assessment, fruit and leaf petiole sampling (once a month) for °Brix measurements, silicon extractions from strawberry leaves on the field

(once a month), silicon contents measurements in fertigation field water, silicon deposition assessments in strawberry plants in a glasshouse experiment in 2017 and field experiments in 2017 and 2018, measurements of growth parameters in untreated and silicon-treated strawberry plants in a hydroponic glasshouse experiment.

2.2. Leaf sample collection, storage and assessments

2.2.1. Leaf sampling for disease assessments

Cupping is the first sign of strawberry leaves infected with *Podosphaera aphanis* (chapter 1, figure 1.19a). Fully infected leaves can be observed with white mycelium on and below the surface of the leaves and petioles (chapter 1, figure 1.19c). Fully expanded leaves showing disease symptoms (cupping or mycelium development on the surface) were sampled by clipping the petioles of the plants with scissors and placing them in zipped polythene sample bags. Leaves were sampled randomly following a zig-zag motion from different plants in beds from each treatment in the silicon fertigation field experiments.

Strawberry leaves were sampled fortnightly from April to August from the 2016 silicon fertigation field experiment. On each sample day, 15 leaves were sampled (per bed) in replicates of 5 beds, such as beds a, b, c, d and e. Therefore, 75 leaves were sampled per treatment (such as 15 untreated leaves from 5 strawberry beds, for example), and 450 leaves were sampled per sample day. In the 2017 silicon fertigation field experiment (April to August), 15 leaves were sampled (per bed) in replicates of 4 beds. Therefore, 60 leaves were sampled per treatment and 240 per sample day. In the 2018 silicon fertigation field experiment (April to September), 15 leaves were sampled (per bed) in replicates of 4 beds. There were 60 leaves sampled per treatment and 240 on each sample day. The total number of strawberry leaves sampled varied between the 2016 to 2018 fertigation field as 5 beds were used (as replicates) in the 2016 field experiment, and only 4 beds (as replicates) were sampled from and used in the 2017 and 2018 field experiments.

After sampling, leaves were put in polythene sample bags labelled with the sample date, treatment and bed, respectively. Sample bags were then transferred into labelled plastic storage boxes and transported back to the cold room in the Science Building (at a constant temperature of 4°C) located in the College Lane Campus, University of Hertfordshire, Hatfield, Hertfordshire, AL10 9AB. The samples stored in the cold room were then assessed the following day for percentage (%) of mycelium cover using the strawberry powdery mildew assessment key (appendix 4) in preparation for further analysis. All 15 leaves per treatment bed (single leaf with 3 leaflets attached) were assessed using a dissection microscope (GX microscopes, 1x & 3x objectives and 10x eyepieces magnification, GT Vision Ltd, Suffolk, CO108LY). Fortnightly leaf samples were taken from the 2016 Ladybird field experiment (figure 2.3), 2017 Amelia field experiment and 2018 Amelia field experiment silicon fertigation field

experiments (figure 2.3). The disease levels assessment key was based on a revised strawberry powdery mildew assessment key developed by Xiaolei Jin (Jin, 2015) (appendix 4).

• Leaf sampling for silicon extraction

In the 2016 to 2018 silicon fertigation field experiments (chapter 3), strawberry plants from the fertigation fields used for disease assessment (section 2.2.1) were also used for silicon extraction to measure silicon contents in strawberry leaves after silicon nutrient treatment. Twenty leaves (sampled separately) were placed in foil boats selected from each strawberry bed. For example, in the untreated, 20 leaves from bed a, 20 leaves from bed b, 20 leaves from bed c, 20 leaves from bed d and 20 leaves from bed e etc. Strawberry leaves for silicon extraction were sampled only once a month from April to September (2016 to 2018).

2.3. Silicon extraction

The Autoclave Induced Digestion (AID) method was used for silicon extractions (Epstein, 1994). The extraction of silicon was carried out once on each sample month. The extraction process followed these steps: Twenty leaves were sampled once a month for silicon extraction and replicated by five strawberry beds per treatment (for example, 20 leaves x 5 beds). The leaves were placed in each foil boat labelled (with treatment name and bed) and oven dried at 60°C for 48 hours. Dried samples of leaves were crushed to powder form using a porcelain pestle and mortar.

100mg (0.1g) of ground leaves samples (from each treatment) were weighed and placed into autoclave-resistant polyethene bottles and labelled according to treatments. Five replications per sample were carried out for consistency. The leaf samples were moistened with 3mls of 50% H_20_2 , followed by the addition of 3mls of 50% N_aOH . The bottles were covered individually with loose-fitting caps and gently mixed using a vortex machine. Mixed samples were placed in large autoclave beakers, wrapped with foil and labelled before being sent into the autoclave at 138 Pka at 120°C for one hour (Taber, Shogren and Lu, 2002). After one hour, samples were allowed to cool down and for sediments to settle before being transferred into labelled 15mL polycarbonate tubes (flacons).

One millilitre of clear dissolved content was drawn from each polyethene bottle using a pipette (P1000) and brought to a final volume of 5ml with deionized (DI) water. Silicon extractions were carried out monthly from April to September in the 2016 to 2018 field experiment. In 2016, each treatment used for silicon extraction had 5 samples (beds) replicates. For example, there were 5 Untreated beds, 5 No silicon with fungicides beds, 5 silicon once a week with fungicides beds, 5 silicon once a week without fungicides beds, and silicon twice a week with fungicides beds and 5 silicon twice without fungicides beds. In 2017 and 2018, each treatment used for silicon extractions had 4 samples (beds) replicates. For example, there were 4 untreated beds, 4 fungicides-only beds, 4 silicon twice a week with fungicides beds and 4 silicon twice without fungicides. Replicates differed between years (2016 to 2018) as the number of beds and treatments in each field differed.

The following procedures were carried out to measure the silicon in the above-diluted samples.

- i) To each sample, 0.75mL of 2.5% boric acid was added. The tubes were then gently mixed with a vortexer. (pipette tips were replaced, and pipette swabbed with 70% industrial methylated spirit (IMS) to prevent contamination).
- ii) Ammonium molybdate (0.25 mL; 54 g/l, pH 7.0) was added to the sample, vortexed and left to stand for 5 minutes.
- iii) After five minutes, 0.125mls of tartaric acid was added to the solution and vortexed.
- iv) Lastly, 0.125mL of reducing solution, which is a combination of two parts;
 A) 2g of N_aSO_{3 and} 0.4g of 1-amino-2-naphthol-4-sulfonic acid in 25ml DI water) and B (25g of NaHSO₃ in 200ml of DI water). Solutions A and B were combined and brought to 250ml final volume with de-ionized (DI) water.
- v) The prepared solution was vortexed and left to stand for 30 minutes before measuring using a spectrophotometer.
- vi) The spectrophotometer (CECIL 1021, 1000 series, Cambridge, CB24 6AZ) reading optical density (OD) was set at 650nm. According to their treatment, one millilitre of the prepared solution was transferred from the polycarbonate tube (Falcons) to the semi-micro non-UV cuvettes.
- vii) Semi-micro non-UV cuvettes containing 1mL of each solution were placed vertically into the spectrophotometer to read the absorbance one at a time. (spectra was at zero (blank) before absorbance readings were taken).

• Silicon standard curve

A silicon calibration curve indicated each sample's true amount of silicon. For the standard measure, 15 of 15mL Falcon tubes were used, and the concentration used for the standard curve was 250mg of 99% silicate powder (25mg/mL). The silicate powder was diluted into 10 mL of deionized water, which was used as a known silicon concentration. The solution was then subsequently diluted into series of concentration of 0.000078, 0.00015, 0.00031, 0.00062, 0.00125, 0.0025, 0.005, 0.01, 0.02, 0.04, 0.08, 0.16, 0.31, 0.63, 1.25, 2.5 mg/mL. Find an example (presented below) of a silicon standard curve graph in figure 2.4 below. The points on the linear graph below were produced using the absorbance of the above-known concentrations above (mg/mL) (figure 2.4). The silicon concentration within a sample can be evaluated using the equation provided by the graph Y= 9.9868x.



Figure 2.4. Silicon concentration standard curve

Figure 2.4 is an example of a silicon standard curve. To calculate each sample concentration based on the standard curve, first find the concentration for each sample absorbance on the standard curve. Multiply the concentration by the dilution factor for each sample. The points on the silicon standard curve above are the absorbance of silicon on the Y axis (vertical) versus silicon concentrations on the X axis (horizontal).

2.4. Silicon content in fertigation water 2016, 2017 and 2018

2.4.1. Sampling

The fertigation water experiment was designed to determine the contents (amounts) of silicon in water flowing through the tubes at different set times when silicon treatments were applied and to understand if the water travelled the whole lengths of each field tunnel used for the experiment. The step-up of the fertigation experiment (2016 to 2018) was in the same experiment fields used for disease assessments. Water samples were collected from the same fields monthly and assessed in the laboratory. The fertigation system was used to allow the smooth transmission of water and silicon nutrients to the strawberry plants as there are insufficient nutrients in coir (which was used on the farm). The fertigation water flows through the drippers (figure 2.5) at 1.8 L per hour. Each burst of watering is 6 minutes. Normal irrigation (fertigation) session is up to 8 times a day depending on the weather example, 8 times in hot weather conditions and up to 6 times in cooler weather temperatures. In the 2016 field experiment, silicon was applied once and twice a week only for three and five minutes.

Sample points were from the near to far-end point of the silicon delivery system (figure 2.6 and 2.7). To replicate samples, four sample bottles were placed on each sample point for set times. For example, for 2 and 4 minutes, 2 bottles are placed on each bed side. Sampling was carried out once a month.

Water samples were collected at various set times and sample points (figure 2.7) to assess the differences in silicon accumulation between times and sample points. Before the sample day, plastic storage bottles were labelled according to sample times and points on the field. The collection was achieved by inserting drippers uprooted from coir bags into sample bottles, as shown in figure 2.5. After water sample collection, the bottles were placed in a sample basket and transported to the cold room at 4°C in the Science Building at the University of Hertfordshire for silicon contents in the fertigation water measurements. Figure 2.5 a to d shows the process of water sampling from the fertigation field.



Figure 2. 5. Fertigation water sampling from strawberry beds. (a) Sample bottle (b) Fertigation dripper in a coir bag (c) Fertigation dripper out of a coir bag (d) Fertigation dripper inserted in a sample bottle.

2.4.2. Measurement

The amount of silicon in each water sample bottle was evaluated using the same silicon titration method as in section 2.3, without using an autoclave. A silicon standard curve was used to determine silicon's (unknown) concentration in the water samples. The spectrophotometer reading optical density (OD) was set at 650nm to obtain each absorbance value. Samples were then transferred from polycarbonate tubes (Falcon) to the semi-micro non- u/v cuvettes according to their treatments. Before treatments were read, the spectra reading was brought to zero (0) using a blank solution (without silicon). Spectra was at zero before actual readings were taken. Cuvettes containing 1ml of each (titrated) water sample were placed vertically into the spectrophotometer one at a time, and readings were recorded respectively. Water and silicon were applied using a fertigation system (figure 2.6). The delivery system is monitored and timed for 5 minutes. See this system below.



Figure 2.6. Silicon delivery through the fertigation system. Silicon is added into the delivery system, which flows with the irrigation water through the irrigation pipes

Figure 2.7 below is the fertigation plan showing the sample points used in sampling water from the fertigation field.



Figure 2.7. Silicon in fertigation field water sample points on the fertigation field. The plans consist of the three sample points used in the fertigation field water experiment. The blue highlighted column are beds b and c out of 5 beds, which were the areas sampled from.

2.5. Laboratory experiment methods

Overall laboratory experiments covered in this thesis include leaf disease assessments for strawberry powdery mildew *Podosphaera aphanis*, silicon extraction from strawberry leaves, silicon contents in fertigation field water measurements and Degree Brix (°Brix) measurements on strawberry fruits and leaf petioles.

• Strawberry fruit and leaf petiole sampling for °Brix measurement

Strawberry fruits and leaf petioles were sampled for °Brix measurements in the 2017 silicon fertigation field experiment. See section 2.5.1 below. Strawberry leaves with long petioles attached were collected from the fertigation field for disease assessments, and petioles were separated in preparation for °Brix measurements.°Brix measurements were for 5 leaf petioles (replicated by 4 treatments), and 10 strawberries (replicated by 4 treatments) were taken monthly from the silicon field experiment in 2017.

2.5.1. °Brix Measurements

All °Brix measurements were carried out according to the standardized instructions provided in the user manual of the hand-held refractometer Eclipse (ThermoFisher Scientific. the UK). The device (figure 2.8b) is specially designed to measure plant juices' refractive index to estimate the sugar contents in plants. Crop juices' refractive index is calibrated in degree °Brix or per cent sucrose.

A °Brix refractometer (figure 2.8b) (Eclipse, Sugar % (°Brix): 0 to32) was used to measure the degree °Brix strawberry fruits and leaf petioles. Petioles were cut into small sections and then placed into a garlic crusher. Liquid from petioles was obtained using the garlic crusher (figure 2.8 a and b). The juices of the strawberry fruits were retrieved by cutting off the tips of the strawberries and gently squeezing the strawberries in a rotating motion. Juices collected from each fruit were transferred directly onto the reactive plastic screen of the refractometer (figure 2.8b). The lid is shut and held up to the eye to view through the optical lens. °Brix readings were recorded for each sample after viewing. The reactive plastic screen was wiped clean with deionised water each time a sample was read. Figure 2.8 below is a photograph of a garlic crusher (a) used in crushing leaf petioles and a refractometer (b) used in measuring sample juices.



Figure 2.8. (a) Garlic crusher (Left) and (b) refractometer (Right)


Figure 2.9. Strawberry °Brix measurements. (a) The refractometer and strawberry fruits; (b) The garlic crusher and a refractometer on a lab bench during sample measurements; (c) Blank reading of °Brix through the optical lens; (d) Readings of °Brix with strawberry juice reading 15 through the optical lens; (e) Readings of °Brix with strawberry juice reading 15 through the optical lens; (e) Readings of °Brix with strawberry juice reading 15 through the optical lens; (a) Readings of °Brix with strawberry juice reading 15 through the optical lens; (b) Readings of °Brix with strawberry juice reading 15 through the optical lens; (c) Readings of °Brix with strawberry juice reading 15 through the optical lens; (c) Readings of °Brix with strawberry juice reading 15 through the optical lens; (c) Readings of °Brix with strawberry juice reading 15 through the optical lens; (c) Readings of °Brix with strawberry juice reading 15 through the optical lens; (c) Readings of °Brix with strawberry juice reading 15 through the optical lens; (c) Readings of °Brix with strawberry juice reading 19 through the optical lens.

The number on the refractometer scale indicates the °Brix value (% mass sucrose) of the solution. The sweetness of plant fruit juices can vary, meaning their °Brix levels will read differently. As mentioned earlier, the average strawberry reads 10, with 14 being good and 16 excellent (Harrill, 1998). The refractometer displays a scale (figure 2.9 c to e) of 0 - 30 (%). This means a strawberry °Brix of 6 is poor (Harrill, 1998). Plant juices are measured on the percentage (%) at which liquids (fruit juices) bend through refraction. Figures 2.9 c to e above are visual presentations to show the different °Brix readings per plant juice.

2.6. Glasshouse experiment methods (pot and compost)

Glasshouse experiments were carried out in 2017 (compost) and 2018 (Hydroponics). Malling Centenary strawberry plants were potted in compost at the University of Hertfordshire, Hatfield College Lane campus glasshouse near the CP Snow building, AL10 9AB (figure 2.10 and figure 2.12) and the UH Bayfordbury Research Station, Lower Hatfield Road, Hertford, SG13 8LD (figure 2.11 and 2.12) for a silicon deposition experiment.



Figure 2.10. a and b: Location map of Bayfordbury Source: Google maps (2020)

The New Horizon multi-purpose compost and the Westland Garden Health Organic Chicken Manure Pellets were used for planting (potting) strawberry plants in both Hatfield and Bayfordbury glasshouses. Strawberry plants were watered manually using a plastic watering can before weekly treatments were added. The sensor watering system automatically detected that the mat and polythene felt were dry at the Hatfield campus glasshouse. Lights in the Hatfield glasshouse were 16 hours daily, and temperatures were kept at 25 °C daily and 8°C at night.



Figure 2.11. a and b Glasshouse experiment at the University of Hertfordshire, Hatfield. Automatic window vents in the glasshouse were opened when temperatures were above 25°C. At the Bayfordbury glasshouse, automatic watering was set for 30 minutes twice daily. The lights (ValuTekTM 400Wn Metal Halide Low Bay Light) and heating in the glasshouses were off during summertime, and the vents in the roof opened automatically when the temperatures were greater than 25°C. During the winter, the lights were on 12 hours per day and temperatures were kept at 25°C on days and 8°C at night to keep temperatures above freezing.



Figure 2.12. a and b Experiments at UH Bayfordbury glasshouse in 2017 (colour of glasshouse photos is caused by an automatic light in the glasshouse, which is programmed to switch on when the sun goes down).

In the 2017 UH Bayfordbury glasshouse experiment (figure 2.12 and 2.13) (pot and compost), 16 Malling Centenary strawberry plants were planted in plastic pots that were 12cm in diameter and 10cm deep (figure 2.13) and moved to the University of Hertfordshire, Hatfield glasshouse (figure 2.11). The plants were divided into two treatments, untreated (control) and treated (with silicon nutrient). All plants were randomly placed on the bench and were manually watered (along with the automatic watering) and then treated once weekly with silicon nutrient (treated) and deionised water (untreated). The duration of the 2017 glasshouse experiment was for eight weeks. Runners (daughter plants) from the Malling Centenary plants were planted in separate pots attached to their mother plants (figure 2.11) and detached when they were mature enough. Figure 2.13 below is a strawberry plant pot with compost from the glasshouse experiment showing how silicon treatments and de-ionized water were applied through the roots of strawberry plants via Gilson pipette tips (once weekly). See table 2.2 for treatments used.



Figure 2.13. Strawberry plant in the Bayfordbury glasshouse pot (compost) experiment. Silicon treatment and deionized water were added via a 50 ml syringe of 50 ml per plant pot via Gilson pipette tips, as shown in the figure.

Thirty bare root Malling Centenary strawberry plants were provided by Maltmas farm from a propagator, and 20 of those plants were then planted in a Hoagland's solution for a hydroponic experiment (to test for strawberry plants growth parameters). The remaining spare plants were potted with compost and chicken manure pellets on 24 January 2018. Bare roots strawberry plants are not planted in soil. They need 1500 hours of cold to recover from their dormant phase (Pestana et al., 2012). The strawberry plants were divided into two treatments. Ten treated (with silicon nutrient) and 10 untreated (control) were treated through a root application via Gilson pipette tips (figure 2.13) and monitored once weekly from January to June 2018. Table 2.2 shows experiments conducted in the glasshouse in 2017 and 2018.

Year	Location	Cultivar	Number	Treatments	Assessments
			of plants used		
2017	Hatfield Silicon deposition experiment (in pot and compost)	Malling Centenary (June bearers)	16	0.017% (v/v) silicon root application, 50 ml per pot. De-ionized water in untreated (control) root application, 50 ml per pot.	Silicon deposition in leaves, leaf petioles and roots. (see chapter 2, section 2.3.3).
2018	Bayfordbury Hydroponic experiment (Hoagland's solution)	Malling Centenary (Ever bearers)	20	0.017% (v/v) silicon root application, 50 ml per hydroponic tub De-ionized water in untreated (ccontrol) root application, 50 ml per pot.	Strawberry plants' growth parameters and quality concerning fruit quality between silicon-treated and untreated plants

 Table 2.2. Glasshouse experiments in 2017 and 2018

2.7. Data analysis

Statistical analysis was applied to data collected in the experimental chapters of this thesis. Firstly, data collected were tested for normality, and statistical tests were applied to understand the mean differences between silicon treatments in the silicon fertigation field experiment (effects of silicon on disease severity reduction) in 2016 and the relationship (correlation) between silicon extraction and disease levels in 2016 in chapter 3. Chapter 4 shows the quantification assessments and analysis of mean differences in cross-section fluorescence intensities of silicon-treated and untreated plants in 2 silicon deposition field experiments in 2017 and 2018. In chapter 5, statistical assessments were conducted to distinguish and understand the mean differences between growth parameters of silicon-treated plants and untreated plants from a glasshouse hydroponic experiment in 2018.

Raw data collected in all experiments were analysed using the tests mentioned below. Firstly, raw data were entered and stored in spreadsheets in Microsoft Excel (2016). Graphs provided in chapter 3 (disease assessments) for the silicon fertigation field experiment from 2016 to 2018 were also created using Microsoft Excel (2016). Statistical analysis throughout this thesis was done using both SPSS, version 26.0 and Microsoft Excel Data Analysis (2021). Before conducting any statistical tests, each data was tested for normality; to determine whether the data was standard or not normally distributed. Checking for normality of data used in chapters 3 to 5 was done using the Shapiro-Wilk test. Statistical tests used in the experimental chapters (3 to 5) were the parametric test: One and Two-way analysis of Variance (ANOVA), Paired Samples test, the nonparametric tests: Spearman's Rank Correlation, Wilcoxon Signed-Rank test and Kruskal-Wallis test. Mean comparisons were evaluated to calculate the differences between silicon-treated and untreated samples and to understand the significance P-value. The probability value (P-value) measures the probability that the difference in observation could have occurred just by random chance. Therefore, the greater the P value, the lower the statistical significance of the observed difference. If the P-value is below the 0.05 level, reject the null hypothesis.

The correlation coefficient or (r) examines the relationship between two quantitative variables. The Spearman's rank correlation was used to analyse the relationship between levels of disease in strawberry plants and amounts of silicon extracted in the 2016 silicon fertigation field experiment, chapter 3. Spearman's rank correlation is a statistical test investigating the degree to which two data sets are correlated and if there is any correlation. To apply Spearman's rank correlation, the researcher must have paired sets of data that are related (Akoglu, 2018).

The One-way ANOVA and Two-way ANOVA are statistical tests that explore the means between data sets to determine their differences. The One-way ANOVA and Two-way ANOVA were used to analyse fluorescence intensity data of the cross-section of leaves, petioles and roots of plants in a glasshouse and field experiments (chapter 4). The ImageJ multidimensional processing software was used to quantify fluorescence intensities between sample cross-sections examined in chapter 4 (Collins, 2007; Jensen, 2013; Bankhead, 2014). The Paired Sample test, nonparametric tests, Wilcoxon Signed-Rank test and Kruskal-Wallis test, were used to determine and calculate the mean differences between the silicon-treated and untreated plant growth parameters in the glasshouse hydroponic experiment. (chapter 5).

2.7.1. Comparisons of disease levels between treatment

The Area Under the Disease Progress Curve (AUDPC) measures disease intensity and resistance. This method works by estimating percentages of affected leaf areas recorded at different times during disease levels and indicating the level of suppression at the start of disease levels (Wu, 2016). (chapter 3, section 3.4). Results were obtained using the formula (Schandry, 2017).

$$AUDPC = \sum_{i=1}^{n-1} \frac{(x_{i+1} + x_i)}{2} (t_{i+1} - t_i)$$

Therefore,

X_i is a measure of disease severity at initial observation,

t is a measure of time,

n is the total number of observations.

Chapter 3 - The use of silicon nutrients in reducing disease levels of strawberry powdery mildew

This chapter covers i) An introduction to the use of a silicon nutrient in reducing susceptibility to strawberry powdery mildew *Podosphaera aphanis*; ii) Rationale, aim, hypothesis and objectives for the 2016 to 2018 silicon fertigation field experiment at Maltmas farm; iii) Material and methods for the 2016 to 2018 silicon fertigation field experiments; iv) Experiment results; v) Discussion and vi) Conclusion.

3.1. Introduction

The work conducted in this research was a continuation of previous work regarding the use of a bioavailable form of silicon in reducing powdery mildew of strawberries caused by *Podosphaera aphanis* by Jin (2015) and Liu (2016). However, this experiment investigated further using different cultivars, Malling Centenary and Amesti, (refer to table 2.1 for a description). A resistant cultivar (Amesti) was used to determine if silicon affected disease reduction as in a susceptible cultivar. Work by Jin (2015) and Liu (2016) showed that the addition of a silicon nutrient through a fertigation system in field experiments reduced strawberry powdery mildew *Podosphaera aphanis*. All strawberry plants used on the farm for the field experiments were chosen by the growers each year for commercial purposes, mainly for fruit production. The Malling Centenary cultivar was used in 2016 and Amesti in the 2017 and 2018 silicon fertigation field experiments. Growers are moving cultivation techniques and are now using all Ever bearer cultivars. Since moving from June bearers to Ever bearers at Maltmas farm, these Ever bearers crops were the crops available and provided for all experiments by the grower.

The cultivars Malling Centenary (June bearers and susceptible) and Amesti (Ever bearers and disease tolerant) were used to understand if silicon is effective in reducing disease levels, particularly in different cultivars, and also to determine if silicon was efficiently taken up by all cultivars, the differences in silicon uptake in these different cultivars, to determine whether applying silicon to strawberry plants twice a week is more beneficial than a once-a-week treatment.

3.2. Bioavailable silicon in plants

The general introduction to silicon in plants is provided in chapter 1, section 1.17; however, there is increasing evidence from research at the University of Hertfordshire that shows benefits from the application of silicon in reducing biotic and abiotic stresses (Fatema, 2014; Jin, 2015; Liu, 2016). Silicon in soil (pH 5.5 to 6) can only be absorbed by plants in the bioavailable form of silicic acid Si (OH)₄ (Scholey et al., 2018). The amounts of silicon in plants may vary from 0.1% to 10% of the plant's dry weight (Frew

et al., 2017). The silicon used is 'Sirius' (the main active ingredient is 70-80 Tetraethyl silicate: other compounds include silicon-based adjuvant and two other non-silicon compounds. Silicon concentration applied to plants in the silicon field experiment was at 0.017%. Fatema (2014) showed that bioavailable silicon increased the density of leaf hairs of strawberry plants in a glasshouse experiment. Work carried out by Jin (2015) showed that silicon increased cuticle thickness and wax formation in the adaxial surface of the leaves of the cultivar Shelley of strawberry plants and reduced powdery mildew disease levels. Liu (2016) also showed in the cultivar Sonata a reduction in strawberry powdery mildew disease levels and the number of two-spotted spider mites of strawberries with silicon in field experiments. The use of the silicon nutrient also showed an increase in the °Brix levels of strawberries (Liu, 2016).

3.2.1. Rationale

Previous work has shown that the application of silicon nutrient can cause a reduction in the levels of powdery mildew disease in strawberry plants cultivars such as Vibrant, Sonata, Shelley, Alexandra and Elegance when applied through the roots and sprays once a week in field experiments. The work reported here in chapter 3 explored the use of silicon on cultivars such as Malling centenary and Amesti applied through the roots of strawberry plants once and twice- a-week on field experiments to determine the effects and differences between the two treatments using different cultivars.

3.2.2. Aim

To investigate the effects of the silicon nutrient delivered through the fertigation system on the development of strawberry powdery mildew (*Podosphaera aphanis*) disease levels in different strawberry plant cultivars.

3.2.3. Hypothesis

Applying silicon nutrient applied once and twice a week through the fertigation system of strawberries can reduce the severity of *Podosphaera aphanis*.

3.2.4. Objectives

- Measure strawberry powdery mildew mycelium percentage cover on strawberry leaves treated with and without silicon nutrient. Disease assessments.
- Determine the relationship between the level of disease infection and the amount of silicon in strawberry plants from the silicon fertigation field experiment 2016.
- Compare the distribution of silicon in strawberry plants through silicon fertigation field experiments (2016 to 2018). Silicon extraction from strawberry leaves.

• Quantify the amount of silicon in the fertigation water from the silicon fertigation field experiments from 2016 to 2018. Silicon contents in water measurements.

3.3. Material and methods

Methods used in this chapter investigating the use of a silicon nutrient in reducing susceptibility to *Podosphaera aphanis* in the 2016 to 2018 fertigation field experiments include.

- i) Strawberry leaf sample collection and storage (chapter 2, section 2.2)
- ii) Disease assessments (chapter 2, section 2.2.1)
- iii) Silicon extraction from strawberry leaves from the silicon fertigation field experiment from 2016 to 2018. (chapter 2, section 2.3).
- iv) Silicon content measurements in fertigation water 2016 to 2018 (chapter 2, section 2.4 (i)).

3.3.1. Cultivars

Strawberry plants used in field experiments were planted in coir bags on raised soil beds and conducted under commercial conditions. The cultivars used were chosen by the growers each year. Malling Centenary (June bearer) cultivars were planted in March 2016 in the Ladybird field tunnel for the silicon fertigation field experiment. Amesti strawberry plants (Ever bearer) cultivars were planted in March 2017 in the Amelia field tunnel for the silicon fertigation field experiment 2017. Amesti strawberry plants were also planted in March 2018 in Amelia field tunnel for the silicon fertigation of Malling Centenary (June bearer) and Amesti (Ever bearer) cultivar strawberry plants are in chapter 1, table 1.2. As mentioned earlier, growers are moving cultivation techniques from growing Junebearing strawberries to growing all Everbearing strawberries. The grower chose these cultivars for the experiments in this chapter to determine if both cultivars acted similarly to silicon (disease reduction and silicon uptake).

3.3.2. Treatments

Six treatments (table 3.1) were applied to Malling Centenary strawberry plants in the silicon fertigation field experiment in 2016. All three fertigation field experiments were 180 metres long, including a central gap (un-sampled and untreated area) of 40 metres. All five strawberry beds from the silicon fertigation field experiment in 2016 were approximately 1 metre apart. In 2017 and 2018 (chapter 2, figure 2.3), four treatments were applied to four strawberry beds, also approximately 1 metre apart. The silicon twice weekly plus fungicides and fungicides only were 70 metres longer than other treatments because of the unusual commercial design. Sample replication is fully presented in chapter 2, section 2.2.1. Refer to appendix 9 to 12 for the list of fungicide

sprays used in the silicon fertigation field experiments from 2016 to 2018. In 2016 and 2017, the experiment ended in August, and in 2018, the experiment ended in September.

Table 3.1.	. Treatments	used in the	e silicon	fertigation	field	experiments	from	2016
to 2018								

Year	Treatments
2016 (Ladybird field) Malling Centenary Crops were planted in coir bags in March. Silicon and fungicides treatment started on 19 April 2016, and the experiment ended on 9 August 2016 (the end of the growing season).	 Untreated (no silicon nutrient and no fungicides No silicon nutrients + fungicides silicon once weekly + fungicides silicon once weekly without fungicides silicon twice weekly + fungicides silicon twice weekly + no fungicides
2017 (Amelia field) Amesti Crops were planted in coir bags in March 2017. Silicon and fungicides treatment started on 9 April 2017, and the experiment ended on 27 August 2017 (the end of the growing season).	 Untreated (no silicon nutrient and no fungicides) fungicides-only silicon twice weekly + fungicides silicon twice weekly + no fungicides
2018 (Amelia field) Amesti Crops were planted in coir bags in March 2018. Silicon and fungicides treatment started on 24 April 2018, and the experiment ended on 21 September 2018 (the end of the growing season).	 Untreated (no silicon nutrient and no fungicides) fungicides-only silicon twice weekly + fungicides silicon twice weekly + no fungicides

All field and glasshouse experiments conducted ended in 2018. There were no experiments carried out in 2019.

3.3.3. Leaf sampling

Strawberry leaves were sampled randomly (following a zig-zag pattern) every two weeks from each strawberry bed in all treatments from the silicon fertigation field experiment from 2016 to 2018. See chapter 2, section 2.2.1, for a general description of leaf sampling methods used in the silicon fertigation field experiment from 2016 to 2018.

3.3.4. Disease assessments: Strawberry powdery mildew (*Podosphaera aphanis*) on the strawberry leaf surface from 2016 to 2018

Fifteen strawberry leaves from 5 strawberry beds (75 leaves per treatment and 450 leaves in total per sample date) were sampled in 2016. Fifteen leaves per 4 strawberry beds (60 leaves per treatment and 240 leaves in total per sample date) in 2017 and 2018 were assessed for strawberry powdery mildew using the disease assessment key (appendix 4). The Area Under the Disease Progress Curve (AUDPC) was calculated to measure disease severity based on the mean percentage of powdery mildew (*Podosphaera aphanis*) coverage per strawberry leaf and sampling date (chapter 2, section 2.2.1).

The AUDPC measures the area under the disease progress curve. The formula for the AUDPC is in chapter 2, section 2.7.1. This method quantifies the size of the disease level based on the samples assessed throughout the season, such as the percentage cover of strawberry leaves (chapter 2, section 2.2.1(i)) and the severity of the disease under different treatments. Refer to chapter 2, section 2.7.1, for the formula of AUDPC. Data were first tested for normality using the Shapiro-Wilk test. The Parametric test, Paired samples test and Nonparametric test; Wilcoxon Signed-Rank test and Kruskal-Wallis test were used to evaluate the mean differences in the level of disease in the field experiment in 2016 (chapter 2, section 2.7).

3.3.5. Water sampling

Silicon content in the fertigation system water was measured to determine whether the silicon applied via the fertigation system was found throughout the fertigation water in the 2016 to 2018 field experiment. Water samples were collected and assessed every month from 2016 to 2018; all experiments ended when the crop season ended. Refer to chapter 2, section 2.4, for the method description used in this experiment.

3.3.6. Silicon extraction

Strawberry plants used for strawberry disease assessments in the silicon fertigation field experiments from 2016 to 2018 were also sampled for leaves used in silicon extractions. Refer to chapter 2, section 2.2.1, for the method description used.

3.4. Results

Figure 3.1 shows disease progress levels of Malling Centenary plants in the silicon fertigation field 2016 experiment from the start of silicon treatments in April to the end of the experiment in August 2016.

3.4.1. Disease levels of strawberry powdery mildew in the silicon fertigation field experiment 2016 (Malling Centenary)



Figure 3.1. Disease levels in the 2016 silicon fertigation field experiment.

The lines in the graph distinguish the different treatments. The vertical axis indicates the average percentage (%) of mycelium covered per leaf from each treatment. The horizontal axis indicates the date of sampling from each treatment. Silicon nutrient (Sirius 0.017v/v) was applied through the fertigation system once and twice weekly in this experiment. The AUDPC values of five treatments are displayed on the top left side of the graph. Silicon applied twice a week with fungicides (d) AUDPC, 375, and silicon applied twice a week without fungicides (e) AUDPC, 410 had the lowest disease levels. The highest disease levels (AUDPC, 3,423) were found in the untreated plot (a).

The effect of fungicides-only application on crops in the plot without silicon is not seen on this graph as there was an alteration of results due to flooded areas of field beds, which appears as if the fungicides treatment had a minimal reduction in disease levels. The flooding in the tunnel was an unexpected result of the field shown on the graph above (figure 3.1). This means the no silicon + fungicides treatment was flooded and excluded from the graph.

Typically, a disease level graph does not go down however, in figure 3.1, results show the silicon once-a-week + fungicides and silicon twice-a-week + fungicides treatments which may have been due to a sampling error. An example of sampling error is sampling leaves without disease towards the end of the experiment or season. The Shapiro Wilk test was used for normality and the parametric tests; Paired Samples test in SPSS version 26.0 were used to process and calculate the differences between treatment means from the 2016 silicon field experiment. The Paired Samples test, a parametric test, was used as the data collected from the 2016 field experiment disease assessments followed a normal distribution. The Paired Samples t-test compared the means of treatment variables. For example, untreated strawberry plants against silicon once-a-week treated crops or the untreated against fungicides-only treated crops to determine the differences in levels of disease between them.



Figure 3.2. Flooded Ladybird in 2016

Similarly, the AUDPC shows that the lowest disease level occurred in treatments with twice weekly silicon + fungicides (AUDPC = 410) and twice weekly silicon + no fungicides (AUDPC = 375). In contrast, the largest disease level occurred in untreated strawberry plants (AUDPC = 3,423). Statistical analysis is provided (appendix 14) using the paired samples t-test to find the differences between disease levels in the treatments from the silicon fertigation field experiment in 2016. The Shapiro-Wilk test was used to test for normality before applying statistical tests. Table 3.2 below is a summary table of findings from the disease levels graph 2016, shown in figure 3.1.

Table 3.2. Summary of findings for disease levels of strawberry powdery mildewin the silicon fertigation field experiment 2016 (Malling Centenary)

Year	Treatments	Statistical test	Statistical findings (see
		applied	appendix 14 for statistical
			workings)
2016	Untreated	Test for normality using	There were reduced levels of
	control (no	Shapiro-Wilk test and	disease in the silicon once-a-
	fungicides, no	average values of statistical	week + fungicides compared to
	silicon) and	differences were tested using	untreated control (no
	silicon once +	parametric test; t-test Paired	fungicides, no silicon)
	fungicides	samples test	treatment. There was a
			statistical difference between
			both treatments (P<0.05). The
			null hypothesis (H0) is
			rejected.
2016	Untreated	Test for normality using	There was no statistical
	control (no	Shapiro-Wilk test and	difference between disease
	fungicides, no	average values of statistical	levels of untreated control (no
	silicon) and	differences were tested using	fungicides, no silicon) and
	silicon once + no	parametric test; t-test Paired	silicon once-a-week + no
	fungicides	samples to test.	fungicides (P>0.05). The null
			hypothesis (H0) is accepted.
2016	Untreated	Test for normality using	There were reduced levels of
	control (no	Shapiro-Wilk test and	disease in the silicon twice-a-
	fungicides, no	average values of statistical	week + fungicides compared to
	silicon) and	differences were tested using	untreated control (no
	silicon twice +	parametric test; t-test Paired	tungicides, no silicon)
	fungicides	samples to test.	treatment (P<0.05). The null
			hypothesis (H0) is rejected.
2016	Untreated	Test for normality using	There were reduced levels of
2010	control (no	Shapiro-Wilk test and	disease in the silicon twice-a-
	fungicides no	average values of statistical	week + no fungicides compared
	silicon) and	differences were tested using	to untreated control (no
	silicon twice +	parametric test: t-test Paired	fungicides. no silicon
	no fungicides	samples to test.	treatment ($P < 0.05$). The null
	0	1 .	hypothesis (H0) is rejected.

In the 2016 silicon fertigation field experiment above (figure 3.1) and appendix 14, results showed a twice weekly application of silicon nutrient (Sirius 0.017% (v/v)) with fungicides and without fungicides had significantly (P<0.05) reduced susceptibility to *Podosphaera aphanis* throughout the experiment period. A parametric Paired Samples t-test showed that disease levels in the untreated were not statistically different (P>0.05) from the silicon once weekly without fungicides however, a reduction in disease levels and statistical difference (P<0.05) is seen in the silicon twice weekly + fungicides and silicon twice weekly + no fungicides compared to the untreated (no fungicides, no silicon) treatments. It is important to understand temperature and humidity can influence the growth and development of *Podosphaera aphanis* (Hall and Jin, 2016).

The paired t-test analysis conducted on means of disease levels between treatments in the 2016 silicon fertigation field experiment showed that there was a significant difference (P < 0.05) found between the untreated (control), the silicon once weekly + fungicides, silicon twice weekly + fungicides and silicon twice weekly + no fungicides treatment in the silicon fertigation field experiment in 2016. (figure 3.1 and table 3.2).

As mentioned previously, the flooded treatment was removed from the graph. The results from disease assessments suggested that disease intensities increased as temperatures increased due to flooding (figure 3.2). Flooding in the tunnel did not impact silicon levels in treatments containing silicon. However, the AUDPC results have confirmed that all four treatments with silicon nutrient showed lower disease levels of *Podosphaera aphanis*. In this case, figure 3.1 and table 3.2 above suggest that the aim that silicon can cause a reduction in strawberry powdery mildew disease levels was met in the 2016 silicon field experiment where treatments such as silicon once- a-week + fungicides, silicon twice a week and fungicides, and silicon twice a week without fungicides were applied, however not in treatments where the silicon once- a-week + no fungicides were applied. The bar chart in figure 3.3 below shows silicon extractions from the leaves of strawberry plants from the 2016 silicon field experiment (April to August).

3.5. Silicon content in strawberry leaves from the silicon fertigation field experiment 2016 (Malling Centenary)



Figure 3.3. Silicon extraction from strawberry leaves (μ g/mg) in the fertigation field in 2016 (at the end of the growing season in August)

Results from figure 3.3 show mean silicon extraction results at the end of the silicon fertigation field experiment season in 2016. The different colour bars indicate the different treatments used in this experiment. The error bars are standard error bars extracted from Excel (2022). The bar chart was produced from the mean silicon contents available (silicon accumulated) in strawberry plants at the end of the season. Each bar (figure 3.3) shown are replicated leaves. Silicon extraction was done from 20 leaves from each bed and each different treatment.

All leaves were sampled randomly through the length of each treatment bed. The Paired Two Samples for Means in Excel (version 16.43) was used to analyse the means of silicon extraction between treatments in the silicon fertigation field experiment in 2016. See table 3.3 below for a summary of the findings of silicon extracted from strawberry leaves from the silicon fertigation field experiment in 2016. Refer to appendix 17 for the statistical workings of silicon extraction between all treatments.

Year	Treatment	Statistical test applied	Statistical findings (see appendix 17 for statistical workings)
2016	Untreated control (no fungicides and no silicon) and silicon once + no fungicides	t- Test Paired sample for average values	P>0.05; there was no statistical difference between silicon levels in the untreated control (no fungicides and no silicon) compared with the once silicon once-a-week + no fungicides. The null hypothesis (H0) is accepted.
2016	Untreated control (no fungicides and no silicon) and silicon once + fungicides	t- Test Paired sample for average values	P>0.05; there was no statistical difference found between levels of silicon in untreated control (no fungicides and no silicon) compared with the silicon once- a-week + fungicides. The null hypothesis (H0) is accepted.
2016	Untreated control (no fungicides and no silicon) and silicon twice-a- week + fungicides	t- Test Paired sample for average values	P>0.05; there was no statistical differences found between levels of silicon in untreated control (no fungicides and no silicon) compared with the silicon twice- a-week + fungicides. The null hypothesis (H0) is accepted.
2016	Untreated control (no fungicides and no silicon) and silicon twice + no fungicides	t- Test Paired sample for average values	P<0.05; there was a statistical difference found between levels of silicon in untreated control (no fungicides and no silicon) compared with the silicon twice + no fungicides. The null hypothesis (H0) is rejected.

 Table 3.3. Summary of silicon extraction from strawberry leaves in the 2016 field experiment

Results from table 3.3 above exhibit a statistical increase (P<0.05) in the levels of silicon extracted from the leaves of plants treated with silicon twice-a-week without fungicides compared with the untreated control. Results showed no statistical increase

(P>0.05) between the levels of silicon extracted from the leaves of the silicon + fungicides, once silicon + no fungicides, silicon twice + fungicides compared to the untreated ones. The bar chart presented in figure 3.3 above shows the levels of silicon remaining in strawberry plants at the end of the season, and no account was taken of the age of strawberry leaves (young and old leaves); however, the experiment was revised and conducted in the following year 2017 where only same age leaves were sampled.

A Spearman's Rank correlation analysis between disease levels and amounts of silicon extracted from the 2016 field experiment revealed a moderate relationship between both variables with no significant difference (P>0.05). Refer to appendix 18 for the statistical workings.

Silicon contents in the fertigation water were carried out once a month and were sampled at 3 delivery points, and the results are shown in table 3.4.

3.5.1.1. Silicon content in fertigation water from the silicon fertigation field experiment 2016

The experiment (table 3.4) was set up in May 2016, and a pre-assessment (initial) was not conducted and examined; however, the experiment was revised to include a pre-assessment sample in the following month, June (table 3.4) and for the 2017 and 2018 fertigation field water experiment. The sampling time was also extended to the end of each season rather than only 2 months in 2016.

The experiment aimed to determine the amounts of silicon that travelled (at different lengths of time) throughout the fertigation field. These measurements were also conducted to show evidence that the silicon applied through the fertigation system is distributed throughout the tunnel. No silicon is present in the water used in fertigation (Liu, 2016). From table 3.4, the near silicon delivery indicates a sample point near the silicon irrigation system. Middle silicon delivery indicates the sample point in the centre of the 3 samples. The far end is the end point of silicon delivery (chapter 2, figure 2.8).

The pre-assessment in June (table 3.4) showed no silicon contents in the samples. This was pure irrigation water flowing through the fertigation tubes sampled first before silicon was added for measurements. Table 3.4 shows a range of results of silicon travelling the length of the 2016 tunnel; however, the expected readings of silicon at the concentration of 0.017% (0.025ml to 0.027ml of bioavailable silicon) added in the fertigation system should be around the same amount of silicon extracted 0.003mg/cm³. No measurements were taken in a steady state example, 1 hour or 24 hours after the application of silicon as the silicon remaining in the water would have been washed out

by the next fertigation event. The farm water contains no silicon, and the farm's reservoir combines drinking and rainwater as the groundwater is saline.

Sample point May	Time of sampling after	Silicon content (mg/ml)
2016	delivery of silicon/min	(averages of two samples) Mg/ml
Near silicon delivery	2	0.011 mg/ml
Near silicon delivery	4	0.016 mg/ml
Middle silicon delivery	2	0.015 mg/ml
Middle silicon delivery	4	0.022 mg/ml
Far-end silicon delivery	2	0.007 mg/ml
Far-end silicon delivery	4	0.019 mg/ml
Sample point June 2016	Time of sampling after delivery of silicon/min	Silicon content (averages of two samples) Mg/ml
Near silicon delivery	0 (i.e., before addition)	0.000
Near silicon delivery	2	0.010 mg/ml
Near silicon delivery	4	0.018 mg/ml
Middle silicon delivery	2	0.010 mg/ml
Middle silicon delivery	4	0.023 mg/ml
Far-end silicon delivery	2	0.015 mg/ml
Far-end silicon delivery	4	0.028 mg/ml

Table 3.4.	Silicon	concentration	(mg/ml) in	fertigation	field	water	2016
1 abic 5.4.	Sincon	concentration	(mg/mi	,	ici ugation	nciu	matti	2010

Moreover, the suggested rate for silicon nutrient used in agriculture is 50 - 100 mL in 100 mL in 100 - 600 L of water (Jin, 2015; Liu, 2016). The grower at Maltmas farm use 100 mL of silicon 'Sirius' in 600 L per hectare, which is calculated as:

 $100 \ mL \ in \ 600 \ L = 0.017\% \ Sirius \ (v/v)$

In 6 minutes, each irrigation tunnel received.

6 minutes / 60 x 2.2 L / hour x 4 drippers x 180 m x 15 beds = 1584 L water

Therefore, the amount of silicon (0.017% Sirius v/v) for one irrigation tunnel is:

 $0.00017 x 1584 L = 0.269 L \cong 270 mL$

Each strawberry plant received approximately 270 mL:

270 mL / 5000 - 5300 plants x 2 tunnels = 0.026 - 0.027 mL Sirius per plant

Sirius contains four different compounds (main active ingredient; 70 - 80 Tetraethyl silicate, silicon-based adjuvant and two other non-silicon compounds); therefore, the actual amount of soluble silicon that is taken up and remaining in the plants and water may be lower than the amounts calculated above (Jin, 2015; Liu, 2016).

3.6. Levels of disease in untreated (uncontrolled) strawberry plants in the silicon fertigation field experiment 2017 (Amesti)

Figure 3.4 is a disease assessment in the 'Amesti cultivar' in the silicon field experiment in 2017 (Amelia field). Disease assessments were carried out in the Amelia field from 9 April to 27 August 2017. Results in figure 3.4 show minimal disease levels in the untreated control plot only, and no disease levels occurred.



Figure 3.4. Disease levels in the 2017 silicon fertigation field experiment.

As there were deficient disease levels in the 2017 field experiment, figure 3.4 only shows the untreated control plot. No disease was found in all other treatments where silicon nutrient or fungicides were applied. No disease was detected in any other treatment (fungicides only, silicon twice + fungicides and silicon twice + no fungicides) in the silicon fertigation field experiment in 2017 as the cultivar Amesti, which is a more resistant cultivar and less susceptible to *Podosphaera aphanis* than Malling Centenary strawberry plants used in the 2016 fertigation field experiment. The growers specifically chose these cultivars. The experiment aimed to understand the effect of silicon in various cultivars besides moving to Ever bearer cultivars, as mentioned previously. A range of other cultivars grew on the farm; however, no Malling Centenary was grown in 2017 and 2018.

Also, the farm was experiencing lower disease pressures in 2017 than in 2016. The lower disease pressures experienced in the 2017 experiment were shown by the minimal amounts of disease only found in the untreated control of the field experiment, suggesting minimal pressures in disease intensities. In the 2016 field experiment, disease levels were higher, suggesting higher intensities of disease (pressures). Temperatures appeared to be less conducive to the development of fungus (*Podosphaera aphanis*), as shown by disease levels in 2017 compared to 2016. Conducive temperatures for powdery mildew fungus development are between 18 to 30°C (Hall and Jin, 2016). The measurement of temperatures under each field experiment was not stored for this experiment as the minimal amounts of disease seen throughout the experiment were not expected (2017 and 2018). Nonetheless, fungicide sprays continued in both experiments. For a list of fungicides used, see appendices 9 to 12.

3.6.1. Silicon content in the strawberry leaves from the silicon fertigation experiment 2017 (Amesti)

Silicon extractions from the leaves of strawberry plants from the 2017 silicon fertigation field experiment (April to August) are presented in the bar chart.



Figure 3.5. Silicon extraction from strawberry leaves ($\mu g/mg$) in the fertigation field in 2017 (at the end of the growing season in August)

Results from Figure 3.5 is a bar chart showing mean silicon extractions results at the end of the silicon fertigation field experiment season in 2017. The colours on the bars indicate the different treatments used in this experiment. The error bars are standard error bars. The bar chart was produced from silicon extraction results of untreated strawberry leaves and silicon-treated strawberry leaves.

The bar chart in figure 3.5 above shows means of 20 leaves per bed (the replicates are the 4 beds), where 20 leaves x 4 beds are 80 leaves per treatment. Refer to appendix 20 for statistical t-Tests Paired Sample for Means for silicon extraction comparisons (differences) between the untreated and silicon treatments in 2017. Table 3.5 summarises the findings of silicon extracted from strawberry leaves at the end of the silicon fertigation field experiment in 2017.

 Table 3.5. Summary table of silicon extraction from strawberry leaves in the silicon fertigation field experiment 2017

Year	Treatment	Statistical test applied	Statistical findings (see appendix 20 for statistical
			workings)
2017	Untreated control (no fungicides and no silicon) and fungicides-only	t- Test Paired sample for average values	P>0.05; no statistical differences were found between levels of silicon in untreated compared with the fungicides- only treatment. The null hypothesis (H0) is accepted
2017	Untreated control (no fungicides and no silicon) and silicon twice-a-week + fungicides	t- Test Paired sample for average values	P<0.05; there was a statistical difference found between levels of silicon in untreated compared with the silicon twice + fungicides. The null hypothesis (H0) is rejected.
2017	Untreated (no fungicides and no silicon) and silicon twice-a-week + no fungicides	t- Test Paired sample for average values	P<0.05; there was a statistical difference found between levels of silicon in untreated compared with the silicon twice-a-week + no fungicides. The null hypothesis (H0) is rejected.

Results from the 2017 field experiment have shown that there was an increase (P<0.05) in silicon contents extracted from the leaves in the silicon twice a week + fungicides and no fungicides compared to the untreated, but no differences (P>0.05) were found between the fungicides only treatment and the untreated ones. Silicon extraction measurements from the silicon fertigation water in the 2017 field experiment are shown in table 3.6 below. There were also three sample points, and the water sampled each month included a pre-assessment sample.

3.6.1.1. Silicon content in fertigation water from the silicon fertigation field experiment 2017

Sample point (May 2017)	Time of sampling after	Silicon content
	delivery of silicon /min	(mg/ml) (averages
		of two samples)
Near silicon delivery		
Near silicon delivery	3	0.024 mg/ml
Near silicon delivery	5	0.029 mg/ml
Middle silicon delivery	3	0.054 mg/ml
Middle silicon delivery	5	0.032 mg/ml
Far-end silicon delivery	3	0.001 mg/ml
Far-end silicon delivery	5	0.009 (mg/ml
Sample point (June 2017)	Time of sampling after	Silicon content
	delivery of silicon	(mg/ml) (averages
		of two samples)
Near silicon delivery	Pre-addition of silicon	0.000
Near silicon delivery	3	0.010 mg/ml
Near silicon delivery	5	0.002 mg/ml
Middle silicon delivery	3	0.006 mg/ml
Middle silicon delivery	5	0.011 mg/ml
Far-end silicon delivery	3	0.005 mg/ml
Far-end silicon delivery	5	0.022 mg/ml
Sample point (of July	Time of sampling after	Silicon content
2017)	delivery of silicon	(mg/ml) (averages
		of two samples)
Near silicon delivery		
Near silicon delivery	3	0.011 mg/ml
Near silicon delivery	5	0.023 mg/ml
Middle silicon delivery	3	0.009 mg/ml
Middle silicon delivery	5	0.018 mg/ml
Far-end silicon delivery	3	0.014 mg/ml
Far-end silicon delivery	5	0.030 mg/ml

Table 3.6. Silicon content (mg/ml) in fertigation field water 2017

Results from table 3.6 show that silicon was found throughout the length of the fertigation experiment in 2017 during silicon application. Although silicon contents vary between samples months, there were no abnormalities in the results presented in table 3.6. The water flowing through the fertigation drippers at 1.8 L per hour allows a burst of watering for 6 minutes; however, one of the aims of measuring the fertigation

water for silicon levels at these separate times was to determine whether the silicon added travelled the lengths of the tunnel during different set times.

3.7. Levels of disease in untreated (uncontrolled) strawberry plants in the silicon fertigation field experiment 2018 (Amesti)

Figure 3.6 is disease assessment also in the 'Amesti cultivar' in the silicon field experiment in 2018 (Amelia field). Disease assessments were carried out in the Amelia field from 24th April to 21st September 2018. Results in figure 3.6 show low disease levels in the untreated control plot only, and no disease levels occurred.



Figure 3.6. Disease levels in 2018 silicon fertigation field experiment.

Figure 3.6 also showed minimal levels of disease in the untreated control. However, no disease levels occurred in the 2018 silicon fertigation field experiment as the cultivar Amesti was also used (refer to section 3.6). No disease was found in the treatment plots where silicon nutrient was applied. This may be a result of using the same cultivar Amesti. The grower still applied fungicide sprays throughout the experiment, supported by a complete list of fungicide sprays used for the silicon fertigation field experiment in 2018 in appendix 9 to 12. Although no disease levels occurred in this experiment (figure 3.6). Silicon extractions from the leaves of strawberry plants from the 2018 silicon fertigation field experiment (April to September) are presented in the bar chart below.

3.7.1. Silicon content in the strawberry leaves from the silicon fertigation field 2018 (Amesti)



Figure 3.7. Silicon extraction from strawberry leaves ($\mu g/mg$) in the silicon fertigation field experiment 2018. (at the end of the growing season in September)

Figure 3.7 silicon extraction results proved that the plants take up the silicon nutrient delivered via the fertigation system. The different bars represent silicon extraction levels. They show that there is a larger quantity of silicon in the silicon twice weekly with and without fungicides compared to the untreated and fungicides-only treatment at the end of the experiment and season. Refer to appendix 22 for the statistical t-Test Paired Two Sample for means results of silicon extraction between the untreated and silicon-treated plants (untreated, fungicides only, silicon twice + fungicides and the silicon twice + no fungicides treatments) in the 2018 field experiment. A summary of findings of silicon extracted from strawberry leaves at the end of the silicon fertigation field experiment in 2018 is in table 3.7.

Table 3.7. Summary table of silicon extraction from strawberry leaves in thesilicon fertigation field experiment 2018

Year	Treatment	Statistical	Statistical findings (see
		test applied	appendix 22 for statistical
			workings)
2018	Untreated control (no fungicides and no silicon) and fungicides-only	t- Test Paired sample for average values	P>0.05; there was no statistical differences or increase found between levels of silicon in untreated control (no fungicides and no silicon) compared with the fungicides-only treatment. The null hypothesis (H0) is accepted.
2018	Untreated control (no fungicides and no silicon) and silicon twice-a-week + fungicides	t- Test Paired sample for average values	P<0.05; there was a statistical difference found between levels of silicon in untreated control (no fungicides and no silicon) compared with the silicon twice-a-week + fungicides. The null hypothesis (H0) is rejected.
2018	Untreated and silicon twice- a-week + no fungicides	t- Test Paired sample for average values	P>0.05; there was no statistical difference found between levels of silicon in untreated control (no fungicides and no silicon) compared with the silicon twice-a-week + no fungicides. The null hypothesis (H0) is accepted.

Results from the 2018 field experiment has shown that there was an increase (P<0.05) in the silicon contents extracted from strawberry leaves from the silicon twice + fungicides compared with the untreated. The results also showed that there were no statistical differences (P>0.05) found between the levels of silicon extracted from the leaves of the silicon twice + no fungicides and the untreated control. Silicon contents in the fertigation water for 2018 is presented in table 3.8 a and b below.

3.7.1.1. Silicon content (mg/ml) in water from the fertigation field experiment 2018

Silicon contents in the silicon fertigation water from the 2018 field experiment are presented in tables 3.8 a and b.

Sample point (as in figure	Time of sampling after	Silicon content
3.4) in May 2018	delivery of silicon/ min	(mg/ml) (averages
		of two samples)
Near delivery	Pre-addition of silicon	
Near delivery	3	0.010 mg/ml
Near delivery	5	0.002 mg/ml
Middle delivery	3	0.006 mg/ml
Middle delivery	5	0.011 mg/ml
Far-end delivery	3	0.005 mg/ml
Far-end delivery	5	0.022 mg/ml
Sample point (as in figure	Time of sampling after	Silicon content
3.4) in June 2018	delivery of silicon/ min	(mg/ml) (averages
		of two samples)
Near delivery	Pre-addition of silicon	
Near delivery	3	0.010 mg/ml
Near delivery	5	0.014 mg/ml
Middle delivery	3	0.009 mg/ml
Middle delivery	5	0.020 mg/ml
Far-end delivery	3	0.012 mg/ml
Far-end delivery	5	0.028 mg/ml
Sample point (as in figure	Time of sampling after	Silicon content
3.4) in July 2018	delivery of silicon/ min	(mg/ml) (averages
		of two samples)
Near delivery	Pre-addition of silicon	
Near delivery	3	0.014 mg/ml
Near delivery	5	0.013 mg/ml
Middle delivery	3	0.004 mg/ml
Middle delivery	5	0.017 mg/ml
Far-end delivery	3	0.005 mg/ml
Far-end delivery	5	0.008 mg/ml

Table 3.8a. Silicon content (mg/mL) in fertigation field water

Sample point (as in	Time of sampling after	Silicon content
figure 3.4) of	delivery of silicon	(mg/ml) (averages of
August		two samples)
Near delivery	Pre-addition of silicon	
Near delivery	3	0.018 mg/ml
Near delivery	5	0.026 mg/ml
Middle delivery	3	0.012 mg/ml
Middle delivery	5	0.014 mg/ml
Far-end delivery	3	0.017 mg/ml
Far-end delivery	5	0.021 mg/ml
Sample point (as in	Time of sampling after	Silicon content
figure 3.4) in June	delivery of silicon	(mg/ml) (averages of
2018		two samples)
Near delivery	Pre-addition of silicon	
Near delivery	3	0.016 mg/ml
Near delivery	5	0.021 mg/ml
Middle delivery	3	0.013 mg/ml
Middle delivery	5	0.032 mg/ml
Far-end delivery	3	0.029 mg/ml
Far-end delivery	5	0.031 mg/ml

Table 3.8b Silicon content (mg/mL) in fertigation field water

Results above in table 3.8 a and b show that the silicon nutrient delivered (via the fertigation system) is distributed throughout the field experiment. No samples were collected 24 hours after silicon was added to the fertigation system as the fertigation system did not run for 24 hours. Refer to chapter 2, section 2.4, for a full description and explanation of the fertigation experiment. The fertigation experiment aimed to determine the amounts of silicon (at different set times) that travelled the lengths of the tunnels and to show that the silicon delivered is distributed throughout the tunnel. The aim of this experiment was met as silicon was found throughout the fertigation fields in 2016 and 2018.

3.8. Discussion

3.8.1. The use of silicon nutrients reduced levels of strawberry powdery mildew *Podosphaera aphanis* in the silicon fertigation field experiment from 2016 to 2018

The experiment conducted in this chapter aimed to investigate further the work reported by Jin (2015), which investigated the use of silicon applied via sprays and roots through a fertigation system once a week only in a 2012 and 2013 (two years) field experiment on cultivars Elegance, Alexandra and Sonata. Liu (2016) showed silicon via root application in reducing the disease levels of strawberries on the cultivar Sonata.

The experiment reported in this chapter investigated in more detail, comparing silicon applied once- a-week with silicon applied twice- a-week treatment through a fertigation system, in reducing levels of strawberry powdery mildew using different cultivars Malling Centenary and Amesti cultivars for three complete growing seasons and measuring silicon contents in the leaves of strawberries and fertigation water for 3 consecutive years. Results from the silicon fertigation field experiment in 2016 showed that the strawberry crop from the silicon twice weekly plus fungicide, silicon twice weekly without fungicides and silicon nutrient once weekly plus fungicides had significantly (P<0.05) lower disease levels compared with the silicon nutrient once weekly without fungicides and the untreated (control) (figure 3.1). The disease levels graph (figure 3.1) revealed that the lowest disease levels occurred in the silicon twice weekly plus fungicides AUDPC (375) treatment. At the same time, the graph also showed that the most significant levels of disease occurred in the untreated (control) with the highest AUDPC (3,423).

A spearman's correlation analysis (appendix 18) was conducted in the silicon fertigation field experiment in 2016 to determine if the amount of silicon extracted had a relationship with disease infection levels and showed a moderate relationship between them. Using a silicon nutrient in the fertigation field experiment in 2016 showed similar results to the work done by Jin (2015). The work reported that the use of a silicon nutrient in a silicon fertigation field experiment in 2012 and 2013 caused a delay (more than 2 weeks) in the start of disease levels of strawberry powdery mildew on a strawberry field. Work done by Liu (2016) also found that silicon nutrient had significantly reduced the disease severity in treatments where silicon nutrient was applied, and the disease was significantly lower (P<0.05) in treatments where silicon nutrient had been applied.

Results from the 2017 and 2018 silicon fertigation field experiments have shown that no disease levels occurred; however, a small amount of disease was found only in the untreated control of strawberry plants in the silicon fertigation field experiments in 2017 and 2018. The farm experienced low disease pressures with high temperatures up

to and above 30°C, which is too high for disease development (Ouellette et al., 2017). Although no disease levels occurred in 2017 and 2018 field experiments, results from this thesis have shown that the addition of a silicon nutrient significantly (P<0.05) reduced the disease severity of *Podosphaera aphanis* in 2016, and similar results have been achieved in previous work done by Jin (2015) and Liu (2016). Jin (2015) showed in her thesis that the application of silicon via foliar and root 'once a week' can reduce the effects of powdery mildew. This chapter shows that silicon reduces powdery mildew in Malling Centenary plants treated once and twice a week via a fertigation system.

3.8.2. Silicon extraction from leaves of strawberry plants from the silicon fertigation field experiment 2016 to 2018

Previous preliminary work at the University of Hertfordshire by Jin (2015) and Liu (2016) has shown that the silicon delivered through (foliar and root) of strawberry plants 'once weekly' in field experiments contained more silicon compared with the untreated plants. Results from this experiment showed that in 2016, there was a statistical increase (P<0.05) found in the levels of silicon extracted from leaves silicon twice a week + no fungicides treatments compared to the untreated control and no statistical differences (P>0.05) in the silicon extracted from the leaves in the silicon once a week + fungicides, silicon once-a-week + no fungicides, silicon twice + fungicides compared to the untreated at the end of the experiment.

Results showed that in the 2017 field experiment, there was a statistical increase (P < 0.05) between the levels of silicon contents of the leaves of silicon-treated plants compared to the untreated control. In the 2018 field experiment, there was a statistical increase (P < 0.05) in the levels of silicon extracted from the leaves in the silicon twice + fungicides and the untreated ones. No differences (P > 0.05) were found in the silicon twice + no fungicides treatment and the untreated ones. Silicon extraction results did not show a trend in the amounts of silicon extracted between the 2016 to 2018 silicon fertigation field experiments. However, other factors could have played a role in the differences in the quantity of silicon remaining in plants after treatments.

3.8.3. Silicon contents in water from the silicon fertigation field experiment from 2016 to 2018

Silicon contents in fertigation water in the silicon fertigation field experiments were measured to show that the silicon delivered through the fertigation system is transported throughout the field. Although previous work by Liu (2015) and Jin (2016) has found that the silicon delivered is distributed in the fertigation water in a one-year experiment, the work reported in this chapter extended the experiment for 3 years consecutively. Conducting this experiment for three years allowed the observation for consistency in

the presence of silicon in the fertigation system after application. This experiment showed that the fertigation system always worked each year and can be a reliable means for transporting silicon, water and all other nutrients to crops. This work suggests that the fertility system's silicon is always transported throughout the fertigation field. Using a fertigation system, silicon nutrient are delivered through the roots of strawberry plants.

3.8.4. Agronomic differences between June bearers and Ever bearer strawberry plants

Malling Centenary strawberry plants were more susceptible to strawberry powdery mildew with high disease levels, as shown in the 2016 silicon fertigation field experiment graph (figure 3.1), while the Amesti strawberry plants used in the silicon fertigation field experiment in 2017 and 2018 were tolerant to the disease with minimal disease levels found only in the untreated plot of both fields. Temperatures (section 3.6) could have also contributed to less disease development in both fields; however, this thesis does not confirm the impact of these temperatures. This chapter suggests, in summary, that silicon can reduce disease levels of strawberry powdery mildew *Podosphaera aphanis* in 2016 (figure 3.1). Silicon was effectively absorbed by strawberry plants (silicon extractions from leaves), and silicon was always found throughout the length of the tunnel and in the fertigation water (silicon contents in fertigation water) in field experiments from 2016 to 2018.

3.9. Conclusion

This chapter's main findings showed that silicon effectively reduces strawberry powdery mildew disease in different strawberry plant cultivars, such as the Malling Centenary cultivars 2016. It showed that silicon as a treatment could delay the start of disease level build-up by up to 29 days (2016). Therefore, the aim to investigate the effects of silicon in reducing strawberry powdery mildew was met in 2016. Although this chapter has shown that silicon can reduce the disease severity, the following chapter (chapter 4) explores how silicon can be involved in enhancing the passive defence pathway of strawberries, thereby reducing susceptibility to strawberry powdery mildew disease.

Chapter 4 – Silicon deposition in strawberry plants

This chapter covers i) An introduction to the deposition of silicon in plants; ii) Rationale, aim, objectives and hypothesis; iii) Material and methods of the chapter; iv) Results; v) Discussion and vi) Conclusion.

4.1. Introduction

In chapter 3, results showed that a silicon nutrient application in a fertigation field experiment in 2016 reduced susceptibility to strawberry powdery mildew (*Podosphaera aphanis*) and also in previous years' field experiments showed by Jin (2015) and Liu (2016). This chapter investigates the deposition, amount and pattern of distribution of silicon and its role in enhancing the passive defence pathway of strawberry plants about strawberry powdery mildew (*Podosphaera aphanis*) disease reduction (Debona et al., 2017).

4.1.1. Silicon deposition in plants

This chapter explores how strawberry powdery mildew disease reduction can be attained with the use and deposition of silicon in strawberry plants. This chapter also explains the location of silicon deposited in plants. Firstly, silicon is present in all plants (Greger, Landberg and Vaculík, 2018). Silicon accumulation differs significantly between plant species due to differences in silicon uptake by the roots (Rodrigues et al., 2015). Work done by Jin (2015) established in a glasshouse experiment on strawberry plants revealed that the silicon nutrient administered through the roots of strawberries was deposited in the epidermal layer of the strawberry leaves and the xylem and epidermis of the petioles of treated strawberry plants. SEM images showed that the wax formation of strawberry leaves was greater and thicker in silicon-treated plants compared with the untreated control (Jin, 2015). Debona et al. (2017) found that silicon deposition and polymerization beneath the cuticle, cell walls and bulliform cells formed a physical barrier causing a reduction in the severity of rice blast disease.

4.2. Rationale, aim, objectives and hypothesis of silicon deposition in strawberry plants

4.2.1. Rationale

Silicon nutrient can reduce levels of strawberry powdery mildew. However, this can be linked to the accumulation (deposition) of silicon in strawberry plants' tissues (leaves and petioles). This chapter sought to examine the amount and pattern of deposition of silicon in strawberry plants. This experiment examined plants growing in a glasshousecontrolled environment and field tunnels.

4.2.2. Aim

This chapter aimed to examine the amounts and pattern of distribution of silicon in leaves, petioles and roots of strawberry plants in glasshouse and field experiments.

4.2.3. Objectives

- Assess the cross-sections of strawberry leaves, leaf petioles and roots of silicontreated and untreated strawberry plants from a glasshouse and field experiments with a LysoTracker Yellow HCK-123 dye.
- Quantify the amounts of silicon deposits in cross-sections (leaves, leaf petioles and roots) of strawberry plants from a glasshouse and field experiments from stained samples viewed through a fluorescence microscope.
- Determine the pattern of silicon deposition in leaves, leaf petioles and roots in both glasshouse and field experiments.

4.2.4. Hypothesis

- There can be higher silicon levels in treated plants compared to untreated plants.
- The silicon can be deposited in a regular manner (form) in the leaves.

4.3. Material and methods

4.3.1. Silicon deposition

The deposition of silicon in strawberry plants experiment was established in the University of Hertfordshire, Hatfield glasshouse in 2017 and the silicon fertigation field experiment in 2017 and 2018. See chapter 2, table 2.2, for treatments used. Strawberry plants in the 2017 and 2018 silicon fertigation field experiments, established for disease assessments, silicon extractions and measurements of silicon content in fertigation water, were harvested for a silicon deposition experiment at the end of the season.

Harvested strawberry plants were placed in large polythene bags and labelled. Eight whole plants per treatment and 32 were harvested from the silicon fertigation field experiments for the deposition examination. Plants were stored in a cold room at 4°C in the Science Building, College Lane Campus at the University of Hertfordshire. Before the examination, the leaves, leaf petioles and roots were carefully detached from the plants using a sharp single-edged surgical carbon-steel blade to prevent any damage.

Ten cross-sections of the leaves, leaf petioles and roots per treatment and 60 crosssections in total were sectioned (from glasshouse and field experiments) each time using new sets of the single-edged surgical carbon-steel blades, following the old traditional method by placing each plant part (leaf, leaf petioles and root) in a carrot and slicing of the tips severally to achieve a one-cell thick section each time. Surgical blades were replaced after four uses. Each cross-section was picked up using a paintbrush and transferred onto labelled 8 x 6 x 2.5 cm microscope glass slides for fluorescence dye staining.

4.3.2. Plant sectioning

At the end of each experiment, ten leaves, ten petioles and ten roots per treatment were sampled from each treatment in the glasshouse and field experiments. All cross-sections were stained using the LysoTracker dye. See section 4.3.3 below for details on the fluorescence dye staining.

4.3.3. LysoTracker Yellow HCK-123 fluorescence dye staining

The LysoTracker Yellow HCK-123, a fluorescence dye, was used in staining crosssections of plants (Desclés et al., 2008; Gröger, Sumper and Brunner, 2008; Shetty et al.,2012; Oh et al., 2018). The LysoTracker dye was obtained from the ThermoFisher Scientific manufacturer. Each plant section was stained separately on slides for two hours in the dark. The Vectashield mounting media from the ThermoFisher Scientific manufacturer was used in drops to preserve fluorescence in samples while staining. After two hours of staining, samples were moved from the microbiology laboratory to the microbiology preparation room for observation. Cross-sections were viewed using the GXML3201 LED fluorescence microscope. Images viewed were saved onto Secure Digital Cards (SD) for further observations and a silicon quantification analysis. Silicon fluorescence intensities (density) were assessed and quantified using the software program ImageJ, which measures the total amounts of fluorescence within each sample. See appendix 23.

4.3.4. GXML3201 LED fluorescence microscope

Each sample slide containing a leaf, leaf petiole and root cross-section was viewed using the GXML3201 LED fluorescence microscope (figure 4.1) to determine the location of silicon deposits through the accumulation of fluorescence intensity. Each
sample slide was viewed at x40 and x400 magnification using a green filter and a wavelength of 450nm. Photos were saved each time by selecting the "capture icon" on the microscope screen onto a removable Secure Digital Card (SD) card for fluorescence intensity measurements. Measurements of fluorescence intensities were conducted using the ImageJ processing program designed for multidimensional scientific images for quantification. Refer to appendix 24 for the protocol for using ImageJ.

4.3.5. Silicon accumulation (fluorescence intensity) measurement

Fluorescence intensities were measured using ImageJ, a multidimensional imaging processing software involving quantifying the fluorescence intensities by downloading direct images from a microscope (Collins, 2007; Jensen, 2013; Bankhead, 2014). The unknown area (blank) of the image (cell) is subtracted from the integrated density (total fluorescence) of the cell, which equals the total quantity of fluorescence (without background fluorescence). The total fluorescence (integrated density) given is divided by the area of the cell, which then equals the actual quantity of fluorescence per given sample.

Therefore:

$$Actual Fluorescence = \frac{Fluorescence of the cell}{Area of the cell}$$
$$= Integrated Intensity - Blank (unknown area of the cell)$$

Fluorescence intensities are recorded in Relative Fluorescence Units (RFU) or counts per second (cps) arbitrary units (Peruski, Johnson and Peruski Jr, 2002; Neumeier, Heck and Feldmann, 2019). Refer to appendix 19 for a protocol for using ImageJ.



Figure 4.1. GXML3201 LED fluorescence microscope with a cross-section sample at x 40 magnification.

4.3.6. Treatments

The glasshouse pot compost experiment had only two treatments; the silicon-treated (0.017% v/v at 50 mls per plant pot) and the untreated control (50 mls of de-ionized water). Field experiments only used silicon treatments (0.017% v/v via fertigation system) and the untreated for deposition comparison and statistical analysis. Silicon was applied through the roots in both experiments.

4.3.7. Cultivars

The cultivar Amesti was used in the silicon fertigation field experiment in 2017 and 2018. The cultivar Malling Centenary was used in the glasshouse pot compost experiment. To describe the Amesti and Malling Centenary cultivars used in this experiment, refer to chapter 1, table 1.4.

4.4. Results

4.4.1. Pattern and deposition of silicon in the leaves, leaf petioles and roots from a glasshouse experiment 2017 (quantification of fluorescence intensity)

See results from the silicon deposition in a glasshouse experiment (pot compost) 2017 in figure 4.2, table 4.1 to 4.2 (appendix 25 to 33). Table 4.1 includes mean fluorescence intensities (untreated and silicon-treated cross-sections) from ImageJ recorded in relative fluorescence units (RFU) and counts per second (cps).

At the end of the glasshouse experiment, cross-sections of ten leaves, ten petioles and ten roots were sampled from each treatment (untreated and silicon-treated). All cross-sections were stained using the LysoTracker Yellow HCK-123 dye. The data collected for the 2017 glasshouse experiment was first tested for normality before using the one-way ANOVA to determine the differences in fluorescence intensities (levels of silicon deposits) among treatments such as untreated and silicon-treated. Refer to appendix 25 to 33 for all (total) replicates of fluorescence (cross-sections) images from the glasshouse experiment.



Figure 4.2. Silicon deposition in glasshouse experiment 2017.

Silicon deposits in green fluorescence. Viewed at x 40 and x 400 magnification. (a) Silicon-treated (leaf) More silicon was found in the upper and lower cuticle, epidermis, palisade layer and leaf vein compared with untreated (b). (c) Silicon-treated (petiole) More silicon was found in the cuticle and xylem than untreated (d). e) Silicon-treated (root) More silicon was found in the treated compared with untreated (f). Fluorescence Intensity quantification - Counts per second (cps) is (a) 7.9cps for the silicon-treated (leaf) (b) 2.2cps for untreated (leaf) (c) 7.7cps for the silicon-treated (petiole) (d) 1.9cps for untreated (petiole) (e) 11.6cps for the silicon-treated (root) and (f) 1.2cps from untreated (root).

Figure 4.2 above show some images of the different intensity levels and deposition pattern in the 2017 glasshouse experiment. Figure 4.2(a) shows the amounts of fluorescence intensity and pattern of silicon deposition in the leaf (7.9cps) that the silicon was laid down regularly in the upper and lower epidermis, palisade layer, xylem and vein of the leaf in silicon-treated plants (b) in the untreated, there was only a background level of silicon (2.2cps), found in the leaf, (c) shows the amounts of fluorescence intensity and pattern of silicon deposition in (7.7cps) in the silicon-treated leaf petioles and is laid down in the epidermis and xylem (d) in the untreated, there is only a background level of silicon (1.9cps), found in the xylem (e) shows the amounts of fluorescence intensity and pattern of silicon deposition in a silicon-treated root (11.6cps), there was only a background level of silicon found in the xylem (f) In the untreated root, (1.2cps). Refer to appendix 25 to 33 for fluorescence intensities (integrated density) in the leaf, leaf petiole, and root from the glasshouse experiment in 2017 is in table 4.1

Ten leaves, ten petioles and ten roots per treatment were sampled for ten cross-sections each from the glasshouse experiment in 2017. Table 4.1 below is a summary table of findings from the glasshouse experiment. Treatments below include the leaf, leaf petiole and root from the untreated and silicon-treated plants. A mean reading was achieved each time from 3 readings per cross-section by ImageJ.

Fluorescence intensities from ImageJ presented in table 4.1 below show the leaves, leaf petioles and roots of silicon-treated plants compared to the untreated ones. Silicon was found throughout (leaves, leaf petioles and roots) of the strawberry plants in the glasshouse (pot compost) experiment in 2017. The images in appendix 25 to 33 were used to produce table 4.1below. Refer to appendix 60 for statistical workings to calculate the differences between the means sample of untreated and silicon-treated plant cross-sections presented in table 4.1 below. A summary table for the means of 10 cross-sections of strawberry plant leaves from the glasshouse experiment in 2017 is in table 4.2.

Table 4.1. Readings from fluorescence intensities (integrated density) in the leaf, leaf petiole and root from a glasshouse experiment 2017. Values are the mean of 3 readings per leaf, leaf petioles and roots.

Strawberry cross-section	Untreated	Silicon-treated
	2.2cps	4.7cps
	1.2cps	2.0cps
	1.1cps	1.9cps
	1.6cps	4.0cps
Leaves	3.8cps	5.7cps
Leaves	1.9cps	12.1cps
	4.5cps	6.3cps
	1.9cps	7.7cps
	3.7cps	3.5cps
	2.2cps	7.9cps
	4.1cps	5.0cps
	3.6cps	5.5cps
	2.3cps	10.7cps
	3.4cps	8.3cps
Leaf petiole	2.2cps	7.7cps
F	1.8cps	9.1cps
	2.0cps	9.0cps
	1.1cps	8.5cps
	3.5cps	5.9cps
	4.0cps	9.4cps
	3.2cps	9.2cps
	6.4cps	13.3cps
	4.8cps	8.6cps
	8.1cps	1.7cps
Root	7.4cps	9.9cps
	5.9cps	10.1cps
	4.3cps	7.9cps
	5.3cps	2.6cps
	7.3cps	8.7cps
	1.2cps	11.6cps

Table 4.2. Summary of 10 fluorescence intensities (integrated density) in the leaf,leaf petiole and root from a glasshouse experiment 2017

Cross-section	Treatments	Statistical test applied	Statistical findings (see
			appendix 60 for
			statistical workings)
	Untreated control (no	Shapiro-Wilk test for	There were statistically
	fungicides, no silicon)	normality and One Way	significant differences
	and silicon twice-a-week	ANOVA to find the	between group means as
Leaves	treated	average values of	determined by One Way
		differences between	ANOVA (F (1,18) =
		untreated and silicon-	[7.142], P = 0.016.
		treated plants.	
	Untreated control (no	Shapiro-Wilk test for	There were statistically
	fungicides, no silicon)	normality and One Way	significant differences
	and silicon twice-a-week	ANOVA to find the	between group means as
Leaf petioles	treated	average values of	determined by One Way
		differences between	ANOVA (F (1, 18) =
		untreated and silicon-	[22.604], P = 0.000.
		treated plants.	
	Untreated (no fungicides,	Shapiro-Wilk test for	There were statistically
	no silicon) and silicon	normality and One Way	significant differences
	twice-a-week treated	ANOVA to find the	between group means as
Roots		average values of	determined by One Way
		differences between	ANOVA (F (1, 18) =
		untreated and silicon-	[31.283], P = 0.000.
		treated plants.	

The work in this chapter suggests that there are more silicon deposits (P < 0.05) in the leaves, leaf petioles and roots of silicon-treated plants compared to the untreated from the 2017 glasshouse experiment. All samples were replicated as shown by their means in table 4.1 and appendix 25 to 33.

Previous work done by Jin (2015) on silicon deposition in the leaf and leaf petiole of strawberry plants in a glasshouse experiment only showed that more silicon was found in the silicon-treated plant's leaf and leaf petiole compared to the untreated ones. However, the level of deposition of silicon nutrients via fluorescence intensities was not quantified or shown in the roots. This chapter also found that silicon nutrients applied through the roots of strawberries in a glasshouse experiment accumulated silicon throughout the plant, such as the leaves, leaf petioles and roots. The work showed the regularity of silicon deposited in the leaves, leaf petioles and roots.

4.4.2. Pattern and deposition of silicon in the leaves, leaf petioles and roots from the silicon fertigation field experiment 2017 (quantification of fluorescence intensity)

Figure 4.3 and appendix 34 to 42 show where silicon is deposited in the plants harvested from the 2017 field experiment. Table 4.3 shows the mean quantification of fluorescence intensities in the leaves, leaf petioles and roots from the 2017 field experiment. The silicon deposition and quantification experiment give a precise idea of the location and amount of silicon within the plant before silicon extraction. At the end of the 2017 field experiment, eight whole strawberry plants per treatment were sampled for ten leaves, ten leaf petioles and ten roots from each treatment (untreated and silicon twice only treatment). All cross-sections were stained using the LysoTracker Yellow HCK-123 dye. See appendix 34 to 42 for more fluorescence images from this experiment in 2017. Refer to appendix 61 for statistical workings. The data collected from the 2017 field experiment was first tested for normality before using the two-way ANOVA to determine the differences in fluorescence intensities among mean treatments. The images presented in figure 4.3 are samples taken from the untreated and silicon twice-a-week only treatment. Means of three readings were done in every ten cross-sections, as shown below. Figure 4.3 shows examples of results images from the fluorescence microscope. Refer to results in table 4.3 for fluorescence intensities measurements, appendix 34 to 42 and appendix 61 for statistical workings.



Figure 4.3. Silicon deposition in silicon field experiment 2017. Silicon deposits in green fluorescence.

Viewed at x 40 and x 400 magnification(a), Silicon-treated (leaf) showed that more silicon was found in the upper and lower cuticle, epidermis and palisade layer compared with untreated (b). (c) Silicon-treated (petiole) More silicon was found in the xylem compared with untreated (d) (e) Silicon-treated (roots) show no silicon was found in both the treated and untreated (f). The fluorescence intensity quantification for each figure is 19.4 cps for (a) silicon twice only (leaf), (b) 7.9 cps for untreated (leaf), (c) 16.7 cps for the silicon twice only (petiole), (d) 10.2 cps for untreated (petiole) (e) 6.9 cps for the silicon twice only (root) and (f) 6.0 cps for untreated (root). Note: the best images captured from the GXML3201 LED fluorescence microscope are provided in the whole of the chapter; refer to appendix 15 for replicates of cross-sections.

The images in figure 4.3 above show some examples of the different intensities and deposition patterns in the 2017 field experiment. Figure 4.3(a) shows the amounts of fluorescence intensity and pattern of silicon deposition in the leaf (19.4cps) that the silicon was laid down regularly in the upper and lower epidermis, cuticle, and palisade layer in silicon-treated plants (b) in the untreated, there was only a background level of silicon (7.9cps), found in the leaf, (c) shows the amounts of fluorescence intensity and pattern of silicon deposition (16.7cps) in the silicon-treated leaf petioles and is laid down in the epidermis and xylem (d) in the untreated, there is some level of silicon (10.2cps), found in the xylem (e) shows the amounts of fluorescence intensity and pattern of silicon deposition in the silicon-treated root (6.9cps), there was only a background level of silicon found in the xylem (f), and in the untreated root, there was also only background level of silicon (6.0cps), found in the xylem. Refer to appendix 34 to 42 for replicates of fluorescence images from the 2017 field experiment. The amounts of silicon in the different treatments were statistically analysed using the cross-section means presented in table 4.3 below.

Ten leaves, 10 petioles and 10 roots per treatment were sampled for 10 cross-sections each from the field experiment in 2017. Treatments below include the leaf, leaf petiole and root from the untreated and silicon twice only treatment (for statistical analysis). Mean readings were achieved each time from 3 readings per cross-section by ImageJ.

From table 4.3 below (2017 field experiment), it appears that more silicon is found in cross-sections of the leaves, leaf petioles and roots of plants treated with silicon compared to the untreated ones. Refer to appendix 61 for statistical tests used to evaluate the means of both treatment means. A summary table for the means of ten cross-sections of strawberry plant leaves, leaf petioles and roots from the field experiment in 2017 is presented in table 4.4

Strawberry cross-section	Untreated	Silicon-treated
	7.99cps	19.4cps
	3.82cps	12.13cps
	4.34cps	8.24cps
	6.91cps	8.34cps
Lagyas	7.71cps	6.61cps
Leaves	8.64cps	10.33cps
	6.53cps	12.63cps
	2.84cps	7.46cps
	6.84cps	10.16cps
	5.32cps	8.32cps
	6.24cps	16.7cps
	6.53cps	8.96cps
	7.11cps	12.2cps
	3.61cps	7.95cps
Leaf petiole	4.73cps	4.66cps
I	6.71cps	13.8cps
	8.41cps	10.56cps
	7.84cps	9.65cps
	5.32cps	3.33cps
	4.43cps	5.22cps
	6.34cps	6.94cps
	2.27cps	7.97cps
	4.45cps	8.44cps
	3.99cps	5.41cps
Root	6.01cps	6.94cps
	7.73cps	8.42cps
	3.24cps	8.14cps
	3.37cps	8.15cps
	6.45cps	10.35cps
	5.45cps	7.85cps

Table 4.3. Fluorescence intensities (integrated density) in the leaf, leaf petiole androot from a field experiment 2017

 Table 4.4. Summary of 10 fluorescence intensities (integrated density) in the leaf,

 leaf petiole and root from a field experiment in 2017

Cross-	Treatments	Statistical test	Statistical findings (see appendix
section		applied	61 for workings)
Leaves,	Untreated and	Shapiro-Wilk test for	There was a statistically significant
leaf	silicon twice-	normality and Two	difference between treatment means
petioles	a-week only	Way ANOVA to find	as interpreted by the Two Way
and	treatment	the average values of	ANOVA (F (3) = 10.047, P = 0.000.
roots		differences between	However, the interaction between
		untreated and silicon	the block and treatment terms was
		twice-treated plants.	not significant, (F $(3) = 1.465$, P =
			0.197 as shown by the output
			model.

Examples of the 2017 silicon fertigation field experiment results are shown in figure 4.3. Results have shown that all plants contain little silicon; however, silicon treatments enhanced the amount of silicon naturally present within the plant. Results (figure 4.3 and Appendix 61) showed that more silicon had accumulated in silicon-treated plants than in untreated ones. In strawberry leaves silicon-treated, more silicon deposits were laid down regularly in the upper and lower cuticle, epidermis and palisade layers compared to the untreated ones.

In strawberry leaf petioles, silicon was found mainly in the xylem of silicon-treated plants compared with the untreated ones. The cross-section of the roots of treated and untreated plants shows that no additional silicon was deposited in the roots (figure 4.3). However, fluorescence quantification has shown that there was little silicon found in the roots of silicon-treated and untreated ones. Images in Figures 4.3i, ii and iii are unstained cut cross-sections (as in figure 4.2) of a strawberry leaf, leaf petiole and root viewed under UV light at x40 magnification through the GXML3201 fluorescence microscope. This chapter presents these images to show the plant's anatomy without the Lysotracker fluorescence dye.

4.4.3. Pattern and deposition of silicon in leaves, petioles and roots from the silicon fertigation field experiment 2018 (quantification of fluorescence intensity)

Figure 4.4 shows some examples of fluorescence images of the cross-section of the leaves, leaf petioles and roots of plants from the 2018 silicon fertigation field experiment. The means used in analysing the cross-sections from this experiment are provided in table 4.5 below.



Figure 4.4. Figure Silicon deposition in silicon field experiment 2018.

Silicon deposits in green fluorescence. Viewed at x 40 and x 400 magnification. (a) Silicon-treated (leaf) showed that more silicon was found in the upper and lower cuticle, epidermis and palisade layer compared with untreated (b). (c) Silicon-treated (petiole) showed that more silicon was found in the xylem than untreated (d). e) Silicon-treated (root) showed that no silicon was found in both the treated and untreated (f). The fluorescence intensity quantification is as follows: (a) 4.98cps silicon twice only (leaf) (b) 2.2 cps for untreated (leaf) (c) 3.72 cps for the silicon twice (petiole) (d) 1.98cps for untreated (petiole) (e) 4.97 cps for the silicon twice (root) and (f) 1.76 cps for untreated (root).

The images in figure 4.4 above show some of the different intensities and deposition patterns in the 2018 field experiment. Figure 4.4(a) shows the amounts of fluorescence intensity and pattern of silicon deposition in the leaf (4.98cps), that the silicon was laid down systematically in the upper and lower epidermis, cuticle and barrier layer of the leaf in silicon-treated plants (b) in the untreated, there was only a background level of silicon (2.2cps), found in the leaf, (c) shows the amounts of fluorescence intensity and pattern of silicon deposition in (3.72cps) in the silicon-treated leaf petioles and is laid down mainly in the xylem (d) in the untreated, there is only a background level of silicon (1.98cps), found in the xylem (e) shows the amounts of fluorescence intensity and pattern of silicon deposition in the silicon-treated root (4.97cps). The untreated shows a similar background level of silicon found in the root's xylem (f) (1.76cps). Refer appendix 43 to 51 for the remaining replicates of fluorescence images from this field experiment. The amounts of silicon in the different treatments were statistically analysed using means of replicates from table 4.5, shown below.

Table 4.5. Readings from fluorescence intensities (integrated density) in the leaves, leaf petioles, and roots from the silicon fertigation field experiment 2018.

-

Strawberry cross-		
section	Untreated	Silicon-treated
	2.54cps	2.90cps
	3.63cps	3.85cps
	3.71cps	6.54cps
	3.32cps	5.25cps
Leaves	3.15cps	3.88cps
	2.22cps	4.96cps
	5.14cps	7.47cps
	4.33cps	5.56cps
	5.79ps	3.36cps
	3.99cps	5.94cps
	2.09cps	3.72cps
	4.33cps	4.76cps
	4.44cps	6.56cps
	6.31cps	7.74cps
Leafnetiole	6.47cps	4.33cps
Lear periore	1.98cps	2.97cps
	4.56cps	7.43cps
	2.54cps	1.38cps
	3.85cps	3.66cps
	6.50cps	5.35cps
	1.76cps	2.03cps
	5.35cps	3.24cps
	4.68cps	2.43cps
	5.63cps	4.04cps
Root	3.33cps	3.71cps
	1.90cps	4.9cps
	5.53cps	4.34cps
	6.21cps	4.39cps
	1.90cps	3.85cps
	2.44cps	4.90cps

Fluorescence intensity images from table 4.5 above show that silicon was mainly found in the leaves and petioles. Still, some amounts were found in the roots of strawberry plants (no additional silicon) in the silicon fertigation field experiment in 2018. Refer to appendix 62 for a statistical test to evaluate the mean differences between both (crosssection) treatments. A summary table for the means of ten cross-sections of strawberry plant leaves, leaf petioles and roots from the field experiment in 2018 is presented in table 4.6 below.

Table 4.6. Summary of 10 fluorescence intensities (integrated density) in the l	eaf,
leaf petiole and root from a field experiment in 2018	

Cross-	Treatments	Statistical test	Statistical findings (see appendix
section		applied	62 for workings)
Leaves,	Untreated	Shapiro-Wilk test	There was a statistically significant
leaf	and silicon	for normality and	difference between block means as
petioles	twice only	Two Way ANOVA	interpreted by the Two Way
and	treatment	to find the average	ANOVA (F (2) = 7.967, P = 0.001.
roots		values of differences	However, the interaction between
		between untreated	the block and treatment terms was
		and silicon twice-	not significant, (F (6) = 1.042 , P =
		treated plants.	0.403.

Table 4.6 above showed the interpretation of the Two Way ANOVA conducted on treatments means from the 2018 field experiment. It revealed that there was a significant difference between block as sig. value P = 0.001 but the interaction between the treatment and block terms is not significant as sig. value is 0.403.

4.5. Discussion

4.5.1. Silicon deposition in strawberry plants from a silicon fertigation field experiment in 2017 and 2018 and a glasshouse experiment (pot compost) in 2017

This chapter aimed to examine the amounts and pattern of distribution of silicon in leaves, petioles and roots of strawberry plants in glasshouse and field experiments. The hypothesis was that there would be higher levels of silicon in treated plants compared to untreated plants, and the silicon would be deposited regularly (form) in the leaves. Although all plants, including the untreated ones, have a background level of silicon, there will be higher levels of silicon in silicon-treated plants than in untreated plants.

Figure 4.2 and table 4.1 show the results from silicon deposition in a glasshouse experiment in 2017 of means of ten cross-sections per leaf, leaf petioles and root results

showed that the strawberry leaves of silicon-treated strawberry plants had more (P<0.05) deposits of silicon laid down regularly in the upper and lower cuticle, epidermis, palisade layer and leaf vein. In the leaf petioles, more (P<0.05) silicon was found deposited in the cuticle and xylem of silicon-treated plants compared to the untreated ones and the roots, more silicon (P<0.05) was found deposited mainly in the xylem of silicon-treated plants compared to the untreated ones. Means of three readings of each of the ten replicates (leaves, leaf petioles and roots) were used to produce the statistical tests. Strawberry leaves from the silicon fertigation field experiments in 2017 and 2018 showed that more (P<0.05) silicon was found in the cuticle and palisade layer of the silicon-treated plants compared with the untreated ones.

Domiciano et al. (2013) showed silicon accumulation was found in foliar tissues, which delayed pathogen ingress into epidermal cells and reduced fungal colonization. Sousa et al. (2013) reported how the amount of silicon deposited in wheat leaves restricted hyphal entry to the epidermal cells, while hyphae successfully invaded several neighbouring leaf cells where silicon was not found. In a glasshouse experiment, Jin (2015) recently experimented with a glasshouse at the University of Hertfordshire to locate silicon accumulation in strawberry plants. The work Jin (2015) conducted revealed through confocal microscopic images that silicon-treated strawberry plants contained more silicon than untreated strawberry plants; moreover, some silicon was found in untreated plants. Compared with the untreated, Silicon was found in the epidermis and the vascular tissue in the leaf petiole compared with the untreated ones.

More recently, work conducted by Mvondo-she and Marais (2019) investigated silicon localization and accumulation in Citrus plants and the Cultivars used for the study were Valencia 'Delta' and Clementine 'Nule'. The experiment was done in a glasshouse at the Experimental Farm of the University of Pretoria in pots containing an artificial growing medium. Coir-Perlite and silicon were applied via the roots (Mvondo-she and Marais, 2019). Their work found through statistical findings that there was a higher amount of silicon deposits in silicon-treated leaves compared with the untreated control. Lastly, this experiment also identified the presence of silicon granules on the surface and around the outer cell surface, forming a double cuticle layer of the lower epidermis in the leaves of silicon-treated Citrus plants compared with the untreated control (Mvondo-she and Marais, 2019).

Figures 4.2 to 4.4 and tables 4.1 to 4.6 showed that in the leaf petioles of silicon-treated plants, more (P<0.05) silicon deposits were laid down regularly in cuticle and xylem compared with the untreated in the glasshouse and field experiment in 2017. However, statistics also revealed that silicon deposits of untreated leaf petioles were not higher (P>0.05) than the silicon-treated leaf petioles in the silicon field experiment in 2018 (figure 4.4). Jin (2015) showed in her thesis that in the leaf and leaf petiole of silicon-treated plants, there was more silicon found in the cuticle and vascular tissue of silicon-treated plants compared with the untreated ones. However, the roots were not tested,

nor were the results quantified or tested for significance. Rao and Susmitha (2017) also found that silicon accumulation in silicon-treated plants formed a cuticle-silicon double layer, which maintained the erectness of rice leaf blades. Figure 4.2 to 4.3 showed that there was more silicon (P<0.05) deposited in the roots of silicon-treated plants in the 2017 glasshouse and field experiment, and no differences (P>0.05) were found in silicon deposits between the leaf petioles and untreated roots from the 2018 field experiment.

Results from this chapter indicated that fluorescence intensity quantification (table 4.1, 4.3 and 4.5) found more silicon in strawberry leaves and petioles, but little amounts found in the roots of strawberry plants in the 2017 and 2018 field experiments. Results from the field experiments in 2017 and 2018 (figure 4.3 and 4.4 and table 4.3 to 4.6) showed that silicon accumulated mainly in the leaves and petioles of strawberry plants compared with the glasshouse pot experiment (figure 4.2 and table 4.1) where large quantities of silicon were found throughout the leaves, petioles and roots of strawberry plants than results from field experiments (figure 4.3 and 4.4 and table 4.1 to 4.6). However, the results from the roots in the field experiments were different and unexpected, in that there were no significant differences seen in quantities of silicon between treatments (untreated and twice only treatment) in the field experiments.

The strawberry plants in the glasshouse were permanently exposed to silicon because they were in pots. In the fields, there were only exposed to the silicon twice a week (two out of 42 fertigation events). Thus, the exposure to silicon was far less in the fields. The hypothesis is that the silicon is taken up by the roots (via a silicon transporter gene) (Ma et al., 2004). The bio-available silicon is transported in the water in the xylem, distributed to the leaves via the petiole, and distributed throughout the leaves in the leaf vein. At the leaf surface, water leaves the plant by guttation, and water vapour leaves the plant by evaporation. The hypothesis is that the silicon in the water in the leaf becomes more concentrated in the leaves and is deposited in the cuticle and on the cell walls.

There was a difference in the amount of silicon deposited in the roots between the glasshouse and the field experiment (table 4.1 to 4.6), and the explanation for this can be the differences in the exposure time of silicon in the glasshouse (pot compost) and field experiment (fertigation). This chapter shows that silicon appears to be deposited within the plant using 10 leaves, 10 leaf petioles, and 10 roots from the glasshouse and fertigation field experiment. The aim, objectives and hypothesis that silicon delivered through a glasshouse experiment and in field experiments is distributed around the plants were met.

4.6. Conclusion

This chapter revealed that the bioavailable silicon nutrient delivered via the fertigation through the roots of strawberry plants was deposited and laid down regularly (distinct crystalline form) (Debona et al., 2017). This chapter showed that silicon could enhance the features of the passive defence pathway (through silicon deposits in the upper and lower epidermis, palisade layer, and leaf vein). It quantified the amounts of silicon deposited (fluorescence intensities) in strawberry plants, which appear not to be shown particularly in strawberries. The deposition and quantification of amounts of silicon in strawberry plants shown in this thesis provided actual measurements of silicon accumulation over time. The aim and objectives of this chapter were to examine the amounts and pattern of distribution of silicon in the leaves and petioles of strawberry plants in a glasshouse and field experiments. These aims and objectives were met. The measurements of the benefits of silicon, such as strawberry plant growth parameters and fruit quality in field and glasshouse experiments, are presented in the following chapter (chapter 5).

Chapter 5 - Measurements of strawberry plant growth parameters and quality about silicon

This chapter covers i) An introduction to the measurements of strawberry plant growth parameters and quality concerning silicon; ii) Rationale, hypothesis, aim and objectives; iii) Material and methods; iv) Results; v) Discussion and vi) Conclusion.

5.1. Measurements of strawberry plant growth parameters and quality concerning silicon

The literature suggests that silicon as a nutrient has significant benefits in growing crops such as rice, wheat, melon, cotton and citrus (Rodrigues et al., 2015). Silicon has been shown to have benefits on non-accumulators; see table 5.1. Table 5.1 suggests that there are benefits in non-silicon accumulators such as an increase in flowering, fruit number, fresh weights, fruit shelf life etc. Although some strawberry growers are concerned about toxicity, the literature suggests that silicon is shown to reduce the toxicity of metals (Liang et al., 2005).

As explained by Liang et al. (2005), one of the main effects of silicon toxicity is reducing the uptake and translocation of metals in plants. Growers fear excess silicon may cause white strawberry fruits (albinism) and reduced yield (AHDB, 2011). AHDB advises using no more than 22mg/L of silicon (optimal ranges) (AHDB, 2011). Work reported in the chapter has focused on the benefits of silicon on strawberry plants. The experiment reported here found some benefits linked with using silicon in growing strawberry plants. Silicon nutrients applied through the roots of strawberry plants growing in Hoagland's solution in a hydroponic experiment increased the numbers of leaves, runners, fruits, chlorophyll contents in the leaves and fresh weights biomass in silicon-treated plants compared with the untreated ones. Refer to chapter 1, section 1.22, and tables 1.7 and 1.8 to review some of the benefits of silicon use.

An example is an experiment conducted by Li et al. (2018), who reported that silicon increased the biomass of cotton plants and bast fibre in plants treated with silicon compared with the control plants. Johnson et al. (2018) also mentioned that silicon promoted yield and upright growth, including stronger and thicker stems, which prevented lodging, promoting conducive exposure of light to the leaves of wheat plants. It is reported that silicon deficiency in plants can affect the normal development of healthy leaves, petioles and roots, causing the plant to be more susceptible to disease infections (Hajiboland et al., 2018). Deus et al. (2019) showed that silicon deficiency symptoms in rice leaf blades included reduced photosynthetic activity and increased susceptibility to diseases such as blast caused by *Pyricularia oryzae* or brown spot caused by *Helminthosporium oryzae*, reduction in grain yield and prone to lodging. The

work reported in this chapter compares non-silicon with silicon-treated plants for silicon deficiency symptoms and the measurements of strawberry plant growth parameters in a glasshouse hydroponic and field experiment (°Brix). Part of the work reported here followed previous work at the University of Hertfordshire. Jin (2015) and Liu (2016) found that silicon increased °Brix levels of strawberries growing on a commercial strawberry farm. Table 5.1 below shows some of the benefits of the application of silicon to plants that are non-accumulators of silicon.

Crop	Uptake of	Benefits	References
species	silicon		
Strawberry	Non- accumulators	Enhanced fruit diameter, weight, glucose, fructose content and shelf life.	(Peri-Felipo,2020)
		Enhanced size, weight, firmness, °Brix of fruit.	(Zahedi et al., 2020)
		Increase in chlorophyll contents, area and numbers of strawberry leaves.	
Tomato	Non- accumulators	Increased yield of fruit and vegetative growth.	(Hoffmann et al., 2020)
		Increased commercial productivity and reduced occurrence of cracked fruits.	
		Enhanced vitamin c content and fruit firmness.	
			(Marodin et al., 2014)
Sunflower	Non- accumulators	Early flowering, increased stem diameter.	(Kamenidou, Cavins and Marek; 2008)
		Improved biomass and increased tolerance to salinity.	(Sacib et al. 2011)
			(Saqib et al., 2011)
Gerbera	Non- accumulators	Increased diameter of flower and thickness of flower stems.	(Savvas et al., 2002)
Petunia	Non- accumulators	Thicker leaves.	(Jana and Jeong, 2014)
Begonia	Non- accumulators	Increased biomass parameters.	(Mills-Ibibofori et al., 2019)

Table 5. 1 Benefits of using silicon on non-accumulators of silicon

Table 5.1 includes only the benefits of using silicon on non-accumulators of silicon, as the strawberry plant is a non-accumulator of silicon. The benefits of silicon to silicon accumulators are well known.

5.2. Rationale

Silicon is not considered an essential element for plant development. However, previous work has shown that silicon can improve strawberry plants' overall quality, ranging from enhanced pollen fertility to the °Brix and firmness of strawberry fruits. The work reported in chapter 5 aims to quantify the growth parameters of strawberry plants in silicon's presence (with) and absence (without).

5.3. Aim

- Evaluate °Brix levels of silicon-treated fruits and leaf petioles with untreated fruits and leaf petioles.
- In a glasshouse hydroponic experiment, measure growth parameters of strawberry plants in the absence and presence of silicon.

5.4. Objectives

- Measure °Brix levels of fruits and leaf petioles in a field experiment
- Conduct weekly assessments on growth parameters; leaf number, runner number, flowering, fruit number, °Brix levels, chlorophyll content and fresh weight biomass at the end of the experiment
- Monitor strawberry plants treated with and without silicon for physiological symptoms in a glasshouse hydroponic experiment.

5.5. Hypothesis

- Plants treated with silicon can show improved growth parameters compared with untreated plants growing without silicon in a glasshouse hydroponic experiment.
- Plants treated with silicon in a silicon field experiment can have elevated °Brix levels compared with untreated plants.

5.6. Material and methods

Material and methods used include strawberry fruits and leaf petioles °Brix measurements (chapter 2, section 2.5.1 and figure 2.8 and 2.9), measurements of strawberry plant growth parameters; numbers of leaves, runners and fruit counting once a week, chlorophyll content measurements (in the leaves) once-a-week, fresh weight

biomass at the end of the experiment in the hydroponic glasshouse are in sections 5.7.1 to 5.7.7 of this chapter. The data collected from field °Brix measurements and the hydroponic glasshouse weekly assessments of strawberry plant growth parameters were first tested for normality, and the Paired Two Samples for means t-test, Paired Sample test and Wilcoxon-Sign Ranked test was then used to analyse the sample means (compare treatment means) collected from both the field and glasshouse hydroponic glasshouse experiment.

5.7. Hydroponics (methods)

Malling Centenary (June bearer) plants were used for the glasshouse hydroponic experiment. This cultivar was chosen to enable the completion of the experiment throughout the growing season. The experiment was set up at the University of Hertfordshire, Bayfordbury glasshouse in 2018. Twenty bare-root Malling Centenary strawberry plants were planted in 5 L tubs (not glass, which contains silicon) containing Hoagland's solution (table 5.2 and figure 5.1). Hoagland's solution is a complete plant nutrient developed by Hoagland and Snyder, which contains the necessary nutrient needed for plants' normal function and growth (Hershey, 1994). The tubs were wrapped with black polythene to reduce light penetration, root growth and encourage algae to grow (figure 5.1). Circular holes were drilled into black plastic lids, and each plant was wrapped below its crown in foam stopper bungs (before inserting through the lid) to keep the crown above the liquid to prevent it from getting wet while its roots remained in the solution. Crowns of strawberry plants were elevated from the solution.

Two strawberry plants were planted per tub and spaced apart to minimize tub crowding (table 5.2 and figure 5.1). Sufficient air supply to the roots was provided by two Oase Pontec PondAir aeration pumps from Aqautix-2U. Eheim T junctions were attached to the aeration tubes to enable the spread and circulation of oxygen to the roots of the strawberry plants in all tubs. (figure 5.1 and 5.2 and table 5.2). Treatments consisted of weekly 50ml applications of silicon nutrient per hydroponic tub. The concentration of silicon nutrient used was 0.017% of Sirius, estimated as 0.025ml to 0.027ml (0.003mg/cm³) (Liu, 2016) for the treated strawberry plants. The untreated set (no silicon nutrient) used de-ionized water only. Refer to figures 5.1, 5.2 and 5.3 and table 5.2 for the materials used and the setup of the hydroponic system.

The hydroponic tubs were topped up once weekly with Hoagland's solution as levels ran low as strawberry plants grew throughout the experiment. Refer to table 5.2 for nutrient components used to prepare Hoagland's solution (Hoagland and Arnon, 1950). Photos of the hydroponic experiment set-up are in figures 5.2 a and b, while the recipe for Hoagland's solution is in appendix 63. Figure 5.1 below is the design and treatment plan used in the glasshouse hydroponic experiment.



Figure 5. 1. A design plan showing the treatment and plants used for the hydroponic experiment in 2018. T = Treated - with silicon. U = Untreated - without silicon, Total number of plants used in experiment = 10 silicon-treated and 10 untreated = 20.



Figure 5.2. Hydroponic experiment in at UH Bayfordbury glasshouse 2018. Strawberry plants were arranged in plastic tubs wrapped in black polythene and treated once-a-week.

Component	Stock	mL Stock
	solution	Solution/1L
Macronutrients		
2M KNO ₃ (Potassium nitrate)	202 g/L	2.5
1M Ca (NO ₃) ₂ .4H ₂ O (Calcium nitrate)	236 g/0.5L	2.5
Iron (Sprint 138 Iron chelate) Ferric EDTA	15 g/L	1.5
2M MgSO ₄ .7H ₂ O Magnesium sulfate	493 g/L	1
1M NH ₄ NO ₃ Ammonium Nitrate	80 g/L	1
Micronutrients		
H ₃ BO ₃ Boric Acid	2.86 g/L	1
MnCl ₂ .4H ₂ O Manganese chloride	1.81 g/L	1
ZnSO ₄ .7H ₂ O Zinc sulfate	0.22 g/L	1
CuSO ₄ .5H ₂ O Copper 11 sulfate	0.051 g/L	1
H ₃ MoO ₄ .2H ₂ O Molybdic Acid	0.09 g/L	1
1M KH ₂ PO ₄ (pH to 6.0)	136 g/L	0.5

Table 5. 2 Nutrient components for Hoagland's solution

Source: Hoagland and Arnon, (1950); Hothem, Marley and Larson, (2003). Protocol for preparation of Hoagland's solution is in Appendix 63.





Figure 5.3. Stages of silicon-treated strawberry plant root development in Hoagland's solution in the UH Bayfordbury glasshouse (a) Roots of silicon-treated plants at the second week of planting. (b) Roots of silicon-treated plants at four weeks of planting. (c) Roots of silicon-treated plants at 20 weeks of planting (near the end of the experiment). (d) Roots of silicon-treated plants at the end (22 weeks) of the experiment were hand-held by the author with a plastic lid.

5.7.1. Number of Leaves

The total number of leaves per plant and treatment was counted once-a-week from the experiment's start (24 January 2018) to end (27 June 2018). When plants stopped producing newer leaves, their numbers were recorded as the same and unchanged.

5.7.2. Number of runners

Strawberry plant runners (daughter plants) produced in this experiment were counted once a week until the experiment ended. Each runner was potted in compost (still attached) growing beside the mother plant established in Hoagland's solution. All runners (figure 5.4) were cut off from the mother plant as soon as they were mature enough on their own.



Figure 5. 4. Strings of strawberry daughter plants (runners) in the hydroponic experiment.

5.7.3. Flowering

Flowering dates on strawberry plants from silicon-treated and untreated plants were recorded as soon as plants began to produce their flowers.

5.7.4. Fruit number

The numbers of fruits produced per plant and per week from the silicon-treated and untreated plants were recorded and picked in preparation for fruit °Brix measurements.



Figure 5. 5. Fruiting in strawberry plants in the hydroponic experiment 2018

5.7.5. Brix measurements

• Field experiment °Brix (Tesco's method) strawberry fruit

The grower used Tesco's method (the grower supplied Tesco). The method used in 2018 was cutting ten individual ripe strawberries around the circumference (middle). Then the juice was squeezed and placed directly on a digital refractometer, and the readings were recorded. Several years ago, Tesco's method was ten mashed strawberries and using a sample of juice to gain the reading with a refractometer (chapter 2, figure 2.8 and 2.9) (Tesco operating procedures). Growers measure petiole °Brix as an indication of how the fruit will develop. Strawberry fruits were sampled three times in July when fruiting was consistent and once in August 2017, when the fruiting season ended in the silicon field experiment 2017. Refer to chapter 2, section 2.5.1 for °Brix sampling methods.

• Field experiment °Brix (leaf petioles)

^oBrix of strawberry leaf petioles of silicon-treated and untreated plants was also evaluated in the 2017 silicon fertigation field experiment. Strawberry leaf petioles were sampled for ten individual leaf petioles per sample day. Sampling was done monthly, from June to September 2017 (chapter 2, figure 2.9).

• Glasshouse experiment °Brix (strawberry fruit)

°Brix from strawberry fruits was measured from the hydroponic experiment. Five individual strawberry fruits from each treatment (silicon-treated and untreated) on two sample days were selected for °Brix measurements. Strawberries were sampled twice throughout the experiment, and their °Brix were measured by the researcher with a refractometer (figure 2.9b). Refer to chapter 2, section 2.5.1, for methods used in sampling. Tesco's method was also used for measurements in the hydroponic experiment.

5.7.6. Chlorophyll

Chlorophyll contents of strawberry leaves were measured weekly during the hydroponic experiment. Measurements were carried out using the chlorophyll meter SPAD-502Plus from Konica Minolta (figure 5.5). The chlorophyll meter SPAD-502Plus is a lightweight and water-resistant. The handheld device is used to measure and determine the relative amount of chlorophyll present in the plant. The device works by measuring the absorbance of the leaf content without causing it to the plant (Minolta, 2009). Figure 5.6 is a photo of the SPAD meter used for chlorophyll measurement in this experiment.



Figure 5. 6. The chlorophyll meter SPAD-502Plus. Lightweight and water-resistant.

Measurements were derived by inserting leaf blades (one leaflet at a time) into the receptor window (see figure 5.6) and closing the measuring head of the chlorophyll meter. Data collected were recorded in μ mol/m2. Measurements were carried out for treated plants (with silicon nutrients) and untreated plants (without silicon nutrient). Strawberry leaves from each treatment were measured by randomly sampling ten leaves. Twenty measurements from untreated and silicon-treated plants were measured once a week. The averages of chlorophyll contents per treatment (untreated and silicon-treated) were recorded, and measurements were repeated each following week. The SPAD-502Plus has a memory space for up to 30 readings. Therefore, each piece of data can be recalled and saved after measurements (figure 5.6).

5.7.7. Fresh weight biomass

Whole strawberry plants (with leaves, leaf petioles, roots and runners still attached) were harvested at the end of the hydroponic experiment (figure 5.7). Each strawberry plant's fresh weight (g) from the silicon-treated and untreated were recorded.



Figure 5. 7. A harvested strawberry plant from the hydroponic experiment at 22 weeks. End of experiment.

5.8. Results

The graph includes average values of °Brix strawberry fruits sampled from the silicon fertigation field experiment in 2017. °Brix levels were sampled on 20 of July 2017, 24 July, 27 of July 2017 and 1 of August 2017 only.

5.8.1. Mean values of 10 strawberry fruit °Brix per treatment from the silicon fertigation field experiment 2017



Figure 5. 8. Means of 10 strawberry fruits °Brix levels from the silicon field experiment 2017. The different bars are average values of strawberry fruits sampled from 20 July to 1 August 2017. Light green is untreated, the blue is the fungicides-only treatment, the yellow is silicon applied twice weekly + fungicides, and the dark green is silicon applied twice weekly without fungicides. Means from 10 strawberries per treatment were used to create the graph shown. The Paired Sample for Means was used to analyse the data collected as it was normally distributed.

^oBrix levels of strawberry fruits from untreated and silicon-treated plants and analysis (appendix 64) are compared in the summary tables 5.3 and 5.4.

Sample	Treatment	Statistical test applied	Statistical findings (see
			workings)
°Brix levels in strawberry fruits	Untreated control (no fungicides, no silicon) and silicon twice- a-week + fungicides	Shapiro-Wilk test for normality and t-Test Paired Two Sample for Means to find the average values of differences between untreated and silicon twice-a-week + fungicides °Brix levels.	P>0.05; no statistical increase was found between strawberry fruits °Brix levels sampled from the silicon twice-a-week + fungicides treatment and untreated control sampled on 20 July 2017. The null hypothesis (HO) is accepted.
°Brix levels in strawberry fruits	Untreated control (no fungicides, no silicon) and silicon twice + no fungicides	Shapiro-Wilk test for normality and t-Test Paired Two Sample for Means to find the average values of differences between untreated and silicon twice-a-week + no fungicides °Brix levels.	P>0.05; no statistical increase was found between strawberry fruits °Brix levels sampled the silicon twice-a-week + no fungicides treatment and untreated sampled on 20 of July 2017. The null hypothesis (HO) is accepted.
°Brix levels in strawberry fruits	Untreated control (no fungicides, no silicon) and silicon twice- a-week + fungicides	Shapiro-Wilk test for normality and t-Test Paired Two Sample for Means to find the average values of differences between untreated and silicon twice-a-week + fungicides ^o Brix levels.	P<0.05; there was a statistical increase in the strawberry fruits ^o Brix levels sampled from the silicon twice-a-week + fungicides treatment compared to untreated sampled 24 of July 2017. The null hypothesis (HO) is rejected.
°Brix levels in strawberry fruits	Untreated control (no fungicides, no silicon) and silicon twice a week + no fungicides	Shapiro-Wilk test for normality and t-Test Paired Two Sample for Means to find the average values of differences between untreated and silicon twice-a-week + no fungicides °Brix levels.	P<0.05; there was a statistical increase in the strawberry fruits ^o Brix levels sampled from the silicon twice-a-week + no fungicides treatment compared to untreated sampled 24 of July 2017. The null hypothesis (HO) is accepted.

 Table 5. 3 Summary of °Brix levels from strawberry fruits from the 2017 field experiment

Table 5.3. shows an increase (P<0.05) in °Brix levels found only in silicon twice + fungicides and no fungicides treatment compared to untreated fruits sampled on 24 of July 2017, but no increase (P>0.05) in °Brix levels of fruits from the silicon twice-a-week + fungicides and no fungicides compared to untreated sampled 20 of July 2017. See table 5.4 for remaining °Brix levels of fruit results sampled on other days.

Table 5. 4 Summary of °Brix levels from strawberry fruits from the 2017 field experiment (continued)

Sample	Treatment	Statistical test	Statistical findings
		applied	(see appendix 64 for
			statistical workings)
^o Brix levels in strawberry fruits	Untreated control (no fungicides, no silicon) and silicon twice-a-week + fungicides	Shapiro-Wilk test for normality and t-Test Paired Two Sample for Means to find the average values of differences between untreated and silicon twice-a-week + fungicides °Brix levels.	P>0.05; no statistical increase was found between strawberry fruits °Brix levels sampled from the silicon twice-a-week + fungicides and untreated treatment sampled on 27 July 2017. The null hypothesis (HO) is accepted.
^o Brix levels in strawberry fruits	Untreated control (no fungicides, no silicon) and silicon twice a week + no fungicides	Shapiro-Wilk test for normality and t-Test Paired Two Sample for Means to find the average values of differences between untreated and silicon twice-a-week + no fungicides °Brix levels.	P>0.05; a statistical increase was found between strawberry fruits °Brix levels sampled from the silicon + no fungicides treatment and untreated sampled 27 of July 2017. The null hypothesis (HO) is accepted.
°Brix levels in strawberry fruits	Untreated control (no fungicides, no silicon) and silicon twice-a-week + fungicides	Shapiro-Wilk test for normality and t-Test Paired Two Sample for Means to find the average values of differences between untreated and silicon twice- a-week + fungicides °Brix levels.	P>0.05; no statistical differences were found between strawberry fruits °Brix levels sampled from the silicon twice + fungicides treatment compared to untreated sampled on 01 of August 2017. The null hypothesis (HO) is accepted.
Brix levels in strawberry fruits	Untreated control (no fungicides, no silicon) and silicon twice a week + no fungicides	Shapiro-Wilk test for normality and t-Test Paired Two Sample for Means to find the average values of differences between untreated and silicon twice- a-week + no fungicides °Brix levels.	P>0.05; there was no statistical difference found between strawberry fruits ^o Brix levels sampled from the silicon twice-a-week + no fungicides treatment compared to untreated sampled 01 of August 2017. The null hypothesis (HO) is accepted.

Conclusions from the summary in tables 5.3 and 5.4 show that °Brix levels of strawberry fruits in the silicon field experiments 2017 using the Paired t -Test Sample for Means showed that there was an increase (P<0.05) in strawberry fruits °Brix of the silicon twice + fungicides and no fungicides treatment compared to untreated sample on the 24 of July 2017 only and there was no increase (P>0.05) in strawberry fruit °Brix levels sampled on the 20 July, 27 of July 2017 and 1 of August 2017.

5.8.2. Means of 10 strawberry leaf petioles °Brix per treatment from the silicon fertigation field experiment 2017

The graph includes average values of °Brix leaf petioles sampled from the silicon fertigation field experiment in 2017. °Brix levels were sampled on 20 June 2017, 18 July, 01 August 2017 and 19 September 2017 only.



Figure 5. 9. Means of °Brix levels in strawberry leaf petioles from the silicon fertigation 2017. The different bars are average values of leaf petioles sampled from 20 June 2017 to 19 September 2017. Light green is untreated, the blue is the fungicides-only treatment, the yellow is silicon applied twice weekly + fungicides, and the dark green is silicon applied twice weekly without fungicides. Means from 10 leaf petioles per treatment were used to create the graph.

Growers routinely measure the °Brix of strawberry leaf petioles to understand the carbohydrates transported from the leaf petioles to the fruits (personal communication from grower). A summary of °Brix levels of leaf petioles from untreated and silicon twice-a-week only with and without fungicides treated plants from the 2017 field experiment is in tables 5.5 and 5.6. Find statistical workings in appendix 65.

Table 5. 5 Summary of °Brix levels from strawberry leaf petioles from the 2017 field experiment

Sample	Treatment	Statistical test applied	Statistical findings
[°] Brix levels in strawberry leaf petioles	Untreated control (no fungicides, no silicon) and silicon twice-a-week + fungicides	Shapiro-Wilk test for normality and t-Test Paired Two Sample for Means to find the average values of differences between untreated and silicon twice-a-week + fungicides °Brix levels.	P>0.05; no statistical increase was found in the °Brix levels of strawberry leaf petioles sampled from silicon twice-a-week + fungicides compared untreated treatment sampled 20 June 2017. The null hypothesis (HO) is accepted.
°Brix levels in strawberry leaf petioles	Untreated control (no fungicides, no silicon) and silicon twice + no fungicides	Shapiro-Wilk test for normality and t-Test Paired Two Sample for Means to find the average values of differences between untreated and silicon twice-a-week + no fungicides °Brix levels.	P>0.05; no statistical increase was found between strawberry leaf petioles ^o Brix levels sampled from the silicon twice-a-week + no fungicides treatment compared to untreated sampled 20 June 2017. The null hypothesis (HO) is accepted.
°Brix levels in strawberry leaf petioles	Untreated control (no fungicides, no silicon) and silicon twice-a-week + fungicides	Shapiro-Wilk test for normality and t-Test Paired Two Sample for Means to find the average values of differences between untreated and silicon twice-a-week + fungicides	P>0.05; no statistical increase was found in strawberry leaf petioles °Brix levels sampled from the silicon twice- a-week + fungicides treatment compared to untreated 18 of July 2017. The null hypothesis (HO) is accepted.
°Brix levels in strawberry leaf petioles	Untreated control (no fungicides, no silicon) and silicon twice-a-week + no fungicides	Shapiro-Wilk test for normality and t-Test Paired Two Sample for Means to find the average values of differences between untreated and silicon twice- a-week + no fungicides	P>0.05; no statistical increase was found in strawberry leaf petioles °Brix levels sampled from the silicon twice- a-week + no fungicides treatment compared to untreated 18 of July 2017. The null hypothesis (HO) is accepted.
°Brix levels in strawberry leaf petioles	Untreated control (no fungicides, no silicon) and silicon twice-a-week + fungicides	Shapiro-Wilk test for normality and t-Test Paired Two Sample for Means to find the average values of differences between untreated and silicon twice-a-week + fungicides	P>0.05; no statistical increase was found in strawberry leaf petioles °Brix levels sampled from the silicon twice- a-week + fungicides treatment compared to untreated sampled 01 of August 2017. The null hypothesis (HO) is accepted.
^o Brix levels in strawberry leaf petioles	Untreated control (no fungicides, no silicon) and silicon twice-a-week + no fungicides	Shapiro-Wilk test for normality and t-Test Paired Two Sample for Means to find the average values of differences between untreated and silicon twice-a-week + no fungicides	P>0.05; no statistical increase was found in strawberry leaf petioles °Brix levels sampled from the silicon twice- a-week + no fungicides treatment compared to untreated sampled 01 of August 2017. The null hypothesis (HO) is accepted.

From 5.5 above there is no increase (P>0.05) found in °Brix levels of leaf petioles between all treatments sampled from 20 June to 1 August 2017. See table 5.6 for samples collected in September 2017.

Table 5. 6 Summary of °Brix levels from strawberry leaf petioles from the 2017 field experiment

	Treatment	Statistical test	Statistical findings (see
Sample		applied	appendix 65 for
			statistical workings)
°Brix	Untreated control	Shapiro-Wilk test for	P>0.05; no statistical
levels in	(no fungicides, no	normality and t-Test	increase was found in
strawberry	silicon) and silicon	Paired Two Sample	strawberry leaf petioles
leaf	twice + fungicides	for Means to find the	°Brix levels sampled
petioles		average values of	from the silicon twice +
		differences between	fungicides treatment
		untreated and silicon	compared to untreated
		twice + fungicides	sampled 19 September
		°Brix levels.	2017. The null
			hypothesis (HO) is
			accepted.
°Brix	Untreated control	Shapiro-Wilk test for	P>0.05; no statistical
levels in	(no fungicides, no	normality and t-Test	increase was found in
strawberry	silicon) and silicon	Paired Two Sample	strawberry leaf petioles
leaf	twice + no	for Means to find the	°Brix levels sampled
petioles	fungicides	average values of	from the silicon twice +
		differences between	no fungicides treatment
		untreated and silicon	compared to untreated
		twice + no fungicides	sampled 19 September
		°Brix levels.	2017. The null
			hypothesis (HO) is
			accepted.

From results shown in table 5.5 and 5.6 there was no increase or differences (P>0.05) between °Brix levels of strawberry leaf petioles sampled from the silicon twice-a-week treatment compared to the untreated from the 2017 silicon fertigation field experiment.
5.8.3. Effects of silicon nutrient on average values of leaf number (per plant and treatment) in the hydroponic experiment

Results in figure 5.10 and appendix 66 revealed that the number of leaves in the hydroponic glasshouse experiment was enhanced from an average value of nine leaves (per plant and treatment) in untreated to 15 leaves (per plant and treatment) in the silicon-treated. Silicon-treated strawberry plants had a higher leaf number compared with untreated. This difference between the two treatments was statistically significant (P<0.05) as interpreted by the paired samples for means t-test analysis in appendix 66. Figure 5.10 shows average values of progress levels (leaves) between silicon-treated and untreated strawberry plants.



Figure 5. 10. Mean numbers of leaves per plant and treatment over time in the hydroponic glasshouse experiment 2018. The graph in figure 5.10 shows the increase in the number of leaves per plant in silicon-treated plants compared with untreated over time.

Mean numbers of leaves per plant between silicon-treated plants and untreated showed a statistical difference (P < 0.05). The paired sample t-test for Means revealed an increase in the leaf number of silicon-treated plants compared to untreated plants. Refer to appendix 66 for statistical workings.

5.8.4. Effects of silicon nutrient on average values of the number of runners (per plant and treatment) in the hydroponic experiment

Figure 5.11 and appendix 67 showed that the average values of the number of strawberry runners increased from three runners per plant in untreated to 6 runners per plant in the silicon-treated over time. A statistically significant difference (P < 0.05) using the Wilcoxon Sign Ranked test was found between the silicon-treated strawberry

plants and untreated. Silicon-treated strawberry plants had a higher number of runners than untreated (figure 5.11 and appendix 67).



Figure 5. 11. Mean numbers of runners in the hydroponic experiment 2018. The graph displays the increase in runners in the hydroponic glasshouse experiment 2018. Mean number per plant and treatment in silicon-treated plants compared to untreated.

Mean numbers of runners per plant between silicon-treated plants and untreated showed a statistical difference (P<0.05). The Wilcoxon-Sign ranked test revealed an increase in the number of runners of silicon-treated plants compared to untreated plants. Refer to appendix 67 for statistical workings.

5.8.5. Effects of silicon nutrient on the number of flowers (per plant and treatment) in the hydroponic experiment

Flowering was first observed in silicon-treated plants. Two silicon-treated plants started (out of ten plants) flowering on 15 May 2018, a week before the untreated, which flowered on 22 May 2018. Flowering and fruiting ended in both treatments by 27 June 2018 (figure 5.12).



Figure 5. 12. Flowering to fruiting period in the hydroponic experiment 2018

5.8.6. Effect of silicon nutrient on the average values of the number of fruits in the hydroponic experiment

Statistical tests using the Wilcoxon-Sign Ranked test between average values of silicontreated and untreated fruit numbers revealed a statistical difference (P < 0.05) between both treatments. It suggests that there were more fruits in the silicon-treated plants than untreated. Refer to appendix 68 and 69 for statistical workings.

5.8.7. Effects of silicon nutrient on the °Brix levels of fruit (per treatment) in the hydroponic experiment

Appendix 70 shows individual °Brix levels readings in five different strawberry fruits from the silicon-treated and untreated. Statistics have not been applied to °Brix sampled from the hydroponics as these are individual readings and so not applicable. From the table in appendix 70, levels of fruit °Brix varied. No statistical difference is shown between the silicon-treated and untreated. Simple averages of °Brix levels from appendix 70 in untreated fruits sampled both on 13 June 2018 and 20 June 2018 read 13.5, and averages of °Brix levels in the silicon-treated fruits sampled both on 13 June 2018 and 20 June 2018 read 15. Silicon-treated fruits suggest a higher °Brix according to these averages but are not confirmed statistically.

5.8.8. Effects of silicon nutrient on average values of chlorophyll levels (per treatment) in leaves of plants grown hydroponically

Means levels of chlorophyll contents in the leaves of silicon-treated plants are compared to untreated using the paired sample for means test. Statistical tests in appendix 71 and 72 showed no statistical increase (P>0.05) in the silicon-treated plants compared to

untreated. Results showed that the chlorophyll contents in the leaves of silicon-treated strawberry plants were not significantly different (P>0.05) from untreated. Means of ten leaves per plant (ten silicon-treated and untreated measurements) were used to analyse data in untreated and silicon-treated (appendix 71 and 72).

5.8.9. Effects of silicon nutrient on the fresh weights (biomass) of whole strawberry plants grown hydroponically

Ten whole strawberry plants with leaves, roots and runners still attached from silicontreated and untreated plants were weighed at the end of the hydroponic experiment, and the raw results are shown in appendix 73. No statistical test applied to this section as the results presented in appendix 73 are based on individual fresh weights of strawberry plants in both untreated and silicon treatment.

Conclusions from simple averages using Excel (2021) between both treatments (appendix 73) revealed silicon-treated plants seemed to have weighed (average = 169g) more than untreated (average = 144g). However, these results from the data presented here are not statistically proven. Figure 5.13 is only shown as an example of a strawberry plant that was treated with silicon (left side) and one that had no silicon treatment (right side). The photo in figure 5.13 shows any difference in the structure or appearance of a plant treated with silicon and without silicon grown in hydroponics. Table 5.7 below summarises the main findings from the hydroponic experiment in 2018.

Growth parameter	Treatments	Statistical test applied	Statistical findings (see appendix 66 to 73 for
P		"PP	statistical workings
Leaves	Untreated control (no silicon) and silicon- treated	Shapiro-Wilk normality test and parametric test; Paired Samples Test	There was a statistical increase ($P < 0.05$) in the average values of the number of silicon-treated strawberry plant leaves compared to untreated. The null hypothesis (HO) is rejected.
Runners	Untreated control (no silicon) and silicon- treated	Shapiro-Wilk normality test and non-parametric test; Wilcoxon Signed Ranked Test	There was a statistical increase (P<0.05) in the average values of the number of silicon-treated plant runners compared to untreated. The null hypothesis (HO) is rejected.
Fruits	Untreated control (no silicon) and silicon- treated	Shapiro-Wilk normality test and non-parametric test; Wilcoxon Signed Ranked Test	There was a statistical increase (P<0.05) in the average values of numbers of silicon-treated plant fruits compared to untreated The null hypothesis (HO) is rejected.
Chlorophyll	Untreated control (no silicon) and silicon- treated	Shapiro-Wilk normality test and parametric test; Paired Samples Test	No statistical increase was (P>0.05) in the average values of numbers of chlorophyll contents of silicon-treated plants compared to untreated. The null hypothesis (HO) is accepted.

Table 5. 7 Summary of findings in the hydroponic experiment 2018

Table 5.7 showed the growth parameters of strawberry plants growing in hydroponics. The results revealed that silicon-treated plants had more leaves, runners and fruits (P<0.05) than untreated plants. Results also showed that silicon did not raise (P>0.05) the chlorophyll contents in the leaves of the silicon-treated strawberry plants compared to untreated. Figure 5.13 compares a silicon-treated plant and an untreated plant for visual purposes.



Figure 5. 13. Silicon-treated (left side) and Untreated (right side) individual strawberry plants from the hydroponic glasshouse experiment 2018.

The observation drawn from the sizes of the plants shown in figure 5.13 is that plants grown without silicon can be smaller than those grown with silicon. However, the significance of this photo (figure 5.13) is not quantified and cannot be generalised.

5.8.10. Observations on plants grown hydroponically without silicon and plants grown with silicon

There were no obvious symptoms, such as premature yellowing of leaves, deformed fruits or stunting in growth observed in untreated and silicon-treated strawberry plants throughout the length of the glasshouse hydroponic experiment. Strawberry plants grown without silicon were smaller in their numbers of leaves, runners and fruits compared to silicon-treated strawberry plants at the end of the experiment. The number of leaves on the plants grown without silicon was significantly lower than the number of leaves on the plants treated with silicon. See figure 5.10 for the graph of the production of leaves. The numbers of runners were significantly more in silicon-treated plants compared with untreated. See figure 5.11 for the graph of the production of runners. No significant differences (P>0.05) were found in chlorophyll levels, °Brix levels and fresh weights of strawberry plants from this experiment. Plants grown without silicon were smaller concerning their leaf and runner numbers.

5.9. Discussion

5.9.1. Effects of silicon nutrient on the °Brix levels of fruits and leaf petioles in the silicon fertigation field experiment 2017 and fruits of the Hydroponic glasshouse experiment

In the field experiment, strawberry fruits in figure 5.8 showed an increase (P<0.05) in °Brix levels of strawberries sampled on 24 July 2017 only and no increase (P>0.05) in strawberries sampled on 20 July 2017, 27 July 2017, 1 and of August 2017. Leaf petioles °Brix results showed there was no significant increase (P>0.05) in levels of °Brix in leaf petioles samples on 20 June 2017, 18 July 2017, 01 August 2017 and 19 September 2017 (figure 5.9). Jin (2015) reported that the average values of °Brix of fruits of strawberries from a field experiment were significantly higher in silicon-treated plants than untreated. Liu (2016) also found a significant increase (P<0.05) in the levels of °Brix in leaf petioles and fruits with the use of silicon.

The glasshouse hydroponic experiment showed an elevation in the sample averages of strawberry fruits °Brix between both treatments, but statistics were not applicable. However, results from the field and glasshouse experiments varied (fruits and leaf petioles). Plants in the field experiment may have been stressed under different climate conditions compared with the glasshouse experiment, which was grown in a controlled environment. As mentioned, other authors have found that strawberry fruit and leaf petiole °Brix in a field experiment is enhanced with silicon use. The work reported here moved on from the field experiment to examine both field experiments and hydroponically grown strawberry plants °Brix.

A higher °Brix of strawberry fruit is important to growers as they satisfy and meet the demands of UK supermarkets. Each plant's sucrose (sugar) content may differ, and other factors, including ripening and direct sunlight, can affect sugar levels (Cao et al., 2015). The work reported in the chapter moved on further from the work reported by Jin (2015) from assessing °Brix levels in field experiments to assessing other growth parameters, such as numbers of leaves, runners, fruits, chlorophyll contents and fresh weights of plants in the hydroponic experiment.

5.9.2. Effects of silicon nutrient on the average values of the number of leaves in the hydroponic experiment

The first assessment in the glasshouse hydroponic experiment was to determine the number of leaves in silicon-treated and untreated strawberry plants as plants matured throughout the experiment. Results showed an increase in the average values of the number of strawberry leaves in silicon-treated strawberry plants compared with untreated (figure 5.10, appendix 66). Statistical analysis using the paired-sample t-test showed that both untreated and silicon-treated plants differed significantly (P<0.05). The production of leaves is one of the signs of a healthy growing plant. The more leaves, the more benefit to the plant as plants receive their energy and sugars from their leaves. More leaves in plants are considered a benefit as this can positively affect yield by fruit production and possibly higher °Brix.

An example of silicon enhancing the number of leaves was shown in a study conducted in Pakistan on Maize in a pot experiment, which showed that silicon significantly (P<0.05) increased the number of leaves on silicon-treated Maize crops compared with untreated crops (Amin et al., 2018). Dehghanipoodeh et al. (2018) also showed that silicon foliar application treatment had a bio-stimulative effect by increasing the leaf area in strawberry plants growing in a pot experiment under natural outdoor conditions. Zahedi et al. (2020) Showed an increase in the number of leaves of strawberry plants using the addition of foliar sprays treatment of Nanoparticles (NPs) containing silicon dioxide (SiO₂) and Selenium (Se) in the cultivar Gaviota.

5.9.3. Effects of silicon nutrient on the average values of the number of runners produced in the hydroponic experiment

The second assessment in this experiment was to determine if silicon positively affected the number of runners laid down by strawberry plants growing in the hydroponics. Both silicon-treated and untreated strawberry plants produced runners. However, more runners were produced in the silicon-treated plants compared to untreated (figure 5.11, appendix 67). Growers cut off runners from mother plants to preserve the plant's strength in producing more flowers and fruits. Runners may not be useful to growers but are beneficial to plant propagators. Statistics (appendix 67) showed that silicontreated plants had more runners than untreated and both treatments were statistically different (P<0.05). This also supports the previous work by Jin (2015), which showed that adding silicon to strawberry plants growing in a glasshouse pot experiment increased the weight of runners of silicon-treated plants compared to untreated plants. Previous work did not focus on the numbers of runners but on their weights (Liu, 2016).

5.9.4. Effects of silicon nutrient on flowering in the hydroponic experiment

Flowering (in two out of ten silicon-treated plants) was a week early in silicon-treated strawberry plants compared with untreated (figure 5.12). An example of early flowering was found in an experiment conducted on *Helianthus annuus* (Sunflower) plants, which showed that early flowering and increased flower quality occurred with optimum silicon treatment compared with untreated control (Kamenidou, Cavins and Marek, 2008). Although the literature has not shown early flowering in strawberry plants, this experiment suggests that silicon can boost flowering times in strawberry plants. Figure 5.12 shows the flowering to fruiting period in the glasshouse hydroponic experiment. Results suggest that silicon-treated plants, which flowered earlier, produced fruits quicker. The more fruits were picked, the more fruits they produced. The average values in this experiment, the fruiting period in silicon-treated plants, were longer than untreated. Therefore, if silicon application can stimulate strawberry plants to flower longer, it could encourage a longer fruit production time (figure 5.12).

5.9.5. Effects of silicon nutrient on the number of fruits produced in the hydroponic experiment

There was an increase in the average values of the number of strawberry fruits from the silicon-treated strawberry plants compared with untreated (appendix 69). Fully ripe fruits counted from the silicon-treated plants were more per plant (P<0.05) than untreated. This work supports (in hydroponics) the work done by Jin (2015) and Liu (2016), which also showed a significant increase (P<0.05) in fruit yield in silicon-treated crops in a glasshouse pot experiment compared with untreated. A study in India showed that foliar silicon application significantly improved soybean growth and yield (Shwethkumari et al., 2017). Another study by Artyszak (2018) showed an increase in the number of grains per cob in Maize. Field experiments conducted in Morocco using sugar beet showed that a silicon application systematically increased root yield and leaf yield by 40% (Prentice, 2017). Results from the hydroponic experiment encouraged early fruiting from silicon-treated plants that had their first flowering.

5.9.6. Effects of silicon nutrient on chlorophyll contents in leaves of plants grown hydroponically

Chlorophyll measurement was a part of the weekly assessment in the hydroponic experiment. Results from statistical analysis using the Paired Sample test in appendix 72 showed no significant increase (P>0.05) of chlorophyll contents in the leaves of silicon-treated plants compared with untreated. A study by Silva et al. (2012) showed that silicon had a beneficial effect on the chlorophyll contents in tomato cultivars. Xie et al., 2014 also reported a positive effect of silicon on chlorophyll contents in Maize (*Zea mays L.*). Their work showed that silicon application in a field experiment increased the total chlorophyll contents of Maize leaves, which shows that the photosynthetic efficiency of Maize was increased with the use of an adequate silicon application.

Work done by Dehghanipoodeh et al. (2018) showed that silicon significantly (P < 0.05) improved the chlorophyll contents of strawberry plants in silicon-treated plants in comparison to untreated plants. Chlorophyll in photosynthesis help plants obtains energy from sunlight (Kobayashi and Masuda, 2019). The more chlorophyll in the leaves, the more sugars they can produce, which in turn may enhance °Brix levels leading to an overall benefit to the plant. More leaves and chlorophyll contents can encourage the production of sugars in leaves, which are transported to the fruit.

5.9.7. Effects of silicon nutrient on the fresh weights (biomass) of plants grown hydroponically

At the end of the experiment, fresh weights (averages, appendix 73) of strawberry plants suggests that silicon-treated plants weighed more than untreated plants. However, this was based on individual whole plants, and statistics were not applicable here and so were not confirmed. Jin (2015) found that both fresh and dry weights of whole strawberry plants (including roots, tops and runners) had greater biomass (P<0.05) in silicon-treated plants than untreated in a glasshouse experiment. Liu (2015) also showed higher strawberry crop biomass of dry and fresh weights in silicon-treated plants (P<0.05) compared with untreated in a glasshouse experiment. Li et al. (2018) reported that silicon significantly increased (P<0.05) the biomass of the cotton plant. Luyckx et al. (2017) showed that fresh leaf weight of hemp plants treated with silicon had larger biomass than untreated. Biomass is important for crop yield and development (Ayal et al., 2019).

Another example is a study conducted in India, which showed that silicon use on peppers significantly (P < 0.05) increased plant height and biomass production (Artyszak, 2018). A study on potato cultivation in Israel showed that silicon fertilization increased the average tuber weight, dry tuber weight, tuber yield and larger fresh weight of potato plants (Vulavala et al., 2016). Amin et al. (2018) reported that silicon applied to Maize crops in a pot experiment increased the number of grains, length of cob and plant height (biomass).

5.9.8. Observations on the health of plants grown hydroponically with and without silicon

In the hydroponic experiment, observations from strawberry plants showed no symptoms of stunting in growth, yellowing of leaves or deformed fruits occurred in untreated plants. However, untreated plants had fewer leaves, fewer runners and fewer fruits. Statistics using the paired-sample t-test and Wilcoxon-Sign Ranked test revealed that silicon-treated strawberry plants significantly increased (P<0.05) in the numbers of silicon-treated leaves, runners and fruits compared to untreated. Still, no significant increase (P>0.05) was found in the chlorophyll contents of leaves compared with untreated. The results in table 5.8 showed the effects of silicon on plants growing in hydroponics in a glasshouse experiment. The symptoms seen of plants not given any silicon in the hydroponic experiment is that they were smaller, as shown by their leaf, runners and numbers of fruits.

5.10. Conclusion

This chapter showed that silicon improved the quality of strawberry plants growing in a hydroponic glasshouse experiment by increasing their numbers of leaves, and runners, stimulating early flowering and fruit number of silicon-treated plants compared with untreated control.

Previous work done by Jin (2015) and Liu (2016) on the benefits of silicon, particularly in strawberry plants, showed that silicon increased the °Brix levels in the fruits of silicon-treated strawberry plants compared with untreated. Disease reduction with silicon used in a two-year experiment and silicon deposition experiment in a one-year glasshouse experiment using fewer samples, which were not quantified, was reported by Jin (2015). However, cultivars explored the work reported in this thesis differently. Experiments were conducted in glasshouse (two years) and field (three years) experiments. A deposition experiment was replicated in a glasshouse and field experiment, which was quantified. Additionally, a hydroponic experiment was set up following these experiments to investigate the growth parameters of strawberry plants growing in Hoagland's solution. This experiment was specific to this thesis and did not carry on from work shown in Jin's thesis.

The work reported in this thesis chapter investigated further by assessing other growth parameters (besides fruit °Brix and biomass as shown by other authors) such as the numbers of leaves, runners, fruits, chlorophyll contents in the leaves, particularly in strawberry plants. As the literature shows, silicon does encourage growth parameters, such as °Brix and biomass. However, this chapter investigated strawberries' overall plant growth parameter and found other benefits to using silicon in growing strawberries, including fruit °Brix. The overall performance of the plant is enhanced. The aims and objectives that silicon can improve growth parameters of strawberry plants are met by the numbers of leaves, runners and fruits in silicon-treated plants. However, not by their chlorophyll contents of the leaves, °Brix of fruits and fresh weights biomass.

Chapter 6 - Overall discussion and conclusion

This chapter covers i) Overall discussion and conclusion of experimental chapters 3 to 5 and ii) Future work.

The key aim of this thesis was to examine the effects of the silicon nutrients delivered through the fertigation system on the development of strawberry powdery mildew (Podosphaera aphanis) disease levels in different strawberry plant cultivars in chapter 3, to examine the amounts and pattern of distribution of silicon in leaves, leaf petioles and roots of strawberry plants in a glasshouse and field experiments in chapter 4, to evaluate °Brix levels of silicon-treated fruits and leaf petioles with untreated fruits and leaf petioles, to measure growth parameters of strawberry plants in the absence and presence of silicon in a glasshouse hydroponic experiment in chapter 5. The research work reported in this thesis has shown that the application of silicon nutrients through the roots of strawberry plants once and twice a week via a fertigation system reduced disease severity in a different strawberry plant cultivar, Malling Centenary, in 2016. Secondly, the work revealed in chapter 4 that the silicon administered to strawberry plants is laid down regularly in the leaves, leaf petioles, and roots. This deposition was quantified (amounts) through fluorescence intensity measurements for two consecutive years in a glasshouse 2017 and field experiments 2017 and 2018. Lastly, the work reported in this thesis also found that silicon can enhance strawberry plant growth features, such as their leaves, runners and fruit number. This thesis's aims, objectives and findings are discussed in the various sections listed here.

6.1. Effects of silicon on powdery mildew disease *Podosphaera aphanis* on Malling Centenary and Amesti Cultivar

A fundamental part of the study was to examine whether the effects of silicon, when applied to different cultivars of strawberries, have similar effects of working shown by other authors mentioned in this thesis. Cultivars used in this study were the Malling Centenary plants in 2016 (figure 3.1) and Amesti in 2017 and 2018 (figures 3.4 and 3.6). The main findings from this experiment were that the use of silicon treatment as nutrients in the fertigation system caused a delay in the start of levels of disease by 29 days (figure 3.1, appendix 14). The work in this thesis revealed that silicon applied once and twice a week with and without fungicides reduced levels of disease (P < 0.05) in a 2016 fertigation field experiment, which meets the first aim of this thesis, to examine whether the addition of silicon through a fertigation system can reduce disease severity of strawberry powdery mildew *Podsophaera aphanis* on a different cultivar, Malling Centenary.

Examples of disease level reduction with the addition of silicon in strawberry plants were reported by Jin (2015) using different cultivars, such as Elegance, Alexandra, Shelley and Sonata. Another example is shown by Liu (2016), also using the strawberry cultivar Sonata, which caused a reduction in the severity of disease levels (P < 0.05). The work of Jin (2015) and Liu (2016), combined with the work in the thesis, have shown disease reduction in 5 consecutive years. This disease reduction has been shown on six different strawberry plant cultivars (Malling Centenary, Amesti, Elegance, Alexandra, Shelley and Sonata). Jin (2015) and Liu (2016) showed that in silicontreated strawberry plants, there was a delay in the onset of disease development by more than two weeks. However, this experiment showed that adding silicon with and without fungicides caused disease reduction (2016) and a much longer delay in building disease levels by 29 days.

These consistent results suggest that using silicon nutrients in the fertigation tubes is a useful part of an integrated disease control programme showing both delays in the start of disease build-up and overall disease reduction. The research reported in the literature has used a variety of forms of silicon and a variety of methods of application. The 'Sirius' (see product description in chapter 2, section 2.1.3 and appendix 5) used in this thesis does not contain potassium or selenium. This bioavailable form of silicon can be used safely in fertigation tubes. Other positive effects of silicon in reducing diseases on other crops are found in chapter 1, table 1.7 and 1.8. Whilst most recorded disease reduction is in crops that are silicon accumulators, reduced levels of powdery mildew were also found in melon (a non-silicon accumulator) using potassium silicate (Barker and Pilbeam, 2006). Results from this experiment have shown that the aim and objectives of the chapter were met in 2016 with strawberry powdery mildew (*Podosphaera aphanis*) disease reduction using a different cultivar from previous work.

6.2. Silicon deposition in strawberry plants cultivar (passive defence pathway)

This research shows that silicon was preferentially deposited in the leaves, leaf petioles and roots of silicon-treated Malling Centenary strawberry plants in a glasshouse pot experiment. This deposition of silicon was laid down in a regular pattern, and the silicon quantification analysis revealed that more silicon (P<0.05) was deposited in the cuticle, epidermis, and palisade layer of treated strawberry leaves compared to untreated strawberry leaves. In the leaf petioles, more deposits of silicon (P<0.05) were laid down mainly in the xylem of the treated petioles compared to untreated petioles. In the roots, there were more deposits of silicon (P<0.05) accumulated in the xylem of silicontreated roots compared to untreated roots (figure 4.2, appendix 25 to 33 and 60).

Previously, a glasshouse work reported by Jin (2015) showed that in the cultivar Shelley, there were more silicon deposits in the epidermis and vascular tissue of the leaf and leaf petiole of a silicon-treated plant. Jin (2015) reported this effect was not

replicated by cross-sections or quantified by the amounts of silicon deposited in the plants. The work was assessed by eye from a glasshouse experiment. The glasshouse experiment in this chapter was quantified and showed a consistent modification of the plant morphology using this silicon.

Furthermore, to the study reported from the glasshouse experiment, field experiments of silicon deposition assessments in 2017 showed that they were also laid down in a regular pattern in silicon-treated plants compared to untreated. The results showed that more deposits (P<0.05) of silicon were found in the cuticle, epidermis, and palisade layer of treated strawberry leaves compared to untreated leaves. More deposits of silicon (P<0.05) were laid down in the xylem of the treated petioles compared to untreated petioles. More silicon was also found laid down in the silicon-treated roots (P<0.05) compared to untreated (figure 4.3, appendix 34 to 42 and 61). The 2018 field study showed that more silicon deposits (P<0.05) were laid down in a regular pattern in the cuticle, epidermis, and palisade layer of treated strawberry leaves compared to untreated leaves. However, there was no significant difference (P>0.05) between the silicon-treated and untreated plants in the petioles and roots. (figure 4.4, appendix 43 to 51 and 62).

Debona et al. (2017) showed in an experiment using rice (silicon accumulator) on the disease rice blast. They hypothesized that the change in morphology caused by the silicon deposits reduced infection by rice blast. See diagram in chapter 1, figure 1.23. The work reported here and shown in chapter 4 suggests a similar modification in strawberry plants (a silicon non-accumulator). The disease reduction shown in chapter 3 could be caused by the morphological modifications induced by the silicon deposition (see future work section). This suggests that silicon can enhance the passive defence pathway of strawberry plants. These experiments reported demonstrate the pattern and quantification of silicon in the leaves, petioles and roots of strawberry plants both in a glasshouse and field experiments for 2 consecutive years, thus meeting the aims and objectives of the chapter in the 2017 glasshouse experiment and 2017 field experiment, which was to examine the amounts and pattern of distribution of silicon in leaves, leaf petioles and roots of strawberry plants. Future work should investigate the link between silicon enhancing the passive defence pathway and reducing disease susceptibility to strawberry powdery mildew *Podosphaera aphanis*.

6.3. Benefits of silicon nutrients use on strawberries

This research study also measured the effects of silicon on the growth parameters of strawberry plants growing in hydroponics. This experiment revealed that the numbers of leaves, runners and fruits of strawberry plants treated with silicon were significantly more than (P<0.05) untreated plants. An example of some benefits linked with silicon use is shown by Zahedi et al. (2020) by using foliar sprays of Nanoparticles (NPs) containing silicon dioxide (SiO₂) and Selenium (Se) (Se/SiO₂-NPs (100 mg L⁻¹) on

strawberry plants. These benefits included increased area and numbers of leaves, chlorophyll content, reduced cracked fruits, enhanced vitamin c content and fruit firmness. In this experiment, it is impossible to distinguish between the effects of silicon, selenium and the use of nanoparticles. However, the study reported here compared silicon-treated plants and untreated plants. The main findings of the experiment revealed that silicon-treated plants had increased (P<0.05) numbers of leaves (figure 5.10 and appendix 66), runners (figure 5.11 and appendix 67) and fruits (appendix 69) compared to untreated plants. Other improved factors included early flowering observed in some plants from the silicon-treated plants compared to untreated (figure 5.12). Fresh weights (averages) based on individual whole strawberry plants showed that silicon-treated plants seemed to have weighed more than untreated plants. However, this is not statistically proven (appendix 73). An increase in the yield of fruit, shelf-life and vegetative growth in the strawberry plants was reported by Peri-Felipo (2020).

Some growers have reported toxicity effects when using potassium silicate (personal communications from grower). This includes albino fruits. However, Maas (2004) only reports albino fruits with the excess use of potassium. At the concentration used (0.017%), no toxicity symptoms were observed in this thesis, with the bioavailable form of silicon (not containing potassium or sodium) used throughout the experiments. No toxicity symptoms were found in work reported by Jin (2015) and Liu (2016). Kamenidou, Cavins and Marek (2008) showed that growth abnormalities were observed when concentrations of 100 and 200 mg \cdot L-1 Si were supplied as potassium silicate (KSiO3) in Gerbera and Sunflower. These high levels of potassium silicate were found to deform flowers.

The literature suggests that strawberry growers are advised not to use more than 22mg/L of silicon. Although growers are frightened of albinism, the academic literature also suggests that this whiteness in strawberries is caused by excess potassium, calcium and nitrogen (Lieten, Horvath and Asard, 2000) rather than silicon. This experiment suggests that the aim to measure the growth parameters of strawberry plants grown in the presence and absence of silicon was met through the increased numbers of leaves, runners and fruits of silicon-treated plants compared to untreated. The study revealed that the aims and objectives were met through the benefits of using silicon in growing strawberry plants in a hydroponic experiment in the presence and absence of silicon.

6.4. Conclusion

The effects of the silicon (bioavailable form) shown in this thesis include higher strawberry crop yield and quality and disease reduction probably mediated via the passive defence pathway. Therefore, the recommendation for growers from this thesis is to incorporate the regular use of silicon as a nutrient via fertigation as part of an integrated management plan to optimise strawberry production.

6.5. Future work

This research thesis has summarised following future work to follow on from experiments reported here and can investigate the following.

a) <u>The link between silicon nutrients enhances the passive defence pathway and</u> reduces disease susceptibility.

To elucidate silicon's effects in enhancing the passive defence pathway using light microscopy and electron microscopy analysis. The experiment would quantify the penetration and subsequent growth of *Podosphaera aphanis* in treated and untreated strawberry plants.

b) Effects of silicon on yield quality and shelf life of strawberries

Strawberry growers are under pressure to produce more class 1 fruits and reduce fruit waste, thus reducing the waste of class 2 fruits (there is no real market for class 2 fruits). This thesis and unpublished work suggest that using silicon nutrients may raise the number of class 1 fruits and increase shelf life. This work would evaluate strawberries harvested from silicon-treated and untreated plants for their shelf life and examine if they were less susceptible to post-harvest decay. Work can also examine whether silicon may affect the firmness of strawberry fruits, thus reducing fruit bruising concerning strawberry fruit picking and transport.

c) Further investigations of the cause of albinism in strawberries

Further work can examine the effects of silicon concentration on strawberries and the interaction of silicon, potassium, calcium and nitrogen. This experiment can investigate the causes of albinism as the literature suggests that there could be a link between high levels of silicon use potentially leading to whiteness in strawberries. The methods for this experiment will include growing strawberry plants in hydroponics containing Hoagland's solution in a glasshouse experiment.

As Hoagland's contains no silicon but contains all the necessary nutrients essential for optimal plant growth, treatment for the hydroponic experiments can include plants growing in Hoagland's alone (untreated) plants growing with a regular weekly dose of silicon application and plants growing with excessive amounts of silicon application, which could be applied once or twice-a-week. This experiment will monitor and observe the growth and development of the strawberry plants throughout their life cycle. During the fruiting season, strawberries will be harvested from all strawberry plants in the hydroponic experiment to assess their colour, flavour and firmness for signs of albinism.

REFERENCES

Abbey, J. A., Percival, D., Abbey, L., Asiedu, S. K., Prithiviraj, B., & Schilder, A. (2019). Biofungicides as alternative to synthetic fungicide control of grey mould (Botrytis cinerea)–prospects and challenges. *Biocontrol science and technology*, 29(3), 207-228.

Adams, T. M. (2019). Pathogenomics of Phytophthora fragariae, the causal agent of strawberry red core disease (Doctoral dissertation, University of Reading).

AHDB.(2011).Optimal ranges for silicon. Retrieved 22, April 2021 fromhttps://ahdb.org.uk/knowledge-library/strawberry-analysis-chart_.

Akoglu, H. (2018). User's guide to correlation coefficients. Turkish journal of emergency medicine, 18(3), 91-93.

Alchetron. (2020). Phytophthora cactorum. Retrieved 2, December 2020 fromhttps://alchetron.com/Phytophthora-cactorum.

Alchetron. (2020). Phytophthora fragariae. Retrieved 2, December 2020 fromhttps://alchetron.com/Phytophthora-fragariae.

Alhousari, F. and M. Greger (2018). "Silicon and mechanisms of plant resistance to insect pests." Plants 7(2): 33.

Allen, G., Halsall, C. J., Ukpebor, J., Paul, N. D., Ridall, G., & Wargent, J. J. (2015). Increased occurrence of pesticide residues on crops grown in protected environments compared to crops grown in open field conditions. *Chemosphere*, *119*, 1428-1435.

Almanac.(2020). Availability of Nutrients at Varying pH Values. Retrieved November, 9, 2020 from*https://www.almanac.com/preparing-soil-planting*.

Amin, M., Ahmad, R., Ali, A., Hussain, I., Mahmood, R., Aslam, M., & Lee, D. J. (2018). Influence of silicon fertilization on maize performance under limited water supply. Silicon, 10(2), 177-183.

Anderson, H. C., Rogers, M. A., & Hoover, E. E. (2019). Low Tunnel Covering and Microclimate, Fruit Yield, and Quality in an Organic Strawberry Production System. HortTechnology, 1(aop), 1-9.

Anjom, F. K., Vougioukas, S. G., & Slaughter, D. C. (2018). Development of a linear mixed model to predict the picking time in strawberry harvesting processes. Biosystems Engineering, 166, 76-89.

Aristya, G. R., Alyza, R., Khoiroh, R., & Daryono, B. S. (2015). chromosome number and time period of mitotic cycle of festival and californica strawberry cultivars (Fragaria x ananassa and Fragaria vesca). KnE Life Sciences, 476-479.

Ariza, M., et al. (2016). "Strawberry achenes are an important source of bioactive compounds for human health." International journal of molecular sciences 17(7): 1103.

Artyszak, A. (2018). Effect of silicon fertilization on crop yield quantity and quality— A literature review in Europe. Plants, 7(3), 54.

Asalf, B., Gadoury, D. M., Tronsmo, A. M., Seem, R. C., Cadle-Davidson, L., Brewer, M. T., & Stensvand, A. (2013). Temperature regulates the initiation of chasmothecia in powdery mildew of strawberry. *Phytopathology*, *103*(7), 717-724.

Atlasbig. (2018). *Top strawberry producing countries*. Retrieved March 2, 2018 fromhttps://www.atlasbig.com/en-us/countries-strawberry-production_.

Autunović, A. (2018). Growing measures for strawberry Malwina in a protected cultivation (Doctoral dissertation, Agronomski fakulet, Sveučilište u Zagrebu.

Ayal, S., Vinocur, B. J., Diber, A., Karchi, H., & Poraty-Gavra, L. (2019). U.S. Patent Application No. 16/243,226.

Ballard, A. E. (2018). June-bearing Strawberry Production with Low Tunnels in Organically Managed High Tunnel (Doctoral dissertation, North Carolina Agricultural and Technical State University).

Bankhead, P. (2014). Analyzing fluorescence microscopy images with ImageJ. ImageJ, 1(195), 10-1109.

Bano, A., & Qureshi, K. M. (2016). Research Article Response of Strawberry (Frageria annanasa) cv. Chandler to Different Mulching Materials Muhammad Sudheer Tariq Department of Biological Sciences, Quaid-I-Azam University, Islamabad, Pakistan. Science, 35(3).

barker, a. v., pilbeam, d., & brown, p. (2006). Nickel. Handbook of plant nutrition, 395-409.

Binder, A., & Parniske, M. (2018). The nuclear pore complex in symbiosis and pathogen defence. Annual Plant Reviews online, 229-254.

Braun, U. (1995). The powdery mildews (Erysiphales) of Europe. VEB Gustav Fischer Verlag.

Braun, U. (2011). The current systematics and taxonomy of the powdery mildews (Erysiphales): an overview. Mycoscience, 52(3), 210-212.

Braun, U., & Cook, R.T.A., (2012). Taxonomic manual of the Erysiphales (Powdery Mildews), cbs-gnaw fungal biodiversity Centre, Utrecht.

Braun, U., Takamatsu, S., Heluta, V., Limkaisang, S., Divarangkoon, R., Cook, R., & Boyle, H. (2006). Phylogeny and taxonomy of powdery mildew fungi of Erysiphe sect. Uncinula on Carpinus species. Mycological Progress, 5(3), 139-153.

Brown-Paul, C. (2016). An expanding business. *Practical Hydroponics and Greenhouses*, (166), 22.

Calleja, E. J., Ilbery, B., & Mills, P. R. (2012). Agricultural change and the rise of the British strawberry industry, 1920–2009. Journal of Rural Studies, 28(4), 603-611.

Calvano, A., Izuora, K., Oh, E. C., Ebersole, J. L., Lyons, T. J., & Basu, A. (2019). Dietary berries, insulin resistance and type 2 diabetes: an overview of human feeding trials. Food & function, 10(10), 6227-6243.

Campos, N. F. (2018). Brexit and migration: why do rocket scientists pick strawberries?. LSE Brexit.

Cao, F., Guan, C., Dai, H., Li, X., & Zhang, Z. (2015). Soluble solids content is positively correlated with phosphorus content in ripening strawberry fruits. Scientia Horticulturae, 195, 183-187.

Castle, D., Grass, I., & Westphal, C. (2019). Fruit quantity and quality of strawberries benefit from enhanced pollinator abundance at hedgerows in agricultural landscapes. Agriculture, Ecosystems & Environment, 275, 14-22.

Chowdhury, J., Coad, B. R., & Little, A. (2018). Cell Wall Responses to Biotrophic Fungal Pathogen Invasion. Annual Plant Reviews online, 1001-1030.

Claire, D., et al. (2018). "High productivity of soilless strawberry cultivation under rain shelters." Scientia Horticulturae 232: 127-138.

Cockerton, H. M., et al. (2018). "Identification of powdery mildew resistance QTL in Fragaria x ananassa." bioRxiv: 196485.

Collins, T. J. (2007). ImageJ for microscopy. Biotechniques, 43(S1), S25-S30.

Corwin, D. L., & Scudiero, E. (2020). Field-scale apparent soil electrical conductivity. *Soil Science Society of America Journal*, 84(5), 1405-1441.

Corwin, D. L., & Yemoto, K. (2019). Measurement of soil salinity: Electrical conductivity and total dissolved solids. Soil Science Society of America Journal, 83(1), 1-2.

Cross, J. V., Fitzgerald, J. D., & Down, G. (2005). Aphids and their control in strawberry. *HDC Factsheet*, 26(05).

Dai, H. (2018). "Adding Increased Value to Strawberry Puree by Adding Xylooligosaccharides to Improve Health."

Danielsen, J. (2018). Short-time effects of water sprinkling on the two-spotted spider mite predator Phytoseiulus persimilis in strawberry, Norwegian University of Life Sciences.

Darrow, G. M. (1966). The strawberry. History, breeding and physiology. The strawberry. History, breeding and physiology.

Datnoff, L. E., Rodrigues, F. A., & Seebold, K. W. (2007). Silicon and plant disease. *Mineral nutrition and plant disease*, 233-246.

Daugaard, H. (2008). Table-top production of strawberries: performance of six strawberry cultivars. *Acta Agriculturae Scandinavica Section B–Soil and Plant Science*, 58(3), 261-266.

De Preter, A., Anthonis, J., & De Baerdemaeker, J. (2018). Development of a robot for harvesting strawberries. IFAC-PapersOnLine, 51(17), 14-19.

Debona, D., Rodrigues, F. A., & Datnoff, L. E. (2017). Silicon's role in abiotic and biotic plant stresses. Annual Review of Phytopathology, 55, 85-107.

Dehghanipoodeh, S., Ghobadi, C., Baninasab, B., Gheysari, M., & Shiranibidabadi, S. (2018). Effect of silicon on growth and development of strawberry under water deficit conditions. Horticultural Plant Journal, 4(6), 226-232.

Desclés, J., Vartanian, M., El Harrak, A., Quinet, M., Bremond, N., Sapriel, G., ... & Lopez, P. J. (2008). New tools for labeling silica in living diatoms. New Phytologist, 177(3), 822-829.

Deus, A. C. F., de Mello Prado, R., Alvarez, R. D. C. F., de Oliveira, R. L. L., & Felisberto, G. (2019). Role of Silicon and Salicylic Acid in the Mitigation of Nitrogen Deficiency Stress in Rice Plants. Silicon, 1-9.

Domiciano G. P., Rodrigues F. A., Guerra A. M. N., Vale F. X. R. (2013). Infection process of *Bipolaris sorokiniana* on wheat leaves is affected by silicon. *Trop. Plant Pathol.* 38 258–263.

Downing, J. (2016). A crossroads for strawberries. California Agriculture, 70(3), 100-100.

Durner, E. F. (2018). "Plug size and density during greenhouse conditioning do not affect late summer and early fall production of June-planted 'Albion'strawberry (Fragaria X ananassa Duch.) in the plasticulture system." Scientia Horticulturae 240: 268-272.

Ellingham, O. H. (2017). Increasing accuracy of Powdery Mildew (Ascomycota, Erysiphales) identification using previously untapped DNA regions (Doctoral dissertation, University of Reading).

Epstein, E. 1994. The anomaly of silicon in plant biology. Proceedings of the National Academy of Science USA, 91, 11-17.

Faedi, W., & Baruzzi, G. (2016). Strawberry breeding. Strawberry: Growth, Development and Diseases./Eds. Husaini, A., Nari, D. CABI, 26-37.

FAO. (2020). FAOSTAT World producing strawberries. Retrieved October, 9, 2020 fromhttp://www.fao.org/faostat/en/#data/QC/visualize

Farminguk. (2019). Tremendous crop of British Strawberry on the market this year. Retrieved January, 12, 2021 from *https://www.farminguk.com/news/-tremendous-crop-of-british-strawberry-on-market-this-year_53591.html*

Farnsworth, D., Hamby, K. A., Bolda, M., Goodhue, R. E., Williams, J. C., & Zalom, F. G. (2017). Economic analysis of revenue losses and control costs associated with the spotted wing drosophila, Drosophila suzukii (Matsumura), in the California raspberry industry. Pest management science, 73(6), 1083-1090.

Fatema K. & Hall, A. M. (2012). Study of role of silicon in control of strawberry powdery mildew (*Podosphaera aphanis*) in a field trial. Aspects of Applied Biology 117, crop protection in Southern Britain, 229-234.

Fatema, K. (2014). The Effect of Silicon on Strawberry Plants and its Role in Reducing Infection by *Podosphaera aphanis*.

Fauteux, F., Rémus-Borel, W., Menzies, J. G., & Bélanger, R. R. (2005). Silicon and plant disease resistance against pathogenic fungi. *FEMS Microbiology letters*, *249*(1), 1-6.

Foito, A., et al. (2018). "Evidence for Health Benefits of Berries." Annual Plant Reviews online: 1-43.

Folta, K. M. (2019). "Breeding new varieties for controlled environments." Plant Biology 21: 6-12.

Forcelini, B. B. and N. A. Peres (2018). "Widespread Resistance to QoI Fungicides of Colletotrichum acutatum from Strawberry Nurseries and Production Fields." Plant Health Progress 19(4): 338-341.

Frew, A., Allsopp, P. G., Gherlenda, A. N., & Johnson, S. N. (2017). Increased root herbivory under elevated atmospheric carbon dioxide concentrations is reversed by silicon-based plant defences. Journal of Applied Ecology, 54(5), 1310-1319.

Fry, W. E. (2012). Principles of plant disease management. Academic Press.

Fuchs, B., & Krauss, J. (2018). Can Epichloë endophytes enhance direct and indirect plant defence?.Fungal Ecology.

Gallegos-Cedillo, V. M., Álvaro, J. E., Capatos, T., Hachmann, T. L., Carrasco, G., & Urrestarazu, M. (2018). Effect of pH and silicon in the fertigation solution on vegetative growth of blueberry plants in organic agriculture. HortScience, *53*(10), 1423-1428.

Garcia, K. P. (2016). Physiology of flowering and diurnal net photosynthetic response in American strawberry cultivars under controlled environment.

Gaudery, D. M., Asalf, B., Heidenreich, M. C., Herrero, M. L., Welser, M. J., Seem, R. C., & Stensvand, A. (2010). Initiation, development, and survival of cleistothecia of Podosphaera aphanis and their role in the epidemiology of strawberry powdery mildew. *Phytopathology*, *100*(3), 246-251.

Glawe, D. A. (2006). Synopsis of genera of Erysiphales (powdery mildew fungi) occurring in the Pacific Northwest. North American Fungi, 1(1), 1-27.

Glawe, D. A. (2008). The powdery mildews: a review of the world's most familiar (yet poorly known) plant pathogens. Annual review of phytopathology, 46.

Goodchild, M. S., Jenkins, M. D., Whalley, W. R., & Watts, C. W. (2018). A novel dielectric tensiometer enabling precision PID-based irrigation control of polytunnelgrown strawberries in coir. Biosystems Engineering, 165, 70-76.

Google maps. (2018). Map of Maltmas Farm. Retrieved, June, 2, 2018 fromhttps://www.google.com/maps/place/Maltmas+Farm/@52.6166604,-2.803304,7z/data=!4m5!3m4!1s0x47d80486bbd193f3:0x72383ce122cef3da!8m2!3d5 2.6110002!4d0.1517182.

Google maps. (2020). Map of Badfordbury. Retrieved, October, 15, 2020 fromhttps://www.google.com/maps/place/Bayfordbury,+Hertford/@51.7639005,-0.1888258,11.81z/data=!4m5!3m4!1s0x4876240283ec6655:0xbdb1f420311ab19e!8 m2!3d51.7783747!4d-0.1032544.

Graça, J. (2015). Suberin: the biopolyester at the frontier of plants. Frontiers in chemistry, 3, 62.

Graham, J., & Brennan, R. (2018). Introduction to the Rubus Genus. In Raspberry (pp. 1-16). Springer, Cham.

Greger, M., Landberg, T., & Vaculík, M. (2018). Silicon influences soil availability and accumulation of mineral nutrients in various plant species. Plants, 7(2), 41.

Grimstad, L., & From, P. J. (2017). Thorvald ii-a modular and re-configurable agricultural robot. IFAC-PapersOnLine, 50(1), 4588-4593.

Gröger, C., Sumper, M., & Brunner, E. (2008). Silicon uptake and metabolism of the marine diatom Thalassiosira pseudonana: solid-state 29Si NMR and fluorescence microscopic studies. Journal of structural biology, 161(1), 55-63.

Hafia-group.(2020). Crop Guide: Strawberry Fertilizer Recommendations. Retrieved November, 8, 2020 from https://www.haifa-group.com/cropguide/vegetables/strawberry-fertilizer/crop-guide-strawberry-fertilizerrecommendations.

Hajiboland, R., Moradtalab, N., Eshaghi, Z., & Feizy, J. (2018). Effect of silicon supplementation on growth and metabolism of strawberry plants at three developmental stages. New Zealand journal of crop and horticultural science, 46(2), 144-161.

Hall, A. M., & Jin, X. (2016, August). Integrated control of strawberry powdery mildew. In VIII International Strawberry Symposium 1156 (pp. 771-776).

Hall, A.M., Jin X & Dodgson, J. (2016). Control of strawberry powdery mildew under protection. Project SF 62 & SF 62a. Factsheet 17/08HDC.

Hancock, J. (2018). Strawberry plant named 'redstart', Google Patents.

Harrill, R. (1998). Using a refractometer to test the quality of fruits and vegetables. Publishing, Éd.

HE, R., & FU, Y. (2018). Control Effect of Pyrethrin and Matrine on Strawberry Red Spider in Greenhouse. Journal of Shanxi Agricultural Sciences, (3), 34.

Hennig, E. I., & Mazzi, D. (2018). Spotted wing drosophila in sweet cherry orchards in relation to forest characteristics, bycatch, and resource availability. Insects, 9(3), 118.

Hernán, G., et al. (2019). "Herbivory and resource availability shift plant defense and herbivore feeding choice in a seagrass system." Oecologia: 1-14.

Hershey, D. R. (1994). Solution culture hydroponics: history & inexpensive equipment. The American Biology Teacher, 56(2), 111-118.

Heuvelink, E., & Voogt, W. (2019). Adding silicon to nutrient solution is worth considering. In Greenhouses: the international magazine for greenhouse growers, 8(1), 16-17.

Hill, G. N., Henshall, W. R., & Beresford, R. M. (2017). Manipulating rainfall to study symptom expression of Botrytis cinerea infection in wine grapes. New Zealand Plant Protection, 70, 301-309.

Hoagland and Arnon (1950). The water-culture method for growing plants without soil. Berkeley, Calif.: University of California, College of Agriculture, Agricultural Experiment Station.

Hobbie, S. E. (2015). Plant species effects on nutrient cycling: revisiting litter feedbacks. Trends in ecology & evolution, 30(6), 357-363.

Hodgkiss, D., Brown, M. J., & Fountain, M. T. (2018). Syrphine hoverflies are effective pollinators of commercial strawberry. Journal of pollination ecology, 22.

Hoffmann, J., Berni, R., Hausman, J. F., & Guerriero, G. (2020). A Review on the Beneficial Role of Silicon against Salinity in Non-Accumulator Crops: Tomato as a Model. Biomolecules, 10(9), 1284.

Hoskins, D. R. (2018). Optimizing a Trapping System to Monitor Spotted Wing Drosophila in Organic Primocane Fruiting Blackberries.

Hothem, S. D., et al. (2003). "Photochemistry in Hoagland's nutrient solution." Journal of plant nutrition 26(4): 845-854.

Husaini, A. M., & Neri, D. (Eds.). (2016). Strawberry: Growth, development and diseases. CABI.

Ichijima, K. (1926). Cytological and genetic studies on Fragaria. Genetics, 11(6), 590.

Jacobs, A. H. (1957). Strawberry hemangiomas—the natural history of the untreated lesion. California medicine, 86(1), 8.

Jana, S., & Jeong, B. (2014). Silicon: the most under-appreciated element in horticultural crops. Trends in Horticultural Research, 4(1), 1-19.

Jensen, E. C. (2013). Quantitative analysis of histological staining and fluorescence using ImageJ. The Anatomical Record, 296(3), 378-381.

Jin X & Hall A. M (2012). Development and maturation of the chasmothecia of Podosphaera aphanis on strawberry. Aspects of Applied Biology 1117, Crop Protection in Southern Britain, 235-239.

Jin X (2015). Epidemiology and control of powdery mildew (Podoshaera aphanis) on strawberry. Unpublished doctoral thesis. Hatfield: University of Hertfordshire.

Johnson, S. N., Ryalls, J. M., Gherlenda, A. N., Frew, A., & Hartley, S. E. (2018). Benefits from below: silicon supplementation maintains legume productivity under predicted climate change scenarios. Frontiers in plant science, 9, 202.

Jubileestrawberries. (2020). Queen of the crop. Retrieved November, 13, 2020 fromhttps://www.jubileestrawberries.co.uk/our-story/.

Júnior, M., de Oliveira, S., Andrade, J. R. D., Reis, L. S., Andrade, L. R. D., & Gonçalves, A. C. D. M. (2018). Soil management and mulching for weed control in cowpea. Pesquisa Agropecuária Tropical, 48(4), 453-460.

Kabaktepe, S., Akata, I., Siahaan, S. A., Takamatsu, S., & Braun, U. (2017). Powdery mildews (Ascomycota, Erysiphales) on Fontanesia phillyreoides and Jasminum fruticans in Turkey. Mycoscience, 58(1), 30-34.

Kablan, L., Lagauche, A., Delvaux, B., & Legrve, A. (2012). Silicon reduces black sigatoka development in banana. Plant disease, 96(2), 273-278.

Kamenidou, S., Cavins, T. J., & Marek, S. (2008). Silicon supplements affect horticultural traits of greenhouse-produced ornamental sunflowers. HortScience, 43(1), 236-239.

Kennedy, J. W. (2016). U.S. Patent Application No. 15/049,413.

Kessel, D. (2018). Potentialities of chitosan for strawberry. Uses in the improvement and conservation of the fruits. Cultivos Tropicales, 39(1), 134-142.

Khodaparast, S. A. (2016). Molecular identification of some anamorphic powdery mildews (Erysiphales) in Guilan province, north of Iran. Mycologia Iranica, 3(2), 127-133.

Kim, D. R., Cho, G., Jeon, C. W., Weller, D. M., Thomashow, L. S., Paulitz, T. C., & Kwak, Y. S. (2019). A mutualistic interaction between Streptomyces bacteria, strawberry plants and pollinating bees. Nature communications, 10(1), 1-10.

Kobayashi, K., & Masuda, T. (2019). Transcriptional control for the chlorophyll metabolism. Metabolism, Structure and Function of Plant Tetrapyrroles: Control

Mechanisms of Chlorophyll Biosynthesis and Analysis of Chlorophyll-Binding Proteins, 133.

Korkmaz, A., Karagöl, A., Akınoğlu, G., & Korkmaz, H. (2018). The effects of silicon on nutrient levels and yields of tomatoes under saline stress in artificial medium culture. Journal of Plant Nutrition, 41(1), 123-135.

Korres, N. E., et al. (2018). Weed Control: Sustainability, Hazards, and Risks in Cropping Systems Worldwide, CRC Press.

Kumar, R. (2018). Organic production of strawberry: A review. IJCS, 6(3), 1231-1236.Kumar, V., Sahay, S., Mir, H., Ahmad, F., & Rashmi, K. (2019). Effectiveness of Plant Growth Regulators on Vegetative Growth of Strawberry cv. Camarosa. Int. J. Curr. Microbiol. App. Sci, 8(7), 1259-1264.

Kupfer, T. R., & Fessler, D. M. (2018). Ectoparasite defence in humans: relationships to pathogen avoidance and clinical implications. Philosophical Transactions of the Royal Society B: Biological Sciences, 373(1751), 20170207.

Lamb, N., & Chuah, M. C. (2018, December). A Strawberry Detection System Using Convolutional Neural Networks. In 2018 IEEE International Conference on Big Data (Big Data) (pp. 2515-2520). IEEE.

Legard, D. (2018). Dietary supplements containing dehydrated strawberry, Google Patents.

Lewers, K. S., et al. (2019). "'Keepsake'Strawberry." HortScience 54(2): 362-367.

Li, M., Li, X., Han, C., Ji, N., Jin, P., & Zheng, Y. (2019). UV-C treatment maintains quality and enhances antioxidant capacity of fresh-cut strawberries. Postharvest Biology and Technology, 156, 110945.

Li, Z., Delvaux, B., Yans, J., Dufour, N., Houben, D., & Cornelis, J. T. (2018). Phytolith-rich biochar increases cotton biomass and silicon-mineralomass in a highly weathered soil. Journal of Plant Nutrition and Soil Science, 181(4), 537-546.

Liang, Y., Nikolic, M., Bélanger, R., Gong, H., & Song, A. (2015). Silicon and Plant– Pathogen Interactions. In *Silicon in Agriculture* (pp. 181-196). Springer Netherlands.

Liang, Y., Wong, J. W. C., & Wei, L. (2005). Silicon-mediated enhancement of cadmium tolerance in maize (Zea mays L.) grown in cadmium contaminated soil. Chemosphere, 58(4), 475-483.

Lieten, P., Horvath, J., & Asard, H. (2000). Effect of silicon on albinism of strawberry. In IV International Strawberry Symposium 567 (pp. 361-364).

Liu, B. (2016). Sustainable strawberry production and management including control of strawberry powdery mildew (Doctoral dissertation, University of Hertfordshire).

Llanos, A., & Apaza, W. (2018). Antifungal activity of five chemical and two biological fungicides for the management of Botrytis cinerea, causal agent of Gray Mold in Strawberry. Peruvian Journal of Agronomy, 2(1), 1-8.

Luyckx, M., Hausman, J. F., Lutts, S., & Guerriero, G. (2017). Impact of silicon in plant biomass production: focus on bast fibres, hypotheses, and perspectives. Plants, 6(3), 37.

Lyaruu, H. V. (2010). The influence of soil characteristics on plant species diversity and their distribution patterns in Western Serengeti, Tanzania. Agricultural Journal Year, 5(3), 234-241.

Ma, J. F., & Yamaji, N. (2006). Silicon uptake and accumulation in higher plants. *Trends in plant science*, *11*(8), 392-397.

Ma, J. F., Mitani, N., Nagao, S., Konishi, S., Tamai, K., Iwashita, T., & Yano, M. (2004). Characterization of the silicon uptake system and molecular mapping of the silicon transporter gene in rice. Plant physiology, 136(2), 3284-3289.

Ma, J. F., Miyake, Y., & Takahashi, E. (2001). Silicon as a beneficial element for crop plants. Studies in plant Science, 8, 17-39.

Maas, J. L. (2004). Strawberry disease management. Diseases of Fruits and Vegetables: Volume II, 441-483.

Maas, J. L. (Ed.). (1998). Compendium of strawberry diseases (Vol. 98). St. Paul, MN: APS press.

Marodin, J. C., Resende, J. T., Morales, R. G., Silva, M. L., Galvão, A. G., & Zanin, D. S. (2014). Yield of tomato fruits in relation to silicon sources and rates. Horticultura Brasileira, 32(2), 220-224.

Marshall, J. W., et al. (2018). Tobacco products and processes, Google Patents.Mattson, N., Daughtrey, M., & Lamb, E. (2010). Investigating silicon nutrition to decrease Pythium root rot severity in snapdragons and New Guinea impatiens.

Maurstad, A. (2018). Food Fight: Violence and Exploitation in Fruit Still Life. essay: critical writing at pomona college, 2(2), 6.

McCartney, L. and M. Lefsrud (2018). "Protected Agriculture in Extreme Environments: A Review of Controlled Environment Agriculture in Tropical, Arid, Polar, And Urban Locations."

Meeboon, J., & Takamatsu, S. (2015). Erysiphe viburni-plicati and Podosphaera photiniae, two new species of Erysiphales (Ascomycota) from Japan. mycoscience, 56(1), 14-23.

Meeboon, J., & Takamatsu, S. (2017). Notes on powdery mildews (Erysiphales) in Thailand III. Erysiphe species on Fabaceae, Fagaceae, Hydrangeaceae and Lamiaceae. Tropical Plant Pathology, 42(4), 239-249.

Meeboon, J., Hidayat, I., & Takamatsu, S. (2016). Notes on powdery mildews (Erysiphales) in Thailand I. Podosphaera sect. Sphaerotheca. Plant Pathol Quar, 6, 142-174.

Meeboon, J., Kokaew, J., & Takamatsu, S. (2018). Notes on powdery mildews (Erysiphales) in Thailand V. Golovinomyces. Tropical Plant Pathology, 43(3), 202-217.

Mills-Ibibofori, T., Dunn, B., Maness, N., & Payton, M. (2019). Use of diatomaceous earth as a silica supplement on potted ornamentals. Horticulturae, 5(1), 21.

Minolta, K. (2009). Chlorophyll meter SPAD-502Plus. Konica Minolta.

Morris, J. R., & Sistrunk, W. A. (2018). The strawberry. Quality and Preservation of Fruits, 181-206.

Mvondo-She, M. A., & Marais, D. (2019). The Investigation of Silicon Localization and Accumulation in Citrus. Plants, 8(7), 200.

Nellist, C. F. (2018). Disease Resistance in Polyploid Strawberry. In The Genomes of Rosaceous Berries and Their Wild Relatives (pp. 79-94). Springer, Cham.

Nelson, P., & Wilhelm, S. (1957). Some anatomic aspects of the strawberry root. Hilgardia, 26(15), 631-642.

Nelson, S. D., et al. (2018). Strawberry plant named 'inspire', US Patent PP29,794.

Nestb, R., & Guery, S. (2017). Balanced fertigation and improved sustainability of June bearing strawberry cultivated three years in open polytunnel. Journal of berry research, 7(3), 203-216.

Neumeier, B. L., Heck, J. G., & Feldmann, C. (2019). Fluorescence-based histamine sensing with inorganic–organic hybrid nanoparticles. Journal of Materials Chemistry C, 7(12), 3543-3552.

Nichols, M. (2013). Coir: 21st century growing medium. Practical Hydroponics and Greenhouses, (138), 14.

Nollet, L. M., & Gutierrez-Uribe, J. A. (Eds.). (2018). Phenolic compounds in food: characterization and analysis. CRC Press.

Ogas, J. (2000). Gibberellins. Current Biology, 10(2), R48.

Oh, H. S., Lee, S. E., Han, C. S., Kim, J., Nam, O., Seo, S., ... & Lee, S. E. (2018). Silicon transporter genes of Fragilariopsis cylindrus (Bacillariophyceae) are differentially expressed during the progression of cell cycle synchronized by Si or light. Algae, 33(2), 191-203.

Oliveira, A. S., Braga, G. U., & Rangel, D. E. (2018). Metarhizium robertsii illuminated during mycelial growth produces conidia with increased germination speed and virulence. Fungal biology, 122(6), 555-562.

Orde, K. M. (2018). "Using Day Neutral Cultivars and Low Tunnels to Extend the Strawberry Harvest Season in the Northeastern United States."

Orde, K. M., & Sideman, R. G. (2019). Low Tunnel and Cultivar Effects on Day-neutral Strawberry Yield and Characteristics in New Hampshire. HortTechnology, 29(6), 795-810.

Ouellette, S., et al. (2017). "Silicon transporters and effects of silicon amendments in strawberry under high tunnel and field conditions." Frontiers in plant science 8: 949.

Palmers. (2018). *Top tips for growing strawberries*. Retrieved, August 11, 2018 from https://www.palmers.co.nz/top-10-tips-for-growing-strawberries/

Pančić, M., & Kiørboe, T. (2018). Phytoplankton defence mechanisms: traits and tradeoffs. Biological Reviews, 93(2), 1269-1303.

Panstruga, R., & Kuhn, H. (2019). Mutual interplay between phytopathogenic powdery mildew fungi and other microorganisms. Molecular plant pathology, 20(4), 463-470.

Parikka, P. K., Vestberg, M., Karhu, S. T., Haikonen, T., & Hautsalo, J. (2016, August). Possibilities for biological control of red core (Phytophthora fragariae). In VIII International Strawberry Symposium 1156 (pp. 751-756).

Pärtel, K. (2016). Application of ultrastructural and molecular data in the taxonomy of helotialean fungi (Doctoral dissertation).

Peris-Felipo, F. J., Benavent-Gil, Y., & Hernández-Apaolaza, L. (2020). Silicon beneficial effects on yield, fruit quality and shelf-life of strawberries grown in different culture substrates under different iron status. Plant Physiology and Biochemistry, 152, 23-31.

Peruski, A. H., Johnson III, L. H., & Peruski Jr, L. F. (2002). Rapid and sensitive detection of biological warfare agents using time-resolved fluorescence assays. Journal of immunological methods, 263(1-2), 35-41.

Pestana, M., Correia, P. J., Saavedra, T., Gama, F., Abadía, A., & de Varennes, A. (2012). Development and recovery of iron deficiency by iron resupply to roots or leaves of strawberry plants. Plant physiology and biochemistry, 53, 1-5.

Piepenbring, M. (2015). mycology.

Pnwhandbooks.org. (2020). Strawberry (Fragaria spp.) Verticillium Wilt. Retrieved 2, December, 2020 fromhttps://pnwhandbooks.org/node/3571/print.

Poling, E. B. (2016). An introductory guide to Strawberry plasticulture. Department of horticultural science, NC State.

Prentice, P. Efficacy of silica in increasing fields in Marocco. In Proceedings of the 7th International Conference on Silicon in Agriculture, Bengaluru, India, 24–28 October 2017; p. 107.

Rana, T. S. (2018). Day-Neutral Strawberry Production in Organically Managed Low Tunnels inside a High Tunnel (Doctoral dissertation, North Carolina Agricultural and Technical State University).

Rao, G. B., & Susmitha, P. (2017). Silicon uptake, transportation and accumulation in Rice. J Pharmacogn Phytochem, 6(6), 290-293.

Rice, K. B., Jones, S. K., Morrison, W., & Leskey, T. C. (2017). Spotted wing drosophila prefer low hanging fruit: insights into foraging behavior and management strategies. Journal of insect behavior, 30(6), 645-661.

Robinson Boyer, L., Feng, W., Gulbis, N., Hajdu, K., Harrison, R. J., Jeffries, P., & Xu, X. (2016). The use of arbuscular mycorrhizal fungi to improve strawberry production in coir substrate. Frontiers in plant science, 7, 1237.

Rodrigues, F. A., Resende, R. S., Dallagnol, L. J., & Datnoff, L. E. (2015). Silicon Potentiates Host Defense Mechanisms Against Infection by Plant Pathogens. In *Silicon and Plant Diseases* (pp. 109-138). Springer International Publishing.

Saharan, G. S., Mehta, N. K., & Meena, P. D. (2019). Powdery Mildew Disease of Crucifers: Biology, Ecology and Disease Management. Springer Singapore.

Sampson, C., & Kirk, W. D. (2016). Predatory mites double the economic injury level of Frankliniella occidentalis in strawberry. BioControl, 61(6), 661-669.

Saqib, R. M., Ashraf, M., Shahzad, S. M., & Imtiaz, M. (2011). Silicon nutrition for mitigation of salt toxicity in sunflower (Helianthus annuus L.). International Journal of Agriculture and Applied Sciences (Pakistan).

Savvas, D., Manos, G., Kotsiras, A., & Souvaliotis, S. (2002). Effects of silicon and nutrient-induced salinity on yield, flower quality and nutrient uptake of gerbera grown in a closed hydroponic system. Journal of Applied Botany, 76(5/6), 153-158.

Schaart, J. G., van de Wiel, C. C., Lotz, L. A., & Smulders, M. J. (2016). Opportunities for products of new plant breeding techniques. Trends in Plant Science, 21(5), 438-449.

Schaller, J., Schoelynck, J., Struyf, E., & Meire, P. (2015). Silicon affects nutrient content and ratios of wetland plants. *Silicon*, 1-7.

Schandry, N. (2017). A practical guide to visualization and statistical analysis of R. solanacearum infection data using R. Frontiers in plant science, 8, 623.

Scholey, D. V., Belton, D. J., Burton, E. J., & Perry, C. C. (2018). Bioavailability of a novel form of silicon supplement. Scientific reports, 8(1), 1-8.

Scholler, M., Schmidt, A., Siahaan, S. A. S., Takamatsu, S., & Braun, U. (2016). A taxonomic and phylogenetic study of the Golovinomyces biocellatus complex (Erysiphales, Ascomycota) using asexual state morphology and rDNA sequence data. Mycological Progress, 15(6), 56.

Sharma, S., & Singh, K. (2019). 24 Harvesting. Strawberries: Production, Postharvest Management and Protection, 24.

Shetty, R., Jensen, B., Shetty, N. P., Hansen, M., Hansen, C. W., Starkey, K. R., & Jørgensen, H. J. L. (2012). Silicon induced resistance against powdery mildew of roses caused by Podosphaera pannosa. Plant pathology, 61(1), 120-131.

Shwethkumari, U.; Prakash, N.B.; Gowda, J.; Chandrappa, M. Effect of foliar application of silicic acid on growth, field and quality of soybean [Glycine max. (L)]. In Proceedings of the 7th International Conference on Silicon in Agriculture, Bengaluru, India, 24–28 October 2017; p. 146.

Siedliska, A., Baranowski, P., Zubik, M., Mazurek, W., & Sosnowska, B. (2018). Detection of fungal infections in strawberry fruit by VNIR/SWIR hyperspectral imaging. Postharvest Biology and Technology, 139, 115-126.

Silva, O. N., Lobato, A. K. S., Ávila, F. W., Costa, R. C. L., Neto, C. O., Santos Filho, B. G., ... & Cardoso, M. S. (2012). Silicon-induced increase in chlorophyll is modulated by the leaf water potential in two water-deficient tomato cultivars. Plant, Soil and Environment, 58(11), 481-486.

Simpson, D. (2018). The economic importance of strawberry crops. The Genomes of Rosaceous Berries and Their Wild Relatives, Springer: 1-7.

Singh, R. K., Kumar, S., Prasad, D. N., & Bhardwaj, T. R. (2018). Therapeutic journey of nitrogen mustard as alkylating anticancer agents: Historic to future perspectives. European journal of medicinal chemistry, 151, 401-433.

Singh, S. K., Kalal, P., & Kumar, P. (2019). 16 Mulching. Strawberries: Production, Postharvest Management and Protection, 16.

Smith, E. (2017, February). Strawberry drip irrigation and fertigation. in 62nd new jersey agricultural convention and trade show (p. 95).

Sousa, R. S., Rodrigues, F. A., Schurt, D. A., Souza, N. F. A., and Cruz, M. F. A. (2013). Cytological aspects of the infection process of Pyricularia oryzae on leaves of wheat plants supplied with silicon. Trop. Plant Pathol. 38, 472–477. doi: 10.1590/S1982-56762013000600002.

Spanu, P. D. (2012). "The genomics of obligate (and nonobligate) biotrophs." Annual review of Phytopathology 50: 91-109.

Stankevičienė, A. (2017). The spread of fungi Erysiphaceae Tul & C. Tul. on the woody plants at the city green plantations in Lithuania. In 9th international conference on biodiversity research, Daugavpils, 26-28 April, 2017: book of abstracts. Daugavpils, Latvia: Saule, 2017.

Steeves, T. A., & Sawhney, V. K. (2017). Essentials of Developmental Plant Anatomy. Oxford University Press.

Stewart, P. J. (2016). Strawberry: Part 4.1 Fragaria History and Breeding. In Genetics, Genomics and Breeding of Berries (pp. 140-163). CRC Press.

Strange, R. N. (2013). Plant disease control: towards environmentally acceptable methods. Springer.

Strawberry analysis chart, Optimal ranges. HDC. Retrieved 19, March, 2021 fromhttps://ahdb.org.uk/knowledge-library/strawberry-analysis-chart.

Strawberryplants.org. (2020). Pineberry and pineberries. Retrieved October, 17, 2020 fromhttps://strawberryplants.org/strawberry-plant/.

Taber, H. G., et al. (2002). "Extraction of silicon from plant tissue with dilute HCl and HF and measurement by modified inductive coupled argon plasma procedures." Communications in soil science and plant analysis 33(9-10): 1661-1670.

Takamatsu, S. (2013). "Origin and evolution of the powdery mildews (Ascomycota, Erysiphales)." Mycoscience 54(1): 75-86.

Takamatsu, S., Niinomi, S., Harada, M., & Havrylenko, M. (2010). Molecular phylogenetic analyses reveal a close evolutionary relationship between Podosphaera (Erysiphales: Erysiphaceae) and its rosaceous hosts. Persoonia: Molecular Phylogeny and Evolution of Fungi, 24, 38.

Teixeira, N. C., Valim, J. O. S., & Campos, W. G. (2017). Silicon-mediated resistance against specialist insects in sap-sucking and leaf-chewing guilds in the Si non-accumulator collard. Entomologia Experimentalis et Applicata, 165(1), 94-108.

Tibbitts, S. A. (2018). Effect of silicon on wheat growth and development in drought and salinity stress.

Toljamo, A., Blande, D., Kärenlampi, S., & Kokko, H. (2016). Reprogramming of strawberry (Fragaria vesca) root transcriptome in response to Phytophthora cactorum. PLoS One, 11(8).

Tülek, S., & Canpolat, S. (2016). Effects of some fungicides against powdery mildew (Erysiphe heraclei DC) of carrot. Bitki Koruma Bülteni, 56(4), 349-358.

Turgut, T. and S. Cakmakci (2018). "Probiotic strawberry yogurts: microbiological, chemical and sensory properties." Probiotics and antimicrobial proteins 10(1): 64-70.

Twitchen, C. A. (2018). Impact of crop management during plant propagation on yield potential of Junebearer and Everbearer strawberry (Fragaria x ananassa Duch.) cultivars, University of Reading.

Veloso, J., & van Kan, J. A. (2018). Many Shades of Grey in Botrytis–Host Plant Interactions. Trends in plant science.

Vissers, G. T. M., Konings, H. A. M., & Obers, J. T. A. M. (2018). U.S. Patent Application No. 14/999,473.

Vorotnikova, E., VanSickle, J. J., & Borisova, T. (2012). The Economic value of the precision disease management system for anthracnose and Botrytis fruit rot for the Florida strawberry industry (No. 1372-2016-109013).

Vulavala, V. K., Elbaum, R., Yermiyahu, U., Fogelman, E., Kumar, A., & Ginzberg, I. (2016). Silicon fertilization of potato: expression of putative transporters and tuber skin quality. Planta, 243(1), 217-229.

Wang, M., et al. (2017). "Role of silicon on plant-pathogen interactions." Frontiers in plant science 8: 701.

Washburn, M. (2018). Alternative Control Methods for Verticillium wilt: A Literature Review.

Whitehouse, A. B., Simpson, D. W., Johnson, A. W., McLeary, K. J., Passey, A. J., Nellist, C. F., ... & Harrison, R. J. (2016, August). Progress in strawberry breeding at NIAB-EMR, East Malling, UK. In *VIII International Strawberry Symposium 1156* (pp. 69-74).

Wu, L., Du, G., Bao, R., Li, Z., Gong, Y., & Liu, F. (2019). De novo assembly and discovery of genes involved in the response of Solanum sisymbriifolium to Verticillium dahlia. Physiology and Molecular Biology of Plants, 25(4), 1009-1027.

Wu, Y. (2016). Modifying critical thresholds in Simcast to achieve better late blight management for moderately susceptible potato and tomato crops.

Xiangming, Xu. (2018). Improving integrated disease management in strawberry. AHDB Horticulture, SF 157.

Xie, Z., Song, F., Xu, H., Shao, H., & Song, R. (2014). Effects of silicon on photosynthetic characteristics of maize (Zea mays L.) on alluvial soil. The Scientific World Journal, 2014.

Yang, L., et al. (2017). "Silicon amendment is involved in the induction of plant defense responses to a phloem feeder." Scientific reports 7(1): 4232.

Zahedi, S. M., Moharrami, F., Sarikhani, S., & Padervand, M. (2020). Selenium and silica nanostructure-based recovery of strawberry plants subjected to drought stress. Scientific reports, 10(1), 1-18.

Zhang, T., Wang, N. F., Zhang, Y. Q., Liu, H. Y., & Yu, L. Y. (2016). Diversity and distribution of aquatic fungal communities in the Ny-Ålesund region, Svalbard (High Arctic). Microbial ecology, 71(3), 543-554.

Zheng, Y., & Clark, M. J. (2013). Optimal growing substrate pH for five Sedum species. HortScience, 48(4), 448-452.

Zitouni, H., Hssaini, L., Ouaabou, R., Viuda-Martos, M., Hernández, F., Ercisli, S., ... & Hanine, H. (2020). Exploring antioxidant activity, organic acid, and phenolic composition in strawberry tree fruits (Arbutus unedo L.) growing in morocco. Plants, 9(12), 1677.

APPENDICES

Appendix 1 Silicon fertigation field trial 2016 (Ladybird field) Silicon fertigation field trial 2016 (Ladybird field)









Appendix 3 Silicon fertigation field trial 2018 Amelia field




SAFETY DATA SHEET

Page 1 of 4

3

Revision

Sirius

					Revision date	20-Aug-2009			
1. IDENTIFICATION OF TH	E SUBSTANCE	/ PREPARAT	TION AND TH	E COMPANY	1				
Product name	Sirius								
Company Telephone	Orion Future T Henwood Hous info@orionft.cc Tel:+44 (0) 775	Orion Future Technology Ltd Henwood House, Henwood. Ashford, Kent, TN24 8DH info@orionft.com Tel:+44 (0) 779 284 3398							
Product code	qafs474a								
2 HAZARDS IDENTIFICAT	ION.								
Main hazards	Flammable Ha damage to eye the aquatic en	rmful by inhal es. Harmful to vironment.	ation. Irritating aquatic organi	to respirator isms, may ca	y system. Risk of se use long-term adve	rious rse effects in			
3. COMPOSITION / INFOR	MATION ON IN	GREDIENTS.							
Hazardous ingredients									
Polyether-modified polysiloxane		Conc. 1-10%	CAS 134180-76-0	EINECS	Symbols/Risk phrase Xn;R20 Xn;R21 Xi;R3 N;R51/53	9 5 B Xi;R41			
Ethyl alcohol (Ethanol)		0.5-1%	64-17-5	200-578-6	F; R11				
Tetraethyl silicate		70-80%	78-10-4	201-083-8	R10 Xn; R20 Xi; R36/3	37			
Alkyloxypolyethyleneoxyethanol		10-20%	68131-40-8		Xn; R22 Xi; R41				
Product Description & Uses	A liquid fertilise	er.							
Product Shelf Life	Recommended	d shelf life 3 y	ears from date	of delivery .					
4. FIRST AID MEASURES									
Skin contact	Irritating to skir attention if irrita	n. Wash off im ation or sympt	mediately with	n plenty of so	ap and water. Seek	medical			
Eye contact	Irritating to eye eyelids open. S	s. Rinse imm Seek medical	ediately with p attention. Risk	lenty of wate of serious da	r for 15 minutes hold amage to eyes.	ding the			
Inhalation	Move the expo	sed person to	o fresh air. See	k medical att	tention.				
Ingestion	Ingestion is irri nervous syster glasses of wate	tating to the re n. DO NOT IN er.	espiratory trac IDUCE VOMIT	t and may ca ſING. Rinse r	use damage to the o mouth thoroughly. D	central rink 1 to 2			
General information	lf you feel unw contaminated o	ell, seek medi clothing before	ical advice (sh e reuse.	ow the label	where possible). Wa	ish all			
5. FIRE FIGHTING MEASU	RES								
Extinguishing media	Use extinguish	ing media ap	propriate to the	e surrounding	fire conditions.				
Fire hazards	Burning produc	ces irritating, t	oxic and obno	xious fumes.					
Protective equipment	Wear suitable	respiratory eq	uipment when	necessary.					

Sirius

	Revision 3 Revision date 20.4ug-2009						
6. ACCIDENTAL RELEASE	MEASURES						
Personal precautions	Wear suitable protective equipment.						
Environmental precautions	Do not allow product to enter drains. Advise local authorities if large spills cannot be contained.						
Clean up methods	For large spills: Absorb with inert, absorbent material. Transfer to suitable, labelled container. Clean spillage area thoroughly with plenty of water. For small spills: Flush down the drain with plenty of water.						
7. HANDLING AND STORA	GE						
Handling	Wear suitable protective equipment. Avoid contact with eyes and skin.						
Storage	Keep out of the reach of children. Store in correctly labelled containers. Store in original container. Keep containers tightly closed.						
8. EXPOSURE CONTROLS	/ PERSONAL PROTECTION						
Exposure limits							
Ethyl alcohol (Ethanol)	WEL 8-hr limit ppm: 1000WEL 8-hr limit mg/m3: 1920WEL 15 min limit ppm: -WEL 15 min limit mg/m3: -						
Respiratory protection	Wear suitable respiratory equipment when necessary.						
Hand protection	Chemical resistant gloves (plastic)						
Eye protection	Avoid contact with eyes. Provide eye wash station. Approved safety goggles.						
Protective equipment	Avoid contact with eyes and skin. Wash all contaminated clothing before reuse. Adopt best Manual Handling considerations when handling, carrying and dispensing.						
9. PHYSICAL AND CHEMIC	AL PROPERTIES						
Description	Liquid.						
Colour	Clear.						
Odour	Slight.						
рН	n/a						
Flash point	35°C						
Flammability limits	Flammable						
Relative density	0.960 - 0.980						
Viscosity	Kinematic Viscosity in 10-6 m²/s at 40°C (ISO 3219) - < 50 cps						
10. STABILITY AND REAC	ΤΙVΙΤΥ						
Stability	Stable under normal conditions.						
Materials to avoid	Strong oxidising agents. Strong acids.						
Hazardous decomposition products	Carbon oxides.						
11. TOXICOLOGICAL INFO	RMATION						
Acute toxicity	Ingestion is irritating to the respiratory tract and may cause damage to the central nervous system.						
Corrosivity	May cause irritation to skin. Irritating to skin.						
Repeated or prolonged exposure	Prolonged or repeated exposure may cause irritation to skin and mucous membranes.						



Appendix 7 Maps of Maltmas farmMap A - Rural Land Register map of Maltmas Farm

Appendix 8 Map B - Distribution of strawberry field in Maltmas Farm

The Amelia field and Ladybird field were areas covered in this thesis between 2016 and 2018



Appendix 9 List of fungicides for the silicon fertigation field trial 2016

5	OFTWARE LIMITED	Ano	ha I	Ċ	¥.	mind i Pr	editer 7	ma
	19		-					
& H. Dur	ncalfe, MALTMAS FARM, FR	IDAYB	RIDGE, WISBEC	CH, CAMBS.	, PE14 OHS, Te	el:(01945 860287)		
214, Stra	awberries, Sonata (1.00	ha)		1.1.	27/06/16	PROLECTUS	1.00 1.000 kg	N/C
01-Oct-1	15-30-Sep-16) Ca	b porte	rid App	dication	P. 14/07/16	HALLMARK WITH ZEON	1.00 0.100 ml	N/C
lanned	Product	Area	Rate //	Applied		TECHNOLOGY		
2/04/16	TRIGGER 3	1.00	3.000 kg	N/C	20/00/110	SW/	1.00 100.000 unit	N/C
9/04/16	SWITCH	1.00	1.000 kg	N/C	30/08/10	CORBEL	1.00 0.750 lt	N/C
	SYSTHANE 20EW	1.00	0.330 lt	N/C		FURTRESS	1.00 0.250 lt	N/C
	SW7	1.00	100.000 unit	N/C		SW/	1.00 100.000 Unit	N/C
	MANZI	1.00	1.000 lt	N/C		TECHNOLOGY	1.00 0.100 mi	N/C
C 105 11 C	HALLMARK WITH ZEON TECHNOLOGY	1.00	0.100 ml	N/C		MASAI	1.00 0.750 kg	N/C
.6/05/16	SIGNUM	1.00	1.250 kg	N/C				
	SVV/	1.00	100.000 unit	N/C				
		1.00	1.000 lt	N/C	_	- milder	Action	
	TECHNOLOGY	1.00	0.100 mi	N/C		2		
4/05/16	SWITCH	1.00	1.000 kg	N/C				
1,05,10	SW/7	1.00	100 000 upit	N/C		IIIh	1'stime	
	SYSTHANE 20EW	1.00	0 330 lt	N/C	merch	when Syster P/	011 Car, 000	~
		1.00	2 000 unit	N/C			k	
	MANZI	1.00	1 000 lt	N/C		h	IE 1	p EL
	SILIGGO	1.00	10.000 kg	N/C	N	13 Cum		/
1/06/16	HALLMARK WITH ZEON	1.00	0.100 ml	N/C				
1,00,10	TECHNOLOGY	1.00	1.000 lt	N/C	,	7 June	300 mls	Systhe
12.00	STEM	1.00	1.000 unit	N/C				2
	SYSTHANE 20EW	1.00	0.330 lt	N/C				, 11
	HORTIPHYTE	1.00	2.000 unit	N/C	0.	1 Juli	16 Norm	w// ho
	SW7	1.00	100.000 unit	N/C	LI			/
3/06/16	SIGNUM	1.00	1.250 kg	N/C			- 1.	c.12
	SW7	1.00	100.000 unit	N/C	16	July	326 mis .	J 51 . 1
	SYSTHANE 20EW	1.00	0.330 lt	N/C				1
	HORTIPHYTE	1.00	2.000 unit	N/C		7 5.1.	12 Corbe	1.
4/06/16	MANZI	1.00	1.000 lt	N/C	1	g - any		
	HORTIPHYTE	1.00	2.000 unit	N/C		1 4	1 hr	mel
	ROVRAL WG	1.00	0.500 kg	N/C	5	1 Ang	12 Wimt	0
	MANZI	1.00	1.000 lt	N/C		/	, 1	/
	SIGNUM	1.00	1.250 kg	N/C		1.0 1	2 Curbe	1
7/06/16	SW7	1.00	100.000 unit	N/C	30	108	1.	15 la
	MANZI	1.00	1.000 lt	N/C		· +	- 250 - 250	
	ROVRAL WG	1.00	1.000 kg	N/C				
2/06/16	SW7	1.00	100.000 unit	N/C				
	MANZI	1.00	1.000 lt	N/C				
	TELDOR	1.00	1.000 kg	N/C				
4/06/16	SW7	1.00	100.000 unit	N/C				
	MANZI	1.00	1.000 lt	N/C				
	SCALA	1.00	1.000 lt	N/C				- 95
7/06/16	MANZI	1.00	1.000 lt	N/C				
	SW7	1.00	100.000 unit	N/C				

Appendix 10 fungicides sprays used for the silicon fertigation field trial 2017 cont'd

MUDDY BOOTS SOFTWARE LIMITED

Applications Summary

Advisor : Neil Holmes

Dua duat			
Product	Area	Rate	Applied
KERB FLO	0.60	2.000 lt	N/C
DEVRINOL	0.60	5.000 lt	N/C
STOMP 400 SC	0.60	3.300 lt	N/C
FENOMENAL	1.20	2.000 kg	N/C
VENZAR FLOWABLE	1.20	0.400 lt	N/C
AMISTAR	1.20	1.000 lt	N/C
ESCAR-GO 3	1.20	4.000 kg	N/C
REGLONE	0.60	2.000 lt	N/C
AMISTAR	1.20	1.000 lt	N/C
HORTIPHYTE	1.20	2.000 unit	N/C
LUNA SENSATION	1.20	0.800 lt	N/C
CALYPSO	1.20	0.250 lt	N/C
MANZI	1.20	2.000 lt	N/C
MANZI	1.20	2.000 lt	N/C
SIGNUM	1.20	1.250 kg	N/C
AMISTAR	1.20	1.000 lt	N/C
CALYPSO	1.20	0.250 lt	N/C
LUNA SENSATION	1.20	0.800 lt	N/C
SCALA	1.20	1.000 lt	N/C
SYSTHANE 20EW	1.20	0.330 lt	N/C
AMISTAR	1.20	1.000 lt	N/C
SYSTHANE 20EW	1.20	0.330 lt	N/C
TELDOR	1.20	1.000 kg	N/C
LUNA SENSATION	1.20	0.800 lt	N/C
NIMROD	1.20	1.000 lt	N/C
TELDOR	1.20	1.000 kg	N/C
SYSTHANE 20EW	1.20	0.330 lt	N/C
Hallmark with Zeon Technology	1.20	0.100 ml	N/C
TELDOR	1.20	1.000 kg	N/C
LUNA SENSATION	1.20	0.800 lt	N/C
TRACER	1.20	0.150 lt	N/C
TELDOR	1.20	1.000 kg	N/C
SYSTHANE 20EW	1.20	0.330 lt	N/C
TRACER	1.20	0.150 lt	N/C
SIGNUM	1.20	1.250 kg	N/C
TRACER	1.20	0.150 lt	N/C
TELDOR	1.20	1.000 kg	N/C
	P16 - 30/09/2017) Product KERB FLO DEVRINOL STOMP 400 SC FENOMENAL VENZAR FLOWABLE AMISTAR ESCAR-GO 3 REGLONE AMISTAR HORTIPHYTE LUNA SENSATION CALYPSO MANZI MANZI SIGNUM AMISTAR CALYPSO LUNA SENSATION SCALA SYSTHANE 20EW TELDOR LUNA SENSATION NIMROD TELDOR LUNA SENSATION TECHNOLOGY TELDOR LUNA SENSATION TRACER SIGNUM ANENTION TRACER SIGNUM	D16 - 30/09/2017) Product Area KERB FLO 0.60 DEVRINOL 0.60 STOMP 400 SC 0.60 FENOMENAL 1.20 VENZAR FLOWABLE 1.20 AMISTAR 1.20 ESCAR-GO 3 1.20 HORTIPHYTE 1.20 HORTIPHYTE 1.20 MINZI 1.20 MANZI 1.20 MANZI 1.20 MANZI 1.20 MANZI 1.20 MANZI 1.20 SIGNUM 1.20 SYSTHANE 20EW 1.20 SYSTHANE 20EW 1.20 NIMROD 1.20 SYSTHANE 20EW 1.20 SYSTHANE 20EW 1.20 NIMROD 1.20 SYSTHANE 20EW 1.20 ULUNA SENSATION 1.20 SYSTHANE 20EW 1.20 TELDOR 1.20 ULUNA SENSATION 1.20 SYSTHANE 20EW 1.20	Product Area Rate KERB FLO 0.60 2.000 lt DEVRINOL 0.60 5.000 lt STOMP 400 SC 0.60 3.300 lt FENOMENAL 1.20 2.000 kg VENZAR FLOWABLE 1.20 0.400 lt AMISTAR 1.20 1.000 lt ESCAR-GO 3 1.20 4.000 kg REGLONE 0.60 2.000 lt AMISTAR 1.20 1.000 lt HORTIPHYTE 1.20 2.000 unit LUNA SENSATION 1.20 0.250 lt MANZI 1.20 2.000 lt MANZI 1.20 2.000 lt SIGNUM 1.20 2.000 lt MANZI 1.20 1.000 lt SALYSO 1.20 0.250 lt LUNA SENSATION 1.20 0.250 lt LUNA SENSATION 1.20 0.250 lt SYSTHANE 20EW 1.20 0.330 lt SYSTHANE 20EW 1.20 0.330 lt TELDOR 1.20 <td< td=""></td<>

©Muddy Boots Software Ltd 2017



Appendix 11 fungicides sprays used for the silicon fertigation field trial 2018 cont'd



Appendix 12 fungicides sprays used for the silicon fertigation field trial 2018 cont'd

Appendix 13 Calculation of silicon nutrients (Sirius) application rate in the 2016 to 2018 silicon fertigation field trials at Maltmas Farm

The silicon fertigation field trials were set under polythene tunnels of 180 metres long. The trials consisted of 5 strawberry beds with 6 strawberry crops growing in coir bags (1 metre). Each field tunnel contained approximately 5,000 - 5300 strawberry plants. Water and silicon nutrient were delivered through 4 irrigation drippers per coir bag. Rate of irrigation was 2.2 L water per hour and each irrigation time was 5 minutes.

The suggested rate for silicon nutrient used in agriculture is 50 - 100ml in 100 - 600 L of water (Jin, 2015; Liu; 2016). The grower used 100ml of silicon nutrient 'Sirius' in 600 L per hectare. Which calculated as

100 mL in 600 L = 0.017% Sirius (v/v) In 5 minutes, each irrigation tunnels received;

5 minutes/60 x 2.2 L / hour x 4 drippers x 180 m x 15 beds = 1584 Litre water

Therefore, the amount of silicon nutrient (0.017% Sirius v/v) for one irrigation tunnel

0.00017 x 1584 L = 0.269 L = approximately 270 ml

Each strawberry plant received approximately 270ml

270ml / 5000 - 5300 plants x 2 tunnels = 0.025 - 0.027 ml Sirius per plant.

As Sirius consists of four different compounds, the actual amount of soluble silicon that was taken up by the plants may be lower than amounts calculated above.

Appendix 14 Chapter 3 - Results workings for disease levels of strawberry powdery mildew in the silicon fertigation field experiment 2016 (Malling Centenary)

Normality test for disease levels in the 2016 field experiment

(a) Normality Test

Null and alternate Hypothesis:

HO: Difference between untreated and silicon + fungicides follow normal distribution **HA**: Difference between untreated and silicon + fungicides does not follow normal distribution

Level of significance = 0.05

Tests of Normality

	Kolm	ogorov-Smir	nov ^a	Shapiro-Wilk			
Statistic df		df	df Sig.		df	Sig.	
Difference	.170	12	.200	.915	12	.246	

*. This is a lower bound of the true significance.

If sig (P) value is less than critical value 0.05 we reject Ho. In the analysis sig value 0.246 greater than critical value 0.05 we accept Ho and conclude that the difference between untreated and silicon and fungicides follows normal distribution. The parametric test (Paired samples test) was used to test for significance difference.

(b) Null and Alternate Hypothesis

Ho: There is no statistical significant difference between untreated and silicon + fungicides

Ha: There is statistical significant difference between untreated and silicon + fungicides-only

Paired Samples Test										
			Std. Error	95% Confidence Differ	e Interval of the rence					
	Mean	Std. Deviation	Mean	Lower	Upper	t	df	Sig. (2-tailed)		
Paired 1 – Untreated – Silicon_ Fungicides_	224.25000	254.42810	73.44707	62.59410	385.90590	3.053	11	.011		

If sig (*P*) value is less than critical value 0.05 we reject Ho. In the analysis sig value.011 less than critical value 0.05 (0.011 < 0.05) we reject Ho and conclude that there is statistical significant difference between untreated and silicon + fungicides treatment.

(c) Normality Test

Null and alternate Hypothesis:

HO: Difference between untreated and silicon + no fungicides follow normal distribution

HA: Difference between untreated and silicon + no fungicides does not follow normal distribution

Level of significance = 0.05

Tests of Normality

	Kolm	ogorov-Smir	nov ^a	Shapiro-Wilk				
	Statistic df Sig.				Statistic df Sig.			
Difference	.216	12	.126	.911	12	.222		

a. Lilliefors Significance Correction

If sig value is less than critical value 0.05 we reject Ho. In the analysis sig value 0.222 is greater than critical value 0.05 we accept Ho and conclude that the difference between untreated and silicon + no fungicides follow normal distribution. The parametric test (Paired samples test) was used to test for significance difference.

(d) Null and Alternate Hypothesis

Ho: There is no statistical significant difference between untreated and silicon + no fungicides

Ha: There is statistical significant difference between untreated and silicon + no fungicides.



If sig sig (*P*) value is less than critical value 0.05 we reject Ho. In the analysis sig value.106 is greater than critical value 0.05 (0.106 > 0.05) we accept Ho and conclude that there is no statistical significance difference between untreated and silicon + no Fungicides.

(e) Normality Test

Null and alternate Hypothesis:

HO: Difference between untreated and silicon twice + fungicides follow normal distribution

HA: Difference between untreated and silicon twice + fungicides does not follow normal distribution

Level of significance = 0.05

Tests of Normality

	Kolmogorov-Smirnov ^a				Shapiro-Wilk	
	Statistic df Sig.			Statistic	df	Sig.
Difference	.169	12	.200	.882	12	.094

*. This is a lower bound of the true significance.

a. Lilliefors Significance Correction

If sig (P) value is less than critical value alpha 0.05 we reject Ho. In the analysis P value.094 greater than critical value 0.05 we accept Ho and conclude that data does follow normal distribution. The parametric test (Paired samples test) is used to measure the difference among the treatment.

(f) Null and Alternate Hypothesis

Ho: There is no statistical significant difference between untreated and silicon twice + fungicides

Ha: There is statistical significant difference between untreated and silicon twice + fungicides.



If sig (*P*) value is less than critical value 0.05 we reject Ho. In the analysis sig value 0.007 less than critical value 0.05 (0.007 < 0.05) we reject Ho and conclude that there is statistical significant difference between untreated and silicon twice + fungicides.

(g) Null and alternate Hypothesis:

HO: Difference between untreated and silicon twice + no fungicides follow normal distribution

HA: Difference between untreated and silicon twice + no fungicides does not follow normal distribution

Level of significance = 0.05

Tests of Normality

	Kolm	ogorov-Smir	nov ^a	:	Shapiro-Wilk		
	Statistic df Sig.			Statistic df Sig.			
Difference	.182	12	.200	.869	12	.063	

*. This is a lower bound of the true significance.

a. Lilliefors Significance Correction

If sig

(P) value is less than critical value alpha 0.05 we reject Ho. In the analysis P value.063 greater than critical value 0.05 we accept Ho and conclude that data does follow normal distribution. The parametric test (Paired samples test) was used to measure the difference among the treatment.

(h) Null and Alternate Hypothesis

Ho: There is no statistical significant difference between Untreated and silicon twice + no Fungicides

Ha: There is statistical significant difference between untreated and silicon twice + no fungicides



If sig (*P*) value is less than critical value 0.05 we reject Ho. In the analysis sig value 0.005 is less than critical value 0.05 (0.005 < 0.05) we reject Ho and conclude that there is statistically significance difference between untreated and silicon twice + no fungicides.

Appendix 15 Raw data for silicon extraction from strawberry leaves in the silicon fertigation field trial 2016 (chapter 3)

Terretoria	Committe data	
Treatments	Sample date	Amount of silicon
Untreated	19/4/2016	0.42µg/mg
No silicon + fungicides only	19/4/2016	0.59µg/mg
Silicon + fungicides	19/4/2016	0.15µg/mg
Silicon + no fungicides	19/4/2016	0.15µg/mg
Silicon twice + fungicides	19/4/2016	0.18µg/mg
Silicon twice + no fungicides	19/4/2016	0.39µg/mg
Treatments	Sample date	Amount of silicon
Untreated	03/05/2016	0.42µg/mg
No silicon + fungicides only	03/05/2016	0.38µg/mg
Silicon + fungicides	03/05/2016	0.24µg/mg
Silicon + no fungicides	03/05/2016	0.33µg/mg
Silicon twice + fungicides	03/05/2016	0.41µg/mg
Silicon twice + no fungicides	03/05/2016	0.44 μg/mg
Treatments	Sample date	Amount of silicon
Untreated	07/06/2016	0.36µg/mg
No silicon + fungicides only	07/06/2016	0.32µg/mg
Silicon + fungicides	07/06/2016	0.51µg/mg
Silicon + no fungicides	07/06/2016	0.64µg/mg
Silicon twice + fungicides	07/06/2016	0.64µg/mg
Silicon twice + no fungicides	07/06/2016	0.68µg/mg
Treatments	Sample date	Amount of silicon
Untreated	05/07/2016	0.39µg/mg
No silicon + fungicides only	05/07/2016	0.89µg/mg
Silicon + fungicides	05/07/2016	0.49µg/mg

Appendix 16 Raw data for silicon extraction from strawberry leaves in the silicon fertigation field trial 2016 cont'd (chapter 3)

Silicon + no fungicides	05/07/2016	0.79µg/mg
Silicon twice + fungicides	05/07/2016	0.55µg/mg
Silicon twice + no fungicides	05/07/2016	0.42µg/mg
Treatments	Sample date	Amount of silicon
Untreated	02/08/2016	0.64µg/mg
No silicon + fungicides only	02/08/2016	0.30µg/mg
Silicon + fungicides	02/08/2016	0.89µg/mg
Silicon + no fungicides	02/08/2016	0.64µg/mg
Silicon twice + fungicides	02/08/2016	0.55µg/mg
Silicon twice + no fungicides	02/08/2016	0.59µg/mg
Treatments	Sample date	Amount of silicon
Untreated	06/09/2016	1.06µg/mg
No silicon + fungicides only	06/09/2016	1.32µg/mg
Silicon + fungicides	06/09/2016	1.79µg/mg
Silicon + no fungicides	06/09/2016	0.48µg/mg
Silicon twice + fungicides	06/09/2016	0.90µg/mg
Silicon twice + no fungicides	06/09/2016	2.92µg/mg

Appendix	17 chapter	3 results	workings	for si	ilicon	extraction	between
treatment	in the 2016	field exp	eriment				

			Silicon
			once +no
		Untreated	fungicides
Mean		0.6125	0.6375
Variance		0.104758333	0.016025
Observations		4	4
		-	
Pearson Correlation		0.854026446	
Hypothesized	Mean		
Difference		0	
df		3	
		-	
t Stat		0.114477137	
P(T<=t) one-tail		0.458045687	
t Critical one-tail		2.353363435	
P(T<=t) two-tail		0.916091374	
t Critical two-tail		3.182446305	

(a) t-Test: Paired Two Sample for Means (Untreated and Silicon once + no fungicides)

Silicon extractions results are used as evidence that the silicon delivered via the fertigation system is taken up by the plants. Results from the table has shown that there was no statistical significant difference (P>0.05) between levels of silicon in untreated strawberry plants and strawberry plants treated with silicon once + no fungicides.

(b) t-Test: Paired Two Sample for Means

		untreated	Once silicon + fungicides
Mean		0.6125	0.442
Variance		0.03165	0.01967
Observations		4	4
Pearson Correlation		0.062925063	
Hypothesized	Mean		
Difference		0	
df		3	
		-	
t Stat		0.983507686	
P(T<=t) one-tail		0.190520022	
t Critical one-tail		2.131846786	
P(T<=t) two-tail		0.381040044	
t Critical two-tail		2.776445105	

Results has shown that there was no statistical significant difference (P>0.05) between levels of silicon in untreated strawberry plant and once silicon + fungicides treatment.

	Untreated	Silicon twice + fungicides
Mean	0.6125	0.66
Variance	0.104758333	0.0274
Observations	4	4
Pearson Correlation	0.833085836	
Hypothesized Mean Difference	0	
df	3	
	-	
t Stat	0.458708876	
P(T<=t) one-tail	0.338817331	
t Critical one-tail	2.353363435	

(c) t-Test: Paired Two Sample for Means (Untreated and Silicon twice + fungicides)

$P(T \le t)$ two-tail	0.677634663
t Critical two-tail	3.182446305

Results from the table has shown that there was no statistical significant difference between (P>0.05) levels of silicon in untreated strawberry plants and strawberry plants treated with silicon twice + fungicides.

Untreated silicon twice + no fungicides Mean 0.6125 1.375 Variance 0.021946667 0.17391 Observations 4 4 Pearson Correlation 0.705408461 Hypothesized Mean Difference 0 df 3 t Stat -5.436016697 P(T<=t) one-tail 0.001428983 t Critical one-tail 2.015048373 P(T<=t) two-tail 0.002857965 t Critical two-tail 2.570581836

(d) t-Test: Paired Two Sample for Means

Results has shown that there was a statistical significant difference (P < 0.05) between levels of silicon in untreated strawberry plants and strawberry plants treated with silicon twice + no fungicides.

Appendix 18 Correlation analysis between silicon extracted and levels of disease in the 2016 field experiment

Before conducting correlation analysis, first of check whether the data collected is normal or not. This was done using the Shapiro Wilk test in table 3.4.

Null and alternate Hypothesis:

Ho: Data follows normal distribution.

Ha: Data does not follow normal distribution.

Level of significance = 0.05

Tests of Normality						
	Kolm	ogorov-Smir	nov ^a	5	Shapiro-Wilk	
	Statistic df Sig.			Statistic	df	Sig.
Silicon_Extraction	.246	36	.000	.689	36	.000
Disease_Level	.360	36	.000	.435	36	.000
a. Lilliefors Significance Correction						

If the value of *P* is less than critical value alpha 0.05, reject Ho and conclude that data does not follow from normal distribution. In the analysis in table 3.4, the sig value of silicon extraction and disease level using the Shapiro wilk test is 0.000 less than critical value 0.05. Reject Ho and conclude that the data does not follow normal distribution. The sig value of Kolmogorov-Smirnov test is less than 0.05, reject Ho and conclude that data does not follow normal distribution. Since the data does not follow normal distribution, a nonparametric test was used to analyse this data. The Spearman's Rank correlation analysis was used to find the correlation (Relationship) between silicon extracted and disease levels in the silicon fertigation field experiment 2016. See table 3.5 for the correlation analysis.

Nonparametric Correlations

		Correlations		
			Silicon_Extrac tion	Disease_Lev el
Spearman's rho	Silicon_Extraction	Correlation Coefficient	1.000	.323
		Sig. (2-tailed)		.055
		Ν	36	36
	Disease_Level	Correlation Coefficient	.323	1.000
		Sig. (2-tailed)	.055	
		N	36	36

The Spearman's Rank correlation analysis measured the strength of relationship between silicon extraction and level of infection in the silicon fertigation field experiment 2016. If the correlation is greater than 0.50 the relationship is strong. Although there was flooding in the silicon field tunnel in 2016 (These results presented here includes the flooded beds) the analysis shown in table 3.5, correlation value 0.323, suggests that there is a positively weak to moderate relationship between silicon extracted and levels of disease in the 2016 silicon fertigation field experiment.

However,

If the *P* value is less than critical value, reject Ho. In the analysis shown in table 3.5, the *P* value.055 is greater than critical value 0.05, accept Ho and conclude that the relationship between silicon extraction and disease level is moderate, but is not statistically significant (P>0.05).

Appendix 19 Raw data for silicon extraction from strawberry leaves in the silicon fertigation field trial 2017 (Chapter 3)

Treatments	Sample date	Amount of silicon
Untreated	9/4/2017	0.17µg/mg
No silicon + fungicides only	9/4/2017	0.25µg/mg
Silicon twice + fungicides	9/4/2017	0.56µg/mg
Silicon twice + no fungicides	9/4/2017	0.66µg/mg
Treatments	Sample date	Amount of silicon
Untreated	23/05/2017	0.20µg/mg
No silicon + fungicides only	23/05/2017	0.32µg/mg
Silicon twice + fungicides	23/05/2017	0.62µg/mg
Silicon twice + no fungicides	23/05/2017	0.55µg/mg
Treatments	Sample date	Amount of silicon
Untreated	06/06/2017	0.15µg/mg
No silicon + fungicides only	06/06/2017	0.19µg/mg
Silicon twice + fungicides	06/06/2017	0.72µg/mg
Silicon twice + no fungicides	06/06/2017	0.76µg/mg
Treatments	Sample date	Amount of silicon
Untreated	04/07/2017	0.22µg/mg
No silicon + fungicides only	04/07/2017	0.28µg/mg
No silicon + fungicides only Silicon twice + fungicides	04/07/2017 04/07/2017	0.28µg/mg 0.60µg/mg
No silicon + fungicides only Silicon twice + fungicides Silicon twice + no fungicides	04/07/2017 04/07/2017 04/07/2017	0.28µg/mg 0.60µg/mg 0.75µg/mg
No silicon + fungicides only Silicon twice + fungicides Silicon twice + no fungicides Treatments	04/07/2017 04/07/2017 04/07/2017 Sample date	0.28µg/mg 0.60µg/mg 0.75µg/mg Amount of silicon
No silicon + fungicides only Silicon twice + fungicides Silicon twice + no fungicides Treatments Untreated	04/07/2017 04/07/2017 04/07/2017 Sample date 01/08/2017	0.28µg/mg 0.60µg/mg 0.75μg/mg Amount of silicon 0.17µg/mg
No silicon + fungicides only Silicon twice + fungicides Silicon twice + no fungicides Treatments Untreated No silicon + fungicides only	04/07/2017 04/07/2017 04/07/2017 Sample date 01/08/2017 01/08/2017	0.28µg/mg 0.60µg/mg 0.75μg/mg Amount of silicon 0.17µg/mg 0.24µg/mg
No silicon + fungicides only Silicon twice + fungicides Silicon twice + no fungicides Treatments Untreated No silicon + fungicides only Silicon twice + fungicides	04/07/2017 04/07/2017 04/07/2017 Sample date 01/08/2017 01/08/2017 01/08/2017	0.28µg/mg 0.60µg/mg 0.75µg/mg Amount of silicon 0.17µg/mg 0.24µg/mg 0.84µg/mg
No silicon + fungicides only Silicon twice + fungicides Silicon twice + no fungicides Treatments Untreated No silicon + fungicides only Silicon twice + fungicides Silicon twice + no fungicides	04/07/2017 04/07/2017 04/07/2017 Sample date 01/08/2017 01/08/2017 01/08/2017 01/08/2017	0.28µg/mg 0.60µg/mg 0.75µg/mg Amount of silicon 0.17µg/mg 0.24µg/mg 0.84µg/mg 0.79µg/mg
No silicon + fungicides only Silicon twice + fungicides Silicon twice + no fungicides Untreated No silicon + fungicides only Silicon twice + fungicides Silicon twice + no fungicides Treatments	04/07/2017 04/07/2017 04/07/2017 Sample date 01/08/2017 01/08/2017 01/08/2017 01/08/2017 Sample date	0.28µg/mg 0.60µg/mg 0.75µg/mg Amount of silicon 0.17µg/mg 0.24µg/mg 0.84µg/mg 0.79µg/mg
No silicon + fungicides only Silicon twice + fungicides Silicon twice + no fungicides Untreated No silicon + fungicides only Silicon twice + fungicides Silicon twice + no fungicides Treatments Untreated No silicon + fungicides only Silicon twice + no fungicides Untreated Untreated	04/07/2017 04/07/2017 04/07/2017 Sample date 01/08/2017 01/08/2017 01/08/2017 01/08/2017 Sample date 05/09/2017	0.28µg/mg 0.60µg/mg 0.75µg/mg Amount of silicon 0.17µg/mg 0.24µg/mg 0.84µg/mg 0.79µg/mg Amount of silicon 0.26µg/mg
No silicon + fungicides only Silicon twice + fungicides Silicon twice + no fungicides Treatments Untreated No silicon + fungicides only Silicon twice + fungicides Silicon twice + no fungicides Silicon twice + no fungicides Untreated No silicon + fungicides Untreated No silicon + fungicides only	04/07/2017 04/07/2017 04/07/2017 Sample date 01/08/2017 01/08/2017 01/08/2017 01/08/2017 Sample date 05/09/2017	0.28µg/mg 0.60µg/mg 0.75µg/mg Amount of silicon 0.17µg/mg 0.24µg/mg 0.84µg/mg 0.79µg/mg Amount of silicon 0.26µg/mg 0.42µg/mg
No silicon + fungicides only Silicon twice + fungicides Silicon twice + no fungicides Treatments Untreated No silicon + fungicides only Silicon twice + fungicides Silicon twice + no fungicides Treatments Untreated No silicon + fungicides Silicon twice + no fungicides Treatments Untreated No silicon + fungicides only Silicon twice + fungicides only Silicon twice + fungicides	04/07/2017 04/07/2017 04/07/2017 Sample date 01/08/2017 01/08/2017 01/08/2017 01/08/2017 01/08/2017 Sample date 05/09/2017 05/09/2017	0.28µg/mg 0.60µg/mg 0.75µg/mg Amount of silicon 0.17µg/mg 0.24µg/mg 0.84µg/mg 0.79µg/mg Amount of silicon 0.26µg/mg 0.42µg/mg 0.51µg/mg

Appendix 20 Chapter 3 Results workings for silicon extraction in the silicon fertigation field experiment 2017

t-Test: Paired Two Sample for Means (Untreated and fungicides-only)

	Untreated	fungicides-only
Mean	0.195	0.128333333
Variance	0.00587	0.005336667
Observations	4	4
Pearson Correlation	0.559229726	
Hypothesized Mean Difference	0	
df	3	
t Stat	0.23218173	
P(T<=t) one-tail	0.412800395	
t Critical one-tail	2.015048373	
P(T<=t) two-tail	0.825600791	
t Critical two-tail	2.570581836	

(a) t-Test: Paired Two Sample for Means

Results from appendix 20 a shown that there were no differences between levels of silicon in both treatment

	Untreated	Silicon twice + fungicides
Mean	0.195	1.055
Variance	0.0023	0.038966667
Observations	4	4
Pearson Correlation	0.605615032	
Hypothesized Mean Difference	0	
df	3	
	_	
t Stat	9.963692477	
P(T<=t) one-tail	0.001075601	
t Critical one-tail	2.353363435	
P(T<=t) two-tail	0.002151202	
t Critical two-tail	3.182446305	

(b) t-Test: Paired Two Sample for Means (Untreated and Silicon twice + fungicides)

Results from the table appendix 20 b has shown that there was a statistical significant difference (P < 0.05) between levels of silicon in untreated strawberry leaves and strawberry leaves treated with silicon twice + fungicides.

	Untreated	Silicon twice + no fungicides
Mean	0.195	0.8125
Variance	0.0023	0.072691667
Observations	4	4
Pearson Correlation	0.69991024	
Hypothesized Mean Difference	0	
df	3	
t Stat	-5.1777685	
P(T<=t) one-tail	0.00699142	
t Critical one-tail	2.35336343	
P(T<=t) two-tail	0.01398284	
t Critical two-tail	3.18244631	

(c) t-Test: Paired Two Sample for Means (Untreated and Silicon twice + no fungicides)

Results has from appendix 20 c has shown that there was a statistical significant difference (P < 0.05) between levels of silicon in untreated strawberry leaves and strawberry leaves treated with silicon twice + no fungicides. The differences in silicon levels at the end of the season in the 2017 (Figure 3.5) compared with the 2016 silicon fertigation field experiment (Figure 3.4) may have contributed to the fact that the same age leaves were sampled in this revised silicon fertigation field experiment for 2017.

Appendix 21 Raw data for silicon extraction from strawberry leaves in silicon fertigation field trial 2018 (chapter 3)

Treatments	Sample date	Amount of silicon
Untreated	3/4/2018	0.35µg/mg
No silicon + fungicides only	3/4/2018	0.43µg/mg
Silicon twice + fungicides	3/4/2018	0.59µg/mg
Silicon twice + no fungicides	3/4/2018	0.52µg/mg
Treatments	Sample date	Amount of silicon
Untreated	08/05/2018	0.51µg/mg
No silicon + fungicides only	08/05/2018	0.45µg/mg
Silicon twice + fungicides	08/05/2018	0.58µg/mg
Silicon twice + no fungicides	08/05/2018	0.45µg/mg
Treatments	Sample date	Amount of silicon
Untreated	19/06/2018	0.40µg/mg
No silicon + fungicides only	19/06/2018	0.57µg/mg
Silicon twice + fungicides	19/06/2018	0.58µg/mg
Silicon twice + no fungicides	19/06/2018	0.59µg/mg
Treatments	Sample date	Amount of silicon
Untreated	03/07/2018	0.47µg/mg
No silicon + fungicides only	03/07/2018	0.47µg/mg
Silicon twice + fungicides	03/07/2018	0.70µg/mg
Silicon twice + no fungicides	03/07/2018	0.67µg/mg

Appendix 22 Chapter 3 Results workings for silicon extraction in the silicon fertigation field experiment 2018

	Untreated	Fungicides only
Mean	0.4325	0.642
Variance	0.066357143	0.357390476
Observations	4	4
Pearson Correlation	0.694210849	
Hypothesized Mean Difference	0	
df	4	
t Stat	-0.17323258	
P(T<=t) one-tail	0.434082502	
t Critical one-tail	1.943180281	
P(T<=t) two-tail	0.868165005	
t Critical two-tail	2.446911851	

(a) t-Test: Paired Two Sample for Means

No statistical differences (P>0.05) were found between untreated and fungicides-only treatment

(b) t-Test: Paired Two Sample for Means (Untreated and Silicon	twice +
fungicides)	

	Untreated	Silicon twice + fungicides	
Mean	0.4325	1.035	
Variance	0.00509167	0.031033333	
Observations	4	4	
Pearson Correlation	-0.3168856		
Hypothesized Mean Difference	0		
df	3		
t Stat	-5.7386375		
P(T<=t) one-tail	0.0052537		

t Critical one-tail	2.35336343
P(T<=t) two-tail	0.01050739
t Critical two-tail	3.18244631

Results has shown that there was a statistical significant difference (P < 0.05) between levels of silicon in untreated strawberry leaves and strawberry leaves treated with silicon twice + fungicides. This shows that there was more silicon in silicon-treated plants compared to untreated plants.

		Silicon
	Untreated	twice + no fungicides
Mean	0.4325	0.9125
Variance	0.005091667	0.06869167
Observations	4	4
Pearson Correlation	-0.577931972	
Hypothesized Mean Difference	0	
df	3	
t Stat	-3.10811476	
P(T<=t) one-tail	0.02647963	
t Critical one-tail	2.353363435	
P(T<=t) two-tail	0.05295926	
t Critical two-tail	3.182446305	

(c) t-Test: Paired Two Sample for Means (Untreated and Silicon twice + no fungicides)

Results from figure 3.7 and table appendix 22 c has showed a border line (P>0.05) between untreated strawberry leaves and strawberry leaves treated with silicon twice + no fungicides. As the two-tail value is 0.052, it is greater than 0.05. Therefore, the difference between silicon contents in both treatment is not significant.

Appendix 23 Protocol for LysoTracker Yellow HCK-123 fluorescence dye staining (Chapter 4)

The LysoTracker Yellow HCK-123 fluorescence dye was used to stain cross-sections separately on slides or petri dishes covered from direct light for 2 hours. The Vectashield mounting medium was used in drops per slide to preserve samples from drying out while staining.

After staining (2 hours) cross-sections are viewed using the GXML3201 LED fluorescence microscope. Silicon fluorescence density was estimated using the software program ImageJ, which measures total fluorescence within a given sample.

Protocol

One vial contains 1µM (50µL) of (stock) Lyso-Tracker dye

De-ionized water (9 mL)

Vectashield (in drops)

1. Dilute $1\mu M$ (50 μL) of dye into 9 mL of de-ionized water

2. Transfer dye solution into a dark protective storage tube

3. Vortex (invert) solution

4. Pipette dye onto sample on slide or petri dish

5. Pipette drops of Vectashield on sample and coverslip to allow mounting medium to disperse over the entire section.

6. Allow samples to stain for 2 hours in the dark. Store dye in a cool dry place after use.

Appendix 24 Silicon fluorescence intensity quantification using ImageJ (Chapter 4)

Download free software program "ImageJ" and select the platform suitable to computer. Preferably Windows version.

- RUN program (ImageJ); To allow ImageJ toolbar display on screen (Top centre of screen)
- 2. Drag and drop sample images (one at a time) unto toolbar on screen; Image is now ready for measurements
- 3. Click on "Analyze" and click "set measurement" from the drop-down menu; Tick (mark) Integrated density ("Int Density") which measures total fluorescence

and "Area" (Un-mark all other set measurements not needed)

- Click on the free-hands tool from the toolbar (yellow tracing line); Trace the fluorescing areas of the cell (sample). Double-click to erase or re-draw
- Click on Analyze → Measure (or Ctrl 'M'); ImageJ will then produce measurements in a result table
- 6. Copy results from table and paste in Microsoft Excel;

In Excel, re-name samples i.e strawberry leaf petiole1 (Untreated control) etc.

7. Using the free-hands tool, measure a blank for each sample, i.e areas without fluorescence (Unknown background area) The blank is the black background.

Measure only a small area for blank. Click Analyze \rightarrow Measure (or Ctrl 'M'). Copy blank measurements from the results table displayed and paste unto a separate column or new spreadsheet on Microsoft Excel for calculations.

The blank (Unknown area) of cell subtracted from the Integrated density (Total fluorescence) of cell equals total quantity of fluorescence (without background fluorescence). The total fluorescence (Integrated density) given divided by the Area of cell then equals actual quantity of fluorescence per given sample.

Therefore;

Integrated density (Total fluorescence) – Blank (Unknown area of cell)

= Fluorescence of cell \div Area of cell

= Actual quantity of fluorescence

Counts per second (cps) is a unit for fluorescence and each value of actual fluorescence can be rounded off to a one or two decimal place value.

Appendix 25 Cross-sections (Replicates) of strawberry leaves in the glasshouse experiment 2017 (x40 and x400 magnifications) Chapter 4

Untreated leaves



Silicon treated leaves

Appendix 26 Cross-sections (Replicates) of strawberry leaves in the glasshouse experiment 2017 (x40 and x400 magnifications) cont'd, Chapter 4



Untreated leaves

Silicon-treated leaves

Images presented through fluorescence images in this appendix indicated that more deposits of silicon is found laid down in a regular manner and not in an amorphous form in the silicon treated plants presented on the right hand side compared to the untreated on the left hand side (Viewed at x40 and x400) throughout. The yellow arrows indicate the areas silicon has been found to accumulate more in this thesis hence the term, use 'Regularity in manner of deposition'. This was not always the case but mostly.

Appendix 27 : Cross-sections (Replicates) of strawberry leaves in the glasshouse experiment 2017 (x40 and x400 magnifications) cont'd, Chapter 4

Untreated leaves

Silicon-treated leaves



Appendix 28 Cross-sections (Replicates) of strawberry leaf petioles in the glasshouse experiments 2017 (x40 and x400 magnifications) Chapter 4



Appendix 29 Cross-sections (Replicates) of strawberry leaf petioles in the glasshouse experiments 2017 (x40 and x400 magnifications) Chapter 4



Untreated petioles Silicon-treated petioles

Appendix 30 Cross-sections (Replicates) of strawberry leaf petioles in the glasshouse experiments 2017 (x40 and x400 magnifications) cont'd, Chapter 4

Untreated (Leaf petiole) <mark>(q)</mark> Silicon treated (Leaf petiole) (r) Xylem Phloem Phloem Xylem X400 X400 <mark>(s)</mark> Untreated (Leaf petiole) Silicon treated (Leaf petiole) (t) Phloem Xylem Xylem X400 X400 Phloem

Untreated petioles

Silicon-treated petioles
Appendix 31 Cross-sections (Replicases) of strawberry roots in the glasshouse experiments 2017 (x40 and x400 magnifications) Chapter 4

Untreated

Silicon-treated roots



Appendix 32 Cross-sections (Replicases) of strawberry roots in the glasshouse experiments 2017 (x40 and x400 magnifications) cont'd, Chapter 4

Untreated

Silicon-treated roots



Appendix 33 Cross-sections (Replicases) of strawberry roots in the glasshouse experiments 2017 (x40 and x400 magnifications) cont'd, Chapter 4



Untreated

Silicon-treated roots

Appendix 34 Cross-sections (Replicates) of strawberry leaves in the silicon fertigation field 2017 (x40 and x400 magnifications) Chapter 4



Untreated leaves

Silicon treated leaves (Silicon twice only)



Appendix 35 Cross-sections (Replicates) of strawberry leaves in the silicon fertigation field 2017 (x40 and x400 magnifications) cont'd Chapter 4

Untreated

Silicon-treated leaves (twice only)



Appendix 36 : Cross-sections (Replicates) of strawberry leaves in the silicon fertigation field 2017 (x40 and x400 magnifications)

Untreated

Silicon-treated leaves (silicon twice only)



Appendix 37 Cross-sections (Replicates) of strawberry leaf petioles in the silicon field experiments 2017 (x40 and x400 magnifications) Chapter 4

Untreated







Appendix 38 Cross-sections (Replicates) of strawberry leaf petioles in the field experiment 2017 (x40 and x400 magnifications) cont'd, Chapter 4

Untreated





Appendix 39 Cross-sections (Replicates) of strawberry leaf petioles in the silicon field experiment 2017 (x40 and x400 magnifications) cont'd, Chapter 4



Untreated

Appendix 40 Cross-sections (Replicates) of strawberry roots in the field experiment 2017 (x40 and x400 magnifications) Chapter 4



Untreated

Silicon-treated roots (Silicon twice only)

Appendix 41 Cross-sections (Replicates) of strawberry roots in the field experiment 2017 (x40 and x400 magnifications) cont'd, Chapter 4

Untreated

Silicon-treated roots (Silicon twice only)



Appendix 42 Cross-sections (Replicates) of strawberry roots in the field experiment 2017 (x40 and x400 magnifications) cont'd, Chapter 4

Untreated

Silicon-treated roots (Silicon twice only)



Appendix 43 Cross-sections (Replicates) of strawberry leaves in the silicon fertigation field 2018 (x40 and x400 magnifications) Chapter 4



Untreated leaves

Silicon treated leaves (Silicon twice only)







Appendix 44 : Cross-sections (Replicates) of strawberry leaves in the silicon fertigation field 2018 (x40 and x400 magnifications) cont'd, Chapter 4

Untreated leaves



Appendix 45 Cross-sections (Replicates) of strawberry leaves in the silicon fertigation field 2018 (x40 and x400 magnifications) cont'd, Chapter 4



Untreated leaves

Appendix 46 Cross-sections (Replicates) of strawberry leaves in the silicon fertigation field 2018 (x40 and x400 magnifications) cont'd

Untreated leaves



Appendix 47 Cross-sections (Replicates) of strawberry leaf petioles in the silicon fertigation field 2018 (x40 and x400 magnifications) Chapter 4

Untreated petioles



Appendix 48 Cross-sections (Replicates) of strawberry leaf petioles in the silicon fertigation field 2018 (x40 and x400 magnifications) cont'd Chapter 4

Untreated petioles



Appendix 49 Cross-sections (Replicates) of strawberry leaf petioles in the silicon fertigation field 2018 (x40 and x400 magnifications) cont'd Chapter 4

Untreated petioles



Appendix 50 Cross-sections (Replicates) of roots in the silicon fertigation field 2018 (x40 and x400 magnifications) Chapter 4



Untreated roots Silicon-treated roots (Silicon twice only)

Appendix 51 Cross-sections (Replicates) of roots in the silicon fertigation field 2018 (x40 and x400 magnifications) cont'd Chapter 4

Untreated roots

Silicon-treated roots (Silicon twice only



Appendix 52 Fluorescence images of untreated and silicon-treated flower stalks and strawberry fruit, from 2018 silicon field, Chapter 4



Silicon deposition in the silicon fertigation field experiment 2018. Silicon deposits in green fluorescence. Viewed at x40 and x400 magnification.



Silicon deposition in the silicon fertigation field experiment 2018. Silicon deposits in green fluorescence. viewed at x 400 magnification only.



Cross section of strawberry fruit viewed through the GXML3201 fluorescence microscope

Appendix 53 Fluorescence images of untreated and silicon-treated only Achenes of strawberry plants, from 2018 silicon field, Chapter 4



Silicon deposition in the silicon fertigation field experiment 2018. Silicon deposits in green fluorescence. Viewed at x400 magnification only Achene (Seed) viewed through UV light under the GXML3201 fluorescence microscope

Appendix 54 (ImageJ data sheet) Raw fluorescence intensities from Chapter 4

	Results- Fluorscence intensities 2019-1.csv					
	Area	Mean	IntDen	RawIntDen		
1	40282	35.740	1439671.000	1439671.000		
2	2657	3.021	8028.000	8028.000		
3	44391	36.327	1612600.000	1612600.000		
4	1593	2.968	4728.000	4728.000		
5	87136	36.880	3213545.000	3213545.000		
6	60646	48.754	2956763.000	2956763.000		
7	2036	3.183	6481.000	6481.000		
8	58404	56.169	3280496.000	3280496.000		
9	5626	5.111	28753.000	28753.000		
10	10524	47.311	497906.000	497906.000		
11	20036	55.895	1119914.000	1119914.000		
12	23575	38.981	918983.000	918983.000		
13	17060	47.137	804162.000	804162.000		
14	11573	54.687	632895.000	632895.000		
15	5688	51.875	295065.000	295065.000		
16	27636	14.851	410434.000	410434.000		
17	3687	2.752	10148.000	10148.000		
18	4076	4.353	17743.000	17743.000		
19	53550	40.668	2177773.000	2177773.000		
20	33597	46.124	1549638.000	1549638.000		
21	9814	3.805	37339.000	37339.000		
22	3913	2.638	10323.000	10323.000		
23	50478	19.244	971403.000	971403.000		
24	3109	1.383	4300.000	4300.000		
25	51780	49.485	2562345.000	2562345.000		
26	1295	6.521	8445.000	8445.000		
27	44482	39.830	1771719.000	1771719.000		
28	24434	55.611	1358805.000	1358805.000		
29	2167	5.919	12827.000	12827.000		
30	259	15.564	4031.000	4031.000		
31	28956	50.188	1453234.000	1453234.000		
32	4048	20.954	84823.000	84823.000		
33	709	3.292	2334.000	2334.000		
34	48889	42.242	2065162.000	2065162.000		
35	28666	97.091	2783217.000	2783217.000		
36	55503	39.749	2206196.000	2206196.000		
37	231053	33.791	7807411.000	7807411.000		
38	14040	107.665	1511611.000	1511611.000		
39	1818	54.742	99521.000	99521.000		
40	3353	64.434	216046.000	216046.000		
41	3176	67.107	213132.000	213132.000		

Appendix 55 (ImageJ data sheet) Raw fluorescence intensities from Chapter 4

		Res	ults- Fluorscend	e intensities 2019-1.csv	
	Area	Mean	IntDen	RawIntDen	
41	3176	67.107	213132.000	213132.000	
42	2128	72.492	154262.000	154262.000	
43	2217	11.499	25494.000	25494.000	
44	308	12.162	3746.000	3746.000	
45	139	4.676	650.000	650.000	
46	1488	81.108	120689.000	120689.000	
47	4682	113.854	533066.000	533066.000	
48	4519	46.538	210305.000	210305.000	
49	4519	46.538	210305.000	210305.000	
50	430	53.251	22898.000	22898.000	
51	717	23.471	16829.000	16829.000	
52	2900	106.479	308790.000	308790.000	
53	3966	43.431	172248.000	172248.000	
54	9017	44.137	397987.000	397987.000	
55	9110	29.369	267549.000	267549.000	
56	4568	40.825	186487.000	186487.000	
57	3155	29.289	92407.000	92407.000	
58	1372	26.878	36877.000	36877.000	
59	1827	49.624	90663.000	90663.000	
60	3240	15.949	51674.000	51674.000	
61	296	25.811	7640.000	7640.000	
62	4038	57.488	232135.000	232135.000	
63	2539	100.273	254593.000	254593.000	
64	2347	55.149	129434.000	129434.000	
65	3815	27.331	104266.000	104266.000	
66	676	88.809	60035.000	60035.000	
67	676	88.809	60035.000	60035.000	
68	5994	98.967	593209.000	593209.000	
69	2519	100.773	253847.000	253847.000	
70	4010	36.218	145235.000	145235.000	
71	1810	12.742	23063.000	23063.000	
72	4038	36.081	145694.000	145694.000	
73	2588	29.172	75496.000	75496.000	
74	1313	24.969	32784.000	32784.000	
75	1948	29.407	57285.000	57285.000	
76	1668	40.923	68260.000	68260.000	
77	3606	39.818	143585.000	143585.000	
78	1860	22.355	41580.000	41580.000	
79	1876	32.436	60849.000	60849.000	
80	713	34.363	24501.000	24501.000	
81	4321	33.934	146630.000	146630.000	

		Res	ults- Fluorscenc	e intensities 2019-1	l.csv
	Area	Mean	IntDen	RawIntDen	
82	6551	19.318	126551.000	126551.000	
83	2255	14.957	33728.000	33728.000	
84	2924	61.148	178796.000	178796.000	
85	1386	17.317	24001.000	24001.000	
86	920	18.291	16828.000	16828.000	
87	4251	17.558	74638.000	74638.000	
88	2701	13.889	37513.000	37513.000	
89	2322	14.037	32594.000	32594.000	
90	14715	31.751	467215.000	467215.000	
91	1036	64.259	66572.000	66572.000	
92	2302	51.682	118972.000	118972.000	
93	4902	19.062	93440.000	93440.000	
94	2815	52.831	148718.000	148718.000	
95	4148	60.899	252607.000	252607.000	
96	2480	23.956	59410.000	59410.000	
97	5409	39.674	214599.000	214599.000	
98	2384	62.877	149899.000	149899.000	
99	1944	72.166	140291.000	140291.000	
100	5492	22.547	123827.000	123827.000	
101	11103	61.526	683125.000	683125.000	
102	4653	42.755	198939.000	198939.000	
103	2849	15.802	45019.000	45019.000	
104	2325	43.623	101423.000	101423.000	
105	2931	56.184	164674.000	164674.000	
106	2329	32.933	76701.000	76701.000	
107	3634	38.574	140177.000	140177.000	
108	4804	64.424	309494.000	309494.000	
109	5352	18.458	98786.000	98786.000	
110	1569	68.651	107713.000	107713.000	
111	1569	24.465	38385.000	38385.000	
112	14901	14.231	212053.000	212053.000	
113	3695	38.734	143123.000	143123.000	
114	5455	58.667	320029.000	320029.000	
115	5557	117.589	653444.000	653444.000	
116	5344	83.075	443955.000	443955.000	
117	5513	62.385	343931.000	343931.000	
118	5206	31.180	162325.000	162325.000	
119	5557	52.031	289134.000	289134.000	
120	5557	50.381	279968.000	279968.000	
121	4949	26.176	129543.000	129543.000	
122	4181	55.367	231488.000	231488.000	

Appendix 56 (ImageJ data sheet) Raw fluorescence intensities; Chapter 4

		Res	sults- Fluorscenc	e intensities 2019-1	l.csv
	Area	Mean	IntDen	RawIntDen	
121	4949	26.176	129543.000	129543.000	
122	4181	55.367	231488.000	231488.000	
123	5557	42.107	233991.000	233991.000	
124	210	15.386	3231.000	3231.000	
125	1120	25.438	28490.000	28490.000	
126	2591	17.764	46027.000	46027.000	
127	1842	56.015	103180.000	103180.000	
128	1540	47.056	72467.000	72467.000	
129	4917	40.169	197511.000	197511.000	
130	2719	41.121	111807.000	111807.000	
131	3700	35.744	132254.000	132254.000	
132	3700	35.744	132254.000	132254.000	
133	2786	41.413	115378.000	115378.000	
134	3418	37.319	127556.000	127556.000	
135	6798	26.133	177655.000	177655.000	
136	5141	47.514	244271.000	244271.000	
137	5141	14.255	73283.000	73283.000	
138	2278	47.303	107757.000	107757.000	
139	5536	35.574	196937.000	196937.000	
140	5414	30.594	165635.000	165635.000	
141	17660	12.654	223465.000	223465.000	
142	17660	47.058	831052.000	831052.000	
143	34719	27.916	969227.000	969227.000	
144	8687	29.059	252432.000	252432.000	
145	4562	25.451	116106.000	116106.000	
146	16257	52.237	849211.000	849211.000	
147	5818	25.731	149703.000	149703.000	
148	1797	24.200	43487.000	43487.000	
149	15467	41.330	639250.000	639250.000	
150	5559	48.645	270417.000	270417.000	
151	5014	25.794	129332.000	129332.000	
152	5978	48.292	288689.000	288689.000	
153	5978	52.086	311372.000	311372.000	
154	3865	47.790	184709.000	184709.000	
155	3865	47.790	184709.000	184709.000	
156	5429	25.232	136987.000	136987.000	
157	1846	27.645	51032.000	51032.000	
158	3518	99.807	351120.000	351120.000	
159	6786	37.725	255999.000	255999.000	
160	7004	31.623	221489.000	221489.000	
161	7004	21.084	147670.000	147670.000	

Appendix 57 (ImageJ data sheet) Raw fluorescence intensities; Chapter 4

		Res	ults- Fluorscence	intensities 2019-1	.csv
	Area	Mean	IntDen	RawIntDen	
161	7004	21.084	147670.000	147670.000	
162	3534	40.630	143588.000	143588.000	
163	4750	35.340	167867.000	167867.000	
164	4750	26.617	126430.000	126430.000	
165	5412	44.717	242009.000	242009.000	
166	5412	89.868	486366.000	486366.000	
167	4285	113.072	484514.000	484514.000	
168	6778	33.676	228257.000	228257.000	
169	311	39.006	12131.000	12131.000	
170	2584	23.439	60567.000	60567.000	
171	4204	26.344	110750.000	110750.000	
172	9859	51.011	502916.000	502916.000	
173	16615	49.074	815367.000	815367.000	
174	4243	38.660	164034.000	164034.000	
175	5488	25.835	141780.000	141780.000	
176	713	30.302	21605.000	21605.000	
177	7144	36.987	264236.000	264236.000	
178	7459	109.554	817163.000	817163.000	
179	7459	109.554	817163.000	817163.000	
180	1735	32.086	55670.000	55670.000	
181	5039	25.588	128936.000	128936.000	
182	4917	22.559	110921.000	110921.000	
183	5664	24.909	141082.000	141082.000	
184	5664	24.090	136446.000	136446.000	
185	8158	105.220	858384.000	858384.000	
186	5281	25.459	134450.000	134450.000	
187	353	44.691	15776.000	15776.000	
188	1756	36.952	64887.000	64887.000	
189	1211	71.141	86152.000	86152.000	
190	1525	69.990	106735.000	106735.000	
191	92	20.685	1903.000	1903.000	
192	3018	63.712	192283.000	192283.000	
193	3193	52.062	166234.000	166234.000	
194	4343	58.165	252610.000	252610.000	
195	568	50.533	28703.000	28703.000	
196	1809	14.457	26153.000	26153.000	
197	1767	26.906	47543.000	47543.000	
198	1093	47.191	51580.000	51580.000	
199	880	61.183	53841.000	53841.000	
200	3947	38.626	152458.000	152458.000	
201	478	30.084	14380.000	14380.000	

Appendix 58 ImageJ data sheet) Raw fluorescence intensities; Chapter 4

		Res	sults- Fluorscence	e intensities 2019-1.csv
	Area	Mean	IntDen	RawIntDen
178	7459	109.554	817163.000	817163.000
179	7459	109.554	817163.000	817163.000
180	1735	32.086	55670.000	55670.000
181	5039	25.588	128936.000	128936.000
182	4917	22.559	110921.000	110921.000
183	5664	24.909	141082.000	141082.000
184	5664	24.090	136446.000	136446.000
185	8158	105.220	858384.000	858384.000
186	5281	25.459	134450.000	134450.000
187	353	44.691	15776.000	15776.000
188	1756	36.952	64887.000	64887.000
189	1211	71.141	86152.000	86152.000
190	1525	69.990	106735.000	106735.000
191	92	20.685	1903.000	1903.000
192	3018	63.712	192283.000	192283.000
193	3193	52.062	166234.000	166234.000
194	4343	58.165	252610.000	252610.000
195	568	50.533	28703.000	28703.000
196	1809	14.457	26153.000	26153.000
197	1767	26.906	47543.000	47543.000
198	1093	47.191	51580.000	51580.000
199	880	61.183	53841.000	53841.000
200	3947	38.626	152458.000	152458.000
201	478	30.084	14380.000	14380.000
202	739	71.622	52929.000	52929.000
203	3153	68.751	216773.000	216773.000
204	3153	64.734	204106.000	204106.000
205	1978	47.892	94730.000	94730.000
206	919	65.252	59967.000	59967.000
207	374	61.037	22828.000	22828.000
208	204	71.431	14572.000	14572.000
209	346	38.760	13411.000	13411.000
210	286	21.545	6162.000	6162.000
211	683	52.919	36144.000	36144.000
212	797	78.881	62868.000	62868.000
213	/9/	/8.881	62868.000	62868.000
214	2676	66.557	178106.000	178106.000
215	1215	51.583	62673.000	62673.000
216	1215	51.583	626/3.000	626/3.000
217	626	/1.158	44545.000	44545.000
218	1390	48.004	66726.000	66726.000

Appendix 59 (ImageJ data sheet) Raw fluorescence intensities; Chapter 4

Appendix 60 Chapter 4 - Results workings Means of 10 fluorescence intensities (integrated density) in the leaf, leaf petiole and root from a glasshouse experiment 2017

Normality test (a) and (b) One Way Anova in cross-sections of untreated and silicon-treated leaves in a glasshouse experiment 2017

A. Normality test, Null and alternate Hypothesis: leaves

Ho: Data follows normal distribution.

Ha: Data does not follow normal distribution.

Level of significance = 0.05

· · · · · · · · · · · · · · · · · · ·							
	Kolm	ogorov-Smir	nov ^a	Shapiro-Wilk			
	Statistic	df	Sig.	Statistic	df	Sig.	
Response	.145	20	.200	.907	20	.057	

Tests of Normality

*. This is a lower bound of the true significance.

a. Lilliefors Significance Correction

Since the sig value.057 is greater than critical value, accept Ho and conclude that data does follow normal distribution. The One Way Anova was then used to determine the average values of differences between fluorescence intensities of untreated leaves and silicon-treated leaves. See table 4.2 b for the One Way Anova test used to evaluate the cross-section average values.

B. One Way Anova

Response					
	Sum of Squares	df	Mean Square	F	Sig.
Between Groups	22.898	1	22.898	7.142	.016
Within Groups	57.710	18	3.206		
Total	80.608	19			

ANOVA

b between silicon-treated leaves and untreated leaves, shows that since the sig value.016 is less than critical value 0.05, reject Ho and conclude that there is a statistical significant difference between the average values of silicon-treated leaves and untreated leaves in the glasshouse experiment. This shows that silicon-treated plants had more silicon deposits (Fluorescence intensity) than untreated.

(a) Normality (b) One Way Anova test in cross-sections of untreated and silicon-treated leaf petioles in a glasshouse experiment 2017

A. Normality test, Null and alternate Hypothesis: Leaf petioles **Ho:** Data follows normal distribution.

Ha: Data does not follow normal distribution.

Level of significance = 0.05

Tests of Normality

	Kolmogorov-Smirnov ^a				Shapiro-Wilk		
	Statistic	df	Sig.	Statistic	df	Sig.	
Response	.177	20	.102	.909	20	.060	

a. Lilliefors Significance Correction

The test for normality between untreated and silicon-treated leaf petioles in table 4.3 (a) show that since the sig value.060 is greater than critical value, accept Ho and conclude that the data does follow normal distribution. Therefore, the One Way Anova was used to determine average values of differences between leaf petioles from untreated and silicon-treated plants. See figure b for the One Way Anova test used.

B. One Way Anova, Null and Alternate Hypothesis: Leaf petioles

Ho: There is no statistical significance difference among the treatment average values **Ha:** There is statistical significance difference among the treatment average values

Response					
	Sum of Squares	df	Mean Square	F	Sig.
Between Groups	81.205	1	81.205	22.604	.000
Within Groups	64.665	18	3.593		
Total	145.870	19			

ANOVA

From table (b) since sig the value.000 is less than critical value 0.05, reject Ho and conclude that there is a statistical significant difference among the treatment average values of leaf petioles of untreated and silicon-treated plants. This shows that there was more silicon deposits (Fluorescence intensity) in the silicon-treated plants compared to untreated.

Table (a) Normality test and (b) One Way Anova in cross-sections of untreated and silicon-treated roots in a glasshouse experiment 2017

A. Normality test, Null and alternate Hypothesis: Roots

Ho: Data follows normal distribution.

Ha: Data does not follow normal distribution.

Level of significance = 0.05

Tests of Normality

	Kolmogorov-Smirnov ^a			Shapiro-Wilk		
	Statistic	df	Sig.	Statistic	df	Sig.
Response	.086	20	.200	.991	20	.999

*. This is a lower bound of the true significance.

a. Lilliefors Significance Correction

From table a of untreated and silicon-treated roots, since the sig value.999 is greater than critical value, accept Ho and conclude that the data does follow normal distribution. The One Way Anova (Table b) was used to evaluate the differences of average values of cross-sections of the roots between untreated and silicon-treated.

B. One Way Anova, Null and Alternate Hypothesis

Ho: There is no statistical significant difference among the treatment average values **Ha:** There is statistical significant difference among the treatment average values

Response					
	Sum of Squares	df	Mean Square	F	Sig.
Between Groups	116.162	1	116.162	31.283	.000
Within Groups	66.838	18	3.713		
Total	183.000	19			

ANOVA

In table b, the sig value.000 is less than critical value 0.05, reject Ho and conclude that there is a statistical significance difference among the treatment average values of roots of untreated and silicon-treated strawberry plants. The One Way Anova showed that average values of silicon-treated plant roots is statistical different from untreated plant roots. Table b shows that there was more silicon deposits (Fluorescence intensity) in the silicon-treated roots compared to untreated roots.

Appendix 61 Chapter 4 - Results workings Means of 10 fluorescence intensities (integrated density) in the leaf, leaf petiole and root from a field experiment 2017

Table a Normality test and b Two Way Anova in cross-sections of untreated and silicon twice only treatment leaves, leaf petioles and roots in a field experiment 2017

A. Normality test, Null and alternate Hypothesis: Leaves, leaf petioles and roots **Ho:** Data follows normal distribution.

Ha: Data does not follow normal distribution.

Level of significance = 0.05

	Kolmogorov-Smirnov ^a			Shapiro-Wilk		
	Statistic	df	Sig.	Statistic	df	Sig.
Response	.118	120	.306	.930	120	.502

Tests of Normality

a. Lilliefors Significance Correction

From table a for untreated and silicon twice only treatment for the leaves, leaf petioles and roots, since the sig value.502 is greater than critical value, accept Ho and conclude that data does follow normal distribution. Therefore, the Two Way Anova was used to determine the difference between the average values of cross-sections from both treatment. The Two Way Anova in table b was used to test for the average values of differences between untreated and silicon twice only treatment in the leaves, leaf petioles and roots of the 2017 field experiment.

B. Two Way Anova

Null and alternate Hypothesis:

Ho: Treatment average values of are not statistically significantly different

Ha: Treatment average values of are statistically significantly different

Ho: There is no statistical significant difference between block average values

Ha: There is statistical significant difference between block average values

Tests of Between-Subjects Effects

Dependent Variable: Response						
Source	Type III Sum of Squares	df	Mean Square	F	Sig.	
Corrected Model	283.544 ^a	11	25.777	3.825	.000	
Intercept	6442.778	1	6442.778	955.918	.000	
Treatment	203.141	3	67.714	10.047	.000	
Block	21.143	2	10.572	1.569	.213	
Treatment * Block	59.260	6	9.877	1.465	.197	
Error	727.907	108	6.740			
Total	7454.229	120				
Corrected Total	1011.451	119				

a. R Squared = .280 (Adjusted R Squared = .207)

Appendix 62 Chapter 4 - Results workings Means of 10 fluorescence intensities (integrated density) in the leaf, leaf petiole and root from a field experiment 2018

Table (a) Normality test and (b) Two Way Anova in cross-sections of untreated and silicon twice only treatment leaves, leaf petioles and roots of the field experiment 2018

A. Normality test, Null and alternate Hypothesis: Leaves, leaf petioles and roots **Ho:** Data follows normal distribution.

Ha: Data does not follow normal distribution.

Level of significance = 0.05

Tests of Normality

	Kolmogorov-Smirnov ^a			Shapiro-Wilk		
	Statistic	df	Sig.	Statistic	df	Sig.
Response	.061	120	.200	.978	120	.051

*. This is a lower bound of the true significance.

a. Lilliefors Significance Correction

From table a for untreated and silicon twice only treatment cross-sections, since the sig value 0.051 is greater than 0.05 accept Ho and conclude that data follows normal distribution. The Two Way Anova was used to find the differences between the average values of untreated and silicon twice only treatment from table b.

B. Two Way Anova

Null and alternate Hypothesis:

Ho: Treatment average values of are not statistically significantly different

Ha: Treatment average values of are statistically significantly different

Ho: There is no statistically significant difference between block average values

Ha: There is statistically significant difference between block average values

Dependent Variable:	Response				
Source	Type III Sum of Squares	df	Mean Square	F	Sig.
Corrected Model	70.205 ^a	11	6.382	2.220	.018
Intercept	2109.498	1	2109.498	733.690	.000
Treatment	6.419	3	2.140	.744	.528
Block	45.815	2	22.907	7.967	.001
Treatment * Block	17.970	6	2.995	1.042	.403
Error	310.521	108	2.875		
Total	2490.224	120			
Corrected Total	380.725	119			

Tests of Between-Subjects Effects

From results in table 4.8b, the Two Way Anova revealed that there was a statistical significant difference between leaves of silicon-treated plants and untreated plants (P<0.05). However, no statistical differences (P>0.05) was found between silicon-treated and untreated leaf petioles and roots in the silicon fertigation field experiment 2018.

4.7b, the sig value of treatment average values shows that there is a statistical significant difference among treatment average values because sig value.000 is less than critical value 0.05. Since the sig value.213 shows that there is no statistical significant difference among block average values. The conclusion from the Two Way Anova conducted between fluorescence intensities of cross-sections values showed that there was a statistical significant difference between cross-sections of leaves in silicon-treated leaves compared to untreated leaves. The results also showed that there was no statistical significance difference between cross-sections of leaf petioles and roots in silicon-treated plants compared to untreated. Data from the Two Way Anova analysis has shown that silicon-treated fluorescence deposits were higher in silicon-treated cross-sections compared to untreated in the 2017 field experiment.
Appendix 63 Hoagland's solution recipe (chapter 5)

To make up Hoagland's stock solution the following procedures were carried out in order

Each stock component was weighed and transferred into separate plastic storage bottles with appropriate labels.

Each made up stock component was pipetted to 800 mL de-ionized water and then filled to 1 Litre.

The solution was mixed thoroughly and transferred into 10 Litre plastic storage tanks.

10 L x 4 plastic storage tanks containing made up Hoagland's solution was added to 5 L plastic boxes in preparation for the planting of the Malling Centenary strawberry plants.

pH and EC was measured with a pH and EC metre before planting the strawberry plants. The pH was between 5.5 to 6.0 and EC was between 1.0dS/m to 1.3dS/m.

Hoagland's solution was topped in hydroponic tubs twice-a-week as levels reduced as plants grow in size. Forty Litres of Hoagland's were prepared prior to top ups

Appendix 64 Chapter 5 - Results workings - Means of 10 strawberry fruits °Brix levels from the silicon field experiment 2017

 i. (i) Test for normality and (ii) & (iii) t-Test Paired Sample for Means Strawberry ^oBrix, Untreated and Silicon twice + fungicides and silicon twice + fungicides (20 of July 2017)

ii. Normality Test

Null and alternate Hypothesis:

Ho: Data follows normal distribution.

Ha: Data does not follow normal distribution.

Level of significance = 0.05

Tests of Normality

	Kolmogorov-Smirnov ^a			Shapiro-Wilk		
	Statistic df Sig.		Statistic df S		Sig.	
Response	.069	36	.200	.986	36	.911

*. This is a lower bound of the true significance.

a. Lilliefors Significance Correction

In table (i) if the sig value is less than critical value 0.05, reject Ho. In the analysis for samples assessed on 20 of July 2017, the sig value 0.911 is greater than critical value 0.05. Accept Ho and conclude that the data follows a normal distribution. The t-Test Paired Two Samples for Means was used to determine the differences between average values of strawberries sampled on 20 of July 2017.

Table (ii) showed that the average values of strawberries °Brix sampled on 20 of July 2017 from the silicon twice + fungicides treatment was not statistically (P>0.05) different from untreated. There was no increase in silicon-treated fruit °Brix compared to untreated.

iii. t-Test: Paired Two Sample for Means: Strawberry °Brix, Untreated and Silicon twice + no fungicides (20 of July 2017)

	Untreated	Silicon x 2 + no fungicides
Mean	9.866666667	9.35555556
Variance	0.6075	1.832777778
Observations	9	9

Pearson Correlation	0.182037825
Hypothesized Mean Difference	0
df	8
t Stat	1.069333782
P(T<=t) one-tail	0.15805972
t Critical one-tail	1.859548038
P(T<=t) two-tail	0.316119439
t Critical two-tail	2.306004135

Table (iii) showed that the average values of strawberries °Brix sampled on 20 of July 2017 from the silicon twice + no fungicides treatment was not statistically (P>0.05) different from untreated. Conclusions from table a to c showed that °Brix levels of strawberry fruits sampled on the 20 of July 2017 in the silicon-treated plants was not statistically different (P>0.05) from untreated.

(iv) Test for normality and (v) & (vi) t-Test Paired Sample for Means Strawberry °Brix, Untreated and Silicon twice + fungicides and silicon twice + fungicides (24 of July 2017)

iv. Normality Test

Null and alternate Hypothesis:

Ho: Data follows normal distribution.

Ha: Data does not follow normal distribution.

Level of significance = 0.05

Tests of Normality

	Kolmogorov-Smirnov ^a			Shapiro-Wilk		
	Statistic df Sig.		Statistic df Si		Sig.	
Response	.161	40	.010	.927	40	.013

a. Lilliefors Significance Correction

As the sig value.013 in table (iv) is less than the critical value 0.05, reject Ho and conclude that data does not follow normal distribution, so the t-Test Paired Two Samples for Means was used to test for the significance difference between untreated and silicon twice + fungicides treatment sampled 24 of July 2017.

	Untreated	Silicon x 2 + fungicides
Mean	8.32	10.94
Variance	2.166222222	4.936
Observations	10	10
Pearson Correlation	0.166568057	
Hypothesized Mean Difference	0	
df	9	
t Stat	-3.378778344	
P(T<=t) one-tail	0.004071372	
t Critical one-tail	1.833112933	
P(T<=t) two-tail	0.008142743	
t Critical two-tail	2.262157163	

v.t-Test: Paired Two Sample for Means: Strawberry °Brix Untreated and Silicon twice + fungicides (24 of July 2017)

Table (v) showed that the average values of strawberries °Brix sampled on 24 of July 2017 from the silicon twice + fungicides treatment was statistically (P<0.05) different from untreated. °Brix levels of strawberry fruits sampled from the silicon twice + fungicides was higher than untreated.

	Untreated	Silicon x 2 + no fungicides
Mean	8.32	11.72
Variance	2.166222222	9.30844444
Observations	10	10
	-	
Pearson Correlation	0.355915667	
Hypothesized Mean Difference	0	
df	9	
	-	
t Stat	2.807032325	
P(T<=t) one-tail	0.010238031	
t Critical one-tail	1.833112933	
P(T<=t) two-tail	0.020476062	
t Critical two-tail	2.262157163	

vi. t-Test: Paired Two Sample for Means: Strawberry °Brix Untreated and Silicon twice + no fungicides (24 of July 2017)

Table (vi) showed the average values of strawberries °Brix sampled on the 24 of July 2017 from untreated and silicon twice + no fungicides was statistically different (P<0.05). Conclusions from table i to vi showed that silicon-enhanced °Brix levels of strawberry fruits sampled from the silicon-treated plants on 24 of July 2017 compared to untreated. Both silicon treatment were higher than untreated on the 24 of July 2017.

vii. Test for normality and (viii) & (ix) t-Test Paired Two Sample for Means Untreated and Silicon twice + fungicides and silicon twice + fungicides (27 of July 2017)

vii. Normality Test

Null and alternate Hypothesis:

Ho: Data follows normal distribution.

Ha: Data does not follow normal distribution.

Level of significance = 0.05

_

Tests of Normality Kolmogorov-Smirnov^a Shapiro-Wilk Statistic df Statistic Sig. df Sig. Response .125 40 .117 40 .027 .937

From table (vii) since the sig value.027 is less than critical value 0.05, reject Ho and conclude that data (Means of strawberries sampled 27 of July 2017) does not follow normal distribution. The t-test Paired Sample for Means was then used to test for significance differences between untreated and silicon twice + fungicides treatment.

viii. t-Test: Paired Two Sample for Means: Strawberry °Brix Untreated and Silicon twice + fungicides (27 of July 2017)

	Untreated	Silicon x 2 fungicides
Mean	9.9	10.94
Variance	2.533333333	4.936
Observations	10	10
Pearson Correlation	0.076667901	
Hypothesized Mean Difference	0	
df	9	
t Stat	-1.249560426	
P(T<=t) one-tail	0.121488767	
t Critical one-tail	1.833112933	
P(T<=t) two-tail	0.242977533	
t Critical two-tail	2.262157163	

Table (viii) showed that the average values of strawberries °Brix sampled on 27 of July 2017 from the silicon twice + fungicides treatment was not statistically (P>0.05) different from untreated.

			Silicon x 2 +
		Untreated	no fungicides
Mean		9.9	10.45
Variance		2.533333333	6.138333333
Observations		10	10
Pearson Correlation		-0.364039577	
Hypothesized	Mean		
Difference		0	
df		9	
t Stat		-0.511926395	
P(T<=t) one-tail		0.310513352	
t Critical one-tail		1.833112933	
P(T<=t) two-tail		0.621026704	
t Critical two-tail		2.262157163	

ix. t-Test: Paired Two Sample for Means: Strawberry °Brix Untreated and Silicon twice + no fungicides (27 of July 2017)

Table (ix) showed that the average values of strawberries sampled on the 27 July from untreated and silicon twice + no fungicides was not statistically different (P>0.05). Conclusions from table (vii, viii and ix) showed that on the 27 of July 2017, Strawberries from the silicon twice + fungicides and silicon twice + no fungicides treatment were not statistically (P>0.05) different from untreated.

(xi) Test for normality and (xii) & (xiii) t-Test Paired Two Sample for Means Untreated and Silicon twice + fungicides and silicon twice + fungicides (01 of August 2017)

xi. Normality Test

Null and alternate Hypothesis:

Ho: Data follows normal distribution.

Ha: Data does not follow normal distribution.

Level of significance = 0.05

Tests of Normality							
	Kolmogorov-Smirnov ^a Shapiro-Wilk						
Statistic df Sig. Statistic df						Sig.	
Response	.103	40	.200	.963	40	.207	

*. This is a lower bound of the true significance.

a. Lilliefors Significance Correction

n table (xi) the sig value.207 is greater than the critical value.05, accept Ho and conclude that data (Means of strawberries sampled 01 of August 2017) follows a normal distribution. The t-Test Paired Two Sample for Means was used to test for significant differences between treatment average values of strawberries sampled on 01 of August 2017.

Ι

	Untreated	Silicon x 2 + fungicides
Mean	9.47	9.66
Variance	2.042333333	3.633777778
Observations	10	10
	_	
Pearson Correlation	0.301492806	
Hypothesized Mean Difference	0	
df	9	
	_	
t Stat	0.222092719	
P(T<=t) one-tail	0.414599768	
t Critical one-tail	1.833112933	
P(T<=t) two-tail	0.829199535	
t Critical two-tail	2.262157163	

xii. t-Test: Paired Two Sample for Means: Strawberry ^oBrix Untreated and Silicon twice + fungicides (01 of August 2017)

Table (xii) showed that the average values of strawberries °Brix sampled on 01 of August 2017 from the silicon twice + fungicides treatment were not statistically (P>0.05) different from untreated.

	Untreated	Silicon x 2 + no fungicides
Mean	9.47	9.73
Variance	2.042333333	1.382333333
Observations	10	10
Pearson Correlation	-0.365755989	
Hypothesized Mean Difference	0	
df	9	
t Stat	-0.381127712	
P(T<=t) one-tail	0.35597482	
t Critical one-tail	1.833112933	
P(T<=t) two-tail	0.711949641	
t Critical two-tail	2.262157163	

xiii. t-Test: Paired Two Sample for Means: Strawberry ^oBrix Untreated and Silicon twice + no fungicides (01 of August 2017)

Table (xiii) showed that there was no significant difference (P>0.05) of average values of °Brix levels of strawberries sampled 01 of August 2017 between untreated and silicon twice + no fungicides treatment. Conclusions from table (xi, xii, xiii) showed that silicon did not increase °Brix levels of strawberries sampled on 01 of August 2017.

Appendix 65 Chapter 5 - Results workings - Means of 10 strawberry leaf petioles °Brix levels from the silicon field experiment 2017

(xiv) Test for normality and (xv) and (xvi) t-test: Paired Two Samples for Means: Leaf petioles °Brix, Untreated and silicon twice + fungicides and silicon twice + no fungicides (20 June 2017)

xiv. Normality Test

Null and alternate Hypothesis:

Ho: Data follows normal distribution.

Ha: Data does not follow normal distribution.

Level of significance = 0.05

	Kolmogorov-Smirnov ^a			Shapiro-Wilk		
	Statistic	df	Sig.	Statistic	df	Sig.
Untreated	.152	10	.200	.955	10	.732
Fungicides	.194	10	.200	.916	10	.324
Silicon_x_2Fungicides	.197	10	.200	.927	10	.418
Silicon_x_2_no_Fungicid es	.172	10	.200	.934	10	.488

Tests of Normality

*. This is a lower bound of the true significance.

a. Lilliefors Significance Correction

If the sig value in table (xiv) is less than the critical value 0.05, reject Ho and conclude that data does not follow normal distribution. Since the sig value is greater than the critical value 0.05 accept the Ho and conclude that data follows normal distribution. The t-Test Paired Two Sample for Means was used to test for significance difference between treatment (Untreated, silicon twice + fungicides and silicon twice + no fungicides) average values ofs for leaf petioles sampled 20 June 2017 (table xv).

	Untreated	Silicon x 2 + fungicides
Mean	8.75	9.86
Variance	1.542777778	3.464888889
Observations	10	10
Pearson Correlation	0.196555163	
Hypothesized Mean Difference	0	
df	9	
t Stat	-1.7337872	
P(T≤=t) one-tail	0.05849371	
t Critical one-tail	1.833112933	
P(T<=t) two-tail	0.116987421	
t Critical two-tail	2.262157163	

xv. t-Test: Paired Two Sample for Means: Leaf petioles °Brix Untreated Silicon twice + fungicides (20 June 2017)

Table (xv) showed that the average values of strawberry leaf petioles °Brix sampled on 20 June 2017 from the silicon twice + fungicides treatment was not statistically (P>0.05) different from untreated.

	Untreated	Silicon x $2 + no$ fungicides
Mean	8.75	8.49
Variance	1.542777778	2.387666667
Observations	10	10
Pearson Correlation	-0.344747179	
Hypothesized Mean Difference	0	
df	9	
t Stat	0.358705056	
P(T<=t) one-tail	0.36404313	
t Critical one-tail	1.833112933	
P(T<=t) two-tail	0.72808626	
t Critical two-tail	2.262157163	

xvi. t-Test: Paired Two Sample for Means: Leaf petioles °Brix Untreated and Silicon twice + no fungicides (20 June 2017)

Table (xvi) showed that the average values of strawberry leaf petioles °Brix sampled on 20 June 2017 from the silicon twice + no fungicides treatment was not statistically (P>0.05) different from untreated. Conclusions from table (xiv, xv, xvi) showed that silicon did not elevate °Brix levels of leaf petioles sampled 20 June 2017.

(xvii) Test for normality and (xviii) and (xix) t-test: Paired Two Samples for Means: Leaf petioles °Brix, Untreated and silicon twice + fungicides and silicon twice + no fungicides (18 of July 2017)

xvii. Normality test

Null and alternate Hypothesis:

Ho: Data follows normal distribution.

Ha: Data does not follow normal distribution.

Level of significance = 0.05

Tests of Normality

	Kolm	ogorov-Smir	nov ^a	5		
	Statistic	df	Sig.	Statistic	tistic df S	
Untreated	.186	10	.200	.958	10	.759
Fungicides	.139	10	.200	.932	10	.464
Silicon_x_2Fungicides	.179	10	.200	.904	10	.240
Silicon_x_2_no_Fungicid es	.249	10	.079	.868	10	.096

*. This is a lower bound of the true significance.

a. Lilliefors Significance Correction

From table (xvii) if the sig value is less than critical value, reject Ho and conclude that data does not follow normal distribution. Since the sig value is greater than critical value 0.05 accept Ho and conclude that data follows normal distribution. The t-Test Paired Two Sample for Means was used to test for significance differences between untreated and silicon twice + fungicides (Table xviii).

	Untreated	Silicon x 2 + fungicides
Mean	8.45	7.9
Variance	2.613888889	1.68444444
Observations	10	10
Pearson Correlation	-0.311360065	
Hypothesized Mean Difference	0	
df	9	
t Stat	0.734640524	
P(T<=t) one-tail	0.240630197	
t Critical one-tail	1.833112933	
P(T<=t) two-tail	0.481260393	
t Critical two-tail	2.262157163	

xviii. t-Test: Paired Two Sample for Means: Leaf petioles °Brix Untreated and Silicon twice + fungicides (18 of July 2017)

Table (xviii) showed that the average values of strawberry leaf petioles °Brix sampled on 18 of July 2017 from the silicon twice + fungicides treatment was not statistically (P>0.05) different from untreated.

	Untreated	Silicon x 2 + no fungicides
Mean	8.45	8.84
Variance	2.613888889	1.556
Observations	10	10
Pearson Correlation	0.226989804	
Hypothesized Mean Difference	0	
df	9	
t Stat	-0.68364964	
P(T<=t) one-tail	0.255705985	
t Critical one-tail	1.833112933	
P(T<=t) two-tail	0.511411971	
t Critical two-tail	2.262157163	

xix. t-Test: Paired Two Sample for Means: Leaf petioles °Brix Untreated and Silicon twice + no fungicides

Table (xix) showed that the average values of strawberry leaf petioles °Brix sampled on 18 of July 2017 from the silicon twice + no fungicides treatment was not statistically (P>0.05) different from untreated. Table (xvii, xviii, xix) suggests that silicon did not increase °Brix levels of leaf petioles in silicon-treated plants sampled 18 of July 2017.

(xxi) Test for normality and (xxii) and (xxiii) t-test: Paired Two Samples for Means: Leaf petioles 'Brix, Untreated and silicon twice + fungicides and silicon twice + no fungicides (01 of August 2017)

xxi. Normality test

Null and alternate Hypothesis:

Ho: Data follows normal distribution.

Ha: Data does not follow normal distribution.

Level of significance = 0.05

	Kolmogorov-Smirnov ^a Shapiro-Wilk					
	Statistic	df	Sig.	Statistic	df	Sig.
Untreated	.298	10	.012	.863	10	.083
Fungicides	.223	10	.172	.921	10	.369
Silicon_x_2Fungicides	des .186 10	.200	.200 .940		.550	
Silicon_x_2_no_Fungicid	.159	10	.200	.964	10	.833

Tests of Normality

*. This is a lower bound of the true significance.

a Lilliefors Significance Correction

In table (xxi) If sig value is less than critical value, reject Ho and conclude that data does not follow normal distribution. Since the sig value is greater than critical value 0.05 we accept Ho and conclude that data follows normal distribution. The t -Test: Paired Two Sample for Means was used to test for significant difference between untreated, silicon twice + fungicides and silicon twice + no fungicides (Table xxii and xxiii).

xxii. t-Test: Paired Two Sample for Means:

8.99 3.016555556 10	8.86 1.387111111
3.016555556 10	1.387111111
10	10
	10
0.447908735	
0	
9	
0.256373744	
0.401713182	
1.833112933	
0.803426364	
0.0(01551(0	
	0 9 0.256373744 0.401713182 1.833112933 0.803426364 2.262157163

Strawberry Leaf petioles °Brix Untreated and Silicon twice + fungicides (01 of August 2017)

Table (xxii) showed that the average values of strawberry leaf petioles °Brix sampled on 01 of August 2017 from the silicon twice + fungicides treatment was not statistical higher (P>0.05) and different from untreated.

xxiii. t-Test: Paired Two Sample for Means: Strawberry Leaf petioles ^oBrix Untreated and Silicon twice + no fungicides (01 of August 2017)

	Untreated	Silicon x 2 + no fungicides
Mean	8.99	7.11
Variance	3.016555556	1.289888889
Observations	10	10
Pearson Correlation	-0.344672099	
Hypothesized Mean Difference	0	
df	9	
t Stat	2.497531922	
P(T<=t) one-tail	0.016999616	
t Critical one-tail	1.833112933	
P(T<=t) two-tail	0.033999232	
t Critical two-tail	2.262157163	

Table (xxiii) showed that the average values of leaf petioles °Brix sampled on 01 of August 2017 from the silicon twice + no fungicides treatment was not statistical higher (P>0.05) and different from untreated. Table (xxi, xxii, xxiii) showed that silicon did not elevate °Brix levels of leaf petioles sampled 01 of August 2017.

(xxiv) Test for normality and (xxv) and (xxvi) t-test: Paired Two Samples for Means: Leaf petioles 'Brix, Untreated and silicon twice + fungicides and silicon twice + no fungicides (19 September 2017)

xxiv. Normality test

Null and alternate Hypothesis:

Ho: Data follows normal distribution.

Ha: Data does not follow normal distribution.

Level of significance = 0.05

Tests of Normality							
	Kolm	ogorov-Smir	nov ^a	:	Shapiro-Wilk		
	Statistic	df	Sig.	Statistic	df	Sig.	
Untreated	.174	10	.200	.947	10	.631	
Fungicides	.158	10	.200	.978	10	.951	
Silicon_x_2Fungicides	.216	10	.200	.908	10	.268	
Silicon_x_2_no_Fungicid es	.197	10	.200	.928	10	.425	

*. This is a lower bound of the true significance.

From table (xxiv) If the sig value is less than critical value, reject Ho and conclude that data does not follow normal distribution. However, since the sig value is greater than critical value 0.05, accept Ho and conclude that data follows normal distribution. The Paired Two Sample for Means was then used to find the differences between untreated and silicon twice + fungicides in table xxv.

xxv. t-Test: Paired Two Sample for Means:

Strawberry Leaf petiole °Brix Untreated and Silicon twice + fungicides (19 September 2017)

	Untreated	Silicon x 2 + fungicides
Mean	8.52	9.13
Variance	1.832888889	3.909
Observations	10	10
Pearson Correlation	-0.517883354	
Hypothesized Mean Difference	0	
df	9	
t Stat	-0.661080749	
P(T<=t) one-tail	0.262560136	
t Critical one-tail	1.833112933	
P(T<=t) two-tail	0.525120271	
t Critical two-tail	2.262157163	

Table 5.12(xxv) showed that the average values of leaf petioles °Brix sampled on 19 September 2017 from the silicon twice + fungicides treatment was not statistically (P>0.05) different from untreated.

xxvi. t-Test: Paired Two Sample for Means: Strawberry Leaf petiole °Brix Untreated and Silicon twice + no fungicides (19 September 2017)

	Untreated	Silicon x 2 + no fungicides
Mean	8.52	9.42
Variance	1.832888889	3.806222222
Observations	10	10
Pearson Correlation	0.26737835	
Hypothesized Mean Difference	0	
df	9	
t Stat	-1.384342366	
P(T<=t) one-tail	0.099805679	
t Critical one-tail	1.833112933	
P(T<=t) two-tail	0.199611358	
t Critical two-tail	2.262157163	

Table (xxvi) showed that in the average values of leaf petioles °Brix levels sampled from 19 September 2017 there was no statistical difference (P>0.05) between untreated and silicon twice + no fungicides treatment. Conclusions from table (xxiv, xxv, xxvi) showed that silicon did not raise °Brix levels of the leaf petioles sampled 19 September 2017.

Appendix 66 Chapter 5 - Results workings - Means number of leaves from the hydroponic experiment 2018

Table (i) Test for normality and (ii) Paired Samples Test – Mean leaf number per plant (Untreated and silicon-treated)

i. Normality Test

HO: Difference between untreated and silicon-treated follows normal distribution

HA: Difference between untreated and silicon-treated does not follows normal distribution

Tests of Normality Kolmogorov-Smirnov^a Shapiro-Wilk Statistic df Sig. Statistic df Sig. Difference .179 23 .054 .925 23 .0

a. Lilliefors Significance Correction

From table (i) If the sig value is less than critical value 0.05, reject Ho. In the analysis in figure 5.14a the sig value 0.084 is greater than critical value 0.05, accept Ho and conclude that the difference between untreated leaves and silicon-treated leaves follows a normal distribution. So the parametric test was used to test for significance difference between both treatment (Table ii).

.084

ii. Null and Alternate Hypothesis

Ho: There is no statistically significant difference between untreated and silicon-treated.

Ha: There is statistically significance difference between untreated and silicon-treated.

Parametric Test: (Paired Samples Test)

			Paired Differences						
				Std. Error	95% Confidenc Differ	e Interval of the rence			
		Mean	Std. Deviation	Mean	Lower	Upper	t	df	Sig. (2-tailed)
Pair 1	Untreated - Silicon Treated	-2.30435	1.98711	.41434	-3.16364	-1.44506	-5.561	22	.000

Paired Samples Test

From table (ii) If the sig value is less than critical value 0.05, reject Ho. The Paired Samples test conducted between average values of number of leaves per plant sig value.000 is less than critical value 0.05 (0.00 < 0.05) reject Ho and conclude that there is a statistical significance difference between untreated leaves and silicon-treated leaves. This showed that there is a significant increase in the average values of number of leaves from the silicon-treated plants compared with untreated in the hydroponics over time

Appendix 67 Chapter 5 - Results workings - Means number of runners from the hydroponic experiment 2018

Table (iii) Test for normality and (iv) non-parametric test; Wilcoxon Signed Ranked Test – Mean runner number per plant (untreated and silicon-treated)

iii. Normality Test

Null and alternate Hypothesis:

Ho: Data follows normal distribution.

Ha: Data does not follow normal distribution.

Tests of Normality

	Kolm	ogorov-Smir	nov ^a	Shapiro-Wilk				
	Statistic	df	Statistic	df	Sig.			
Difference	.437	23	.000	.582	23	.000		

a. Lilliefors Significance Correction

In the analysis (Table iii) the sig value of difference represent paired difference also does not follow normal distribution. Therefore, to test for significance difference the non-parametric test; Wilcoxon Sign Ranked Test is used in table iv to test the differences between untreated and silicon-treated plants.

iv. Non-Parametric Test

Wilcoxon Sign Ranked Test

Null and Alternate Hypothesis

Ho: There is no statistically significant difference between untreated and silicon-treated.

Ha: There is statistically significance difference between untreated and silicon-treated.

Test Statistics^a

	Silicon_Treat ed - Untreated
Z	-4.000 ^b
Asymp. Sig. (2-tailed)	.000

a. Wilcoxon Signed Ranks Test

b. Based on negative ranks.

The Wilcoxon Sign Ranked Test conducted between silicon-treated runners and untreated runners showed that the sig value.000 is less than the critical value 0.05 (0.00 <0.05) which concludes that the differences between untreated runners and silicon-treated runners is statistically different. Silicon-treated plants had more runners compared to untreated (P<0.05).

Appendix 68 Mean numbers of fruits per plant tubs counted at two sample dates

Sample date	Untreated	Silicon-treated
13/06/2018	2	5
13/06/2018	1	3
13/06/2018	3	2
13/06/2018	1	4
Fruits were not produced in	From the 13 June 2018,	From the 13 June 2018,
all plants from the	fruits were sampled from	fruits were sampled from
experiment on the	only 4 plant tubs.	only 4 plant tubs
13/06/2018		
From the 20 June fruits	From the 20 June 2018,	From the 20 June 2018,
were sampled from all 5	fruits were sampled from	fruits were sampled from all
plants tubs	all 5 plant tubs	5 plant tubs
20/06/2018	2	5
20/06/2018	1	3
20/06/2018	2	3
20/06/2018	1	4
20/06/2018	2	3

Appendix 69 Chapter 5 - Results workings - Mean number of strawberry fruits from the hydroponic experiment 2018

Table (vi) Test for normality and (vii) Non-parametric test; Wilcoxon Signed Ranked Test – Mean number of fruits per plant (Untreated and silicon-treated)

vi. Normality test

Null and alternate Hypothesis

HO: Difference between untreated and silicon-treated follows normal distribution

HA: Difference between untreated and silicon-treated does not follows normal distribution

	Kolm	ogorov-Smir	nov ^a	Shapiro-Wilk				
	Statistic	df	Sig.	Statistic	df	Sig.		
Difference	.237	9	.156	.820	9	.035		

Tests of Normality

a. Lilliefors Significance Correction

According to Shapiro Wilk Test in table (vi) the sig value.035 is less than critical value 0.05, reject Ho and conclude that difference between the average values of number of untreated fruits and silicon-treated fruits does not follows normal distribution. The non-parametric test; Wilcoxon Sign Ranked Test was used to find the differences between the two treatment variables (vii).

vii. Non-Parametric Test

Wilcoxon Sign Ranked Test

Null and Alternate Hypothesis

Ho: There is no statistically significant difference between Untreated and silicon-treated.

Ha: There is statistically significance difference between untreated and silicon-treated.

	Silicon_Treat ed - Untreated
Z	-2.461 ^b
Asymp. Sig. (2-tailed)	.014

Test Statistics^a

a. Wilcoxon Signed Ranks Test

b. Based on negative ranks.

From table (vii) the test showed that the sig value.014 is less than critical value of 0.05, reject Ho and conclude that there is a statistical significant difference (P<0.05) between the average values of number of silicon-treated fruits compared with the number of untreated fruits.

°Brix per plant	Untreated	Silicon-treated
13/06/2018	9	14
	10	14
	17	20
	10	16
	16	13
20/06/2018	16	12
	10	15
	16	19
	15	13
	16	14

Appendix 70 Strawberry fruit °Brix from 5 individual strawberries

Appendix 71 Mean of 10 leaves, per plant (chlorophyll content) from untreated and silicon-treated plants

Sample date	Untreated	Silicon-treated
06/03/2018	33.9	30.9
14/03/2018	21.0	51.8
20/03/2018	32.5	97.3
27/03/2018	40.0	88.4
04/04/2018	40.5	55.9
11/04/2018	43.2	11.2
17/04/2018	20.0	13.2
25/04/2018	49.5	44.5
02/05/2018	55.7	53.2
09/05/2018	33.0	15.9
15/05/2018	28.9	41.3
22/05/2018	23.8	55.4
30/05/2018	33.5	57.7
06/06/2018	40.0	55.4
13/06/2018	36.4	57.3
20/06/2018	39.0	43.9
27/06/2018	22.3	46.5

Appendix 72 Chapter 5 - Results workings - Means number of chlorophyll contents of the leaves from the hydroponic experiment 2018

Table (x) Test for normality and (xi) Paired Samples Test – Mean chlorophyll content per plant (Untreated and silicon-treated)

x. Normality Test

HO: Difference between untreated and silicon-treated follows normal distribution

HA: Difference between untreated and silicon-treated does not follows normal distribution

Kolmogorov-Smirnov ^a Shapiro-Wilk Statistic df Sig. Statistic df Sig. Difference 084 23 200 [*] 976 23 825	,												
Statistic df Sig. Statistic df Sig.		Kolm	ogorov-Smir	Shapiro-Wilk									
Difference 084 23 200 [°] 976 23 825		Statistic	df	Statistic	df	Sig.							
Difference .004 23 .200 .370 23 .023	Difference	Difference .084 23		.200	.976	23	.825						

Tests of Normality

*. This is a lower bound of the true significance.

a Lilliefors Significance Correction

In the Shapiro wilk test (Table x) done between average values of number of silicontreated and untreated chlorophyll content of strawberry leaves, the sig value.825 is greater than the critical value 0.05, accept Ho and conclude that the difference between untreated chlorophyll content and silicon-treated chlorophyll content follows a normal distribution. So the parametric test (Table xi) was used to test for significance difference.

xi. Null and Alternate Hypothesis

Ho: There is no statistically significant difference between Untreated and silicontreated.

Ha: There is statistically significance difference between untreated and silicontreated.

	Paired Samples Test													
				Std. Error	95% Confidence Interval of the Difference									
		Mean	Std. Deviation	Mean	Lower	Upper	t	df	Sig. (2-tailed)					
Pair 1	Untreated - Silicon_Treated	-8.24348	25.16054	5.24633	-19.12371	2.63675	-1.571	22	.130					

The Paired Sample Test shown in table (xi) assessed the difference between average values of number of chlorophyll content of untreated and silicon-treated. Since the sig value.130 is greater than critical value 0.05, accept Ho and conclude that there is no statistical significant (P>0.05) difference between untreated and silicon-treated chlorophyll contents in the leaves.

Appendix 73 fresh weights biomass (No fruits were included) of 10 individual whole strawberry plants (Untreated and silicon-treated)

The table (xii) is drawn from individual fresh weights of strawberry plants from the hydroponics at the end of the experiment.

(xii)	Untreated	Silicon-treated plants
	plants (g)	(g)
	120	148
	151	155
	128	190
	134	164
	160	189
	138	178
	173	168
	147	177
	140	196
	156	129

Untreated (average) = 144g (approx.) and silicon-treated plants (average) = 169g (approx.)

Appendix 74 COSHH FORMS (Field experiments)

Substance(s) information											
Description Activity	of	 Maltmas farm (Wisbech) – Silicon nutrients (Sirius) application a) fertigation system b) Spray bottles (Hand sprayers) 									
Risk Level for	this I	^s LOW									
Activity Location of Activity Maltmas farm, Wisbech, Friday Bridge, Cambridgeshire, UK											
Product Name	Con n	mpositio	Quantity	-	Concent on	Concentrati on CAS Numb			ber UN Number		
Strawberry Leaves	N	/A	160 - 200 le sample date	eaves per e	-		-		-		
Silicon nutrient (Sirius) "Si(OH)4"	N	/A	3mls (2.5 silicon nut 1.25L spray	mls) of rients in y bottles	0.25%		-		-		
Silicon nutrients (Sirius) "Si(OH)4"	N	/A	Silicon twice wee fertigation	nutrients ekly via system	0.017%		-		-		
Schedule 5 CTSA		Chemica Weapon Convent	ıl s ion?		Drug Pr	ecursors?					
Hazard Classifica Please complete appendix i	tion (tic ^C using mor	ck as appropria te than 3 chemi	ite) icals or dealing wit	th any flamma	ble, oxidising of	r explosiv	ve substance				
	ext	remely fla	mmable	RANNAGE			Highly	/ flammal	ole		
۱	Fla	mmable					Explos	sive			
	Dar env	ngerous vironment	to t	he CORROSIV			Corros	sive			
	Act	utely Toxi	c	TOXIC			Toxic				
	Irri	tant		HARMFU			Harmf	ful			
★	Ser car etc.	sitising, cinogenic,	mutagen allergen	ic,	٩		Oxidis	sing			
😼 📀	Bio	hazard					Radio	active			

		>	Extremely	,		Low					Oth	er			
<u>*</u>	$\mathbf{\nabla}$		temperatu	re											
Route(s	ntr	y (tick as appropriate													
Skin			Inhalation	✓	Eye Co	ontact	✓	Ing	gestion		1	Other			
Contact			(-)									(specify)			
Use of S	Substa	ance	e(s)			0:1:		4	ta (Cinia			4h a famma ar			
How should the substance be used?							n nu Ha Ap	ind soplie	sprayed ed throug	by stud	d on lents fertig	or research ation syste	ners mers	y grov	vers.
Who wi	ill be e	expo	sed to the subs	tanc	e?	Stude	nts,	rese	archers a	and gro	owers	5			
How much is used during the activity?					/?	<u>Silico</u> - -	n nu Sil Sil C.	icon icon	e nts (Sir n nutrien n sprays	<u>ius) a</u> ts (Sir (Sirius	oplic: ius) ii s) - 1.	ation n fertigatio 25L	on sy	stem ((N/A)
Duration exposed to the substance(s)?					?	 <u>Silicon nutrients (Sirius) application</u> Silicon nutrients in fertigation system – 30 minutes Silicon sprays (Sirius) - 30 minutes – 1 hour D. 									
What an	re the	cons	sequences of ex	pos	ıre?	See page 3-4									
Does th certain	e subs individ	stand	ce pose additio s <i>(e.g. expectar</i>	nal 1 11 ma	risks to others)	Immunocompromised Individuals or individuals with predisposing factors; e.g. pregnant, long-term antibacterial antibiotics treatment, organ transplant patients, etc. from micro-organisms and chemicals.									
Will the substance(s) or composition change during the activity and have these changes been considered during this assessment?					osition e these g this	No.									
Contro	l Mea	sure	es												
Does the Substance(s) require authorisation does age pose additional risks) Can a less hazardous substance be used to c contact your supplier for additional information What controls are required for this substance ventilated areas, not in spray/mist form loco						to retai o the s tion) [u e, othen	n? (i ame <i>use th</i> than ust ve	.e. i job <i>he h</i> n Pe enti	s it used ? (if you ierarchy ersonal P lation, a	very i do not of con Protecti uthoris	nfreq t know trols, we E sed pe	uently or w, please) quipment (ersons only	Yes Yes (PPE)? (E.	No ✓ No ダ g. well
Authori chemica quantiti	ised p als; re les of l	erso plac eave	ons only (i.e. ce chemicals be es out on bench	resea back	archer a into ap	and lab	oorat ate s	ory	technicinge once	ians. I e comj	Do no	ot use exc l work. D	cess on't	volur leave	nes of e large

Is any H	Personal Protective Equipr	ce?		Yes ✓	No				
	Hearing protection (State type required)	Gloves (State type required)				Overalls/clothing (State type equired)			
	N/A		Latex/ Nitrile Gloves	Laboratory Coat					
	Mask/respirator (State type required)		Other (State type required)		Eye Pro	Protection (State type d)			
	N/A		No Open Shoes		Protective glasses, when handling chemicals				
Are a contra i	ny PPE combinations ndicated?	N/A							
Other]	Precautions and Emerge	ncy Proc	edures						
Handlir Transpo	ng & Storage & ort	 All chemicals listed are segregated and stored away from combustible materials in the laboratory. Use correct PPE, when handling chemical substances. Carefully carry chemicals, with lids on, don't carry more than can acfely menage. 							
Have p informa	ersons undertaking this an ation and training in its us nicated to all involved with	ctivity and using this substance(s) been provided with e? (Ensure that the findings in this risk assessment are th this activity and that they are aware of the required							
control.	s).	Sirius (Silicon nutrient):						
		In contact with eyes: Rinse immediately with plenty of water for 15 minutes holding the eyelids open. Seek medical attention.							
First Aid Measures		• <u>If inhaled</u> : Move the exposed person to fresh air. Seek medical attention.							
		• <u>If ingested</u> : Do not induce vomiting. Rinse mouth thoroughly. Drink 1 to 2 glasses of water.							
		• <u>Signs and symptoms of exposure:</u> Irritating eyes, respiratory tract, nausea. If irritation or symptoms persist, seek medical advice.							
Fire Fig	ghting Measures	 <u>Sirius (Silicon nutrient)</u> - Product is not flammable. Use appropriate media for adjacent fire. Use flooding quantities of water to cool containers, keep away from common metals. 							

Emergency procedure in case of fire: Sound alarm, evacuate, leave all									
belongings, taking nearest exit, don't return to the building un									
	instructed to do so.								
Accidental Release Measures	material. T ughly with	ransfer plenty							
Liquids: Sirius (Silicon nutrient) – dispose of as special waste in compliance with local and national regulations. Empty containers can be cleaned with water or deionized water and sent for disposal or recycling.									
		Yes	No						
Health Surveillance required?	Health Surveillance required?								
Workplace Exposure Limits Re	ef: EH40								
Is/are the substance(s) used subject to the referenced in the latest edition of the HSE's Yes No									
EH40 document? (If yes, then please refer to the UH Workplace Exposure Limits Guidance and complete Appendix)									
DSEAR Risk Assessment			1						
Is/are the substance(s) used s	Yes	No							
Atmospheres Regulations? (If yes, then please complete App		~							
Assessment of Risk									
Are all the controls detailed curre	Yes	No							
	✓								
If these controls are not in plac (<i>Please note – COSHH substance</i>)	e or additional controls are required, please state acti es must NOT be used if adequate controls measures are r	on to be t 10t in plac	aken. <i>e)</i> .						
Remedial Actions Required	Date	for							
	completi	on							
N/A	N/A								
Are all Hazards to health adequat	Yes	No							
	-	✓							
Are there one herends not sale	ing to substances that wood to be someidened for this	notivit-2							
Are there any hazards not relating to substances that need to be considered for this activity?									

Hazard	What harm	Who	Controls
Use of Sharps (Scalpel, scissors)	Cuts, self-inoculation	Researche r, Technical Staff	Use with care, don't run in the laboratory
Use of Pipettes	Repetitive movement, self-inoculation	Researche r, Technical Staff	Use with care, if using for long period of time, take regular breaks.
Use of Glassware (inc. slides)	If broken, cuts, self-inoculation	Researche r, Technical Staff	Handle glass with care and do not drop. If broken glass, dispose of in appropriate sharps container.
Sitting at bench or cabinet to use microscope	Physical/ ergonomical- sitting a long time	Researche rs	Take regular breaks to move and stretch
Electrical Laboratory equipment	Electric shock and burns from contact with live parts	Researche rs, Other members of the lab	Ensure the equipment is not damaged before use and is full plugged in. Ensure socket is turned off before inserting plug. Electrical equipment should be regularly PAT tested.
Slips/trips/falls	Injury from falling over	Researche rs, all members of the laboratory	Do not have loose wires and cables on the floor. Immediately clean up any spillages on the ground. Be careful on uneven ground.
Fire	Burns	Researche rs, all members of the laboratory	Follow local procedures.
Allergy to the spores	to the Allergic Reaction		Don't have lots of leaves open on the workbench. Clean bench after assessment. In event of allergic reaction, seek local first aider.
Use of Spectrophotome ter	UV	Researche rs/ Laborator y	Do not use with lid open

		technician	
		s	
Temporary blinding due to levels of illumination being set too high	Damage to eyes	Researche rs	Reduce light intensity immediately. Take regular breaks from using the microscope.
Eye strain and long-term ocular problems	Damage to eyes	Researche r	Take regular breaks from using the microscope.
Falling objects- microscope falling of bench	Physical- heavy object falling on person	Researche r, other members of the laboratory	Ensure microscope is away from the edge of the bench
Physical injury to head and eyes being pushed into microscope	Physical	Researche r	Be aware of positioning of microscope to not hit self against it

Appendix 75 COSHH forms (Laboratory and Glasshouse experiments)

Substance(s) information												
Description of Activity			 Laboratory experiment: Silicon extraction from strawberry leaves from a silicon fertigation field trial. Silicon localization in strawberry plants (Strawberry leaf, petiole and root cross- section for fluorescence microscopy) Glasshouse experiment: Silicon deposition in strawberry plants (Cross-sections for confocal microscopy) 									
Risk Level for this Activity			Low									
Location of Activity			 Laboratory experiments: ➢ New science building (Room 2J013) College lane, University of Hertfordshire. Glasshouse experiments: Impus, Lower Hatfield road, Hertford mpus 									
Product Name Comp			position	ition Quantity		Concentration		CAS Number		UN Num ber		
Strawbe leaves	erry	N/A	A	Approx. 100 per sample		N/A	N/A N		N/A		N/A	
Chemicals listed N/A			A	Quantity listed		N/A		N	N/A		N/A	
Schedule 5 CTSA?			Explosive regulations 2014?			Chemi Conve	ical Weap ntion?	ons		Drug Precursor	s?	
			extremely flammable Please complete appendix 2			r abietie			Highl Pleas apper	y flamm e co ndix 2	able omplete	
8	۲	~	Flammable Please complete appendix 2							Explosive Please complete appendix 2		
*	×		Dangerous to the environment			CORROSIV]	Corro	osive		
VERY			Acutely To:	xic		TOXIC		(Toxic	;		
IRRITANT		~	Irritant			HARMFUI	()]	Harm	ful		

	Irritant			HARMFU	()		l Harm	ful			
★ ◆	Sensitising carcinogen	c, aller	mutagenic, genic, etc.				Oxidi Please appen	sing e comj dix 2	plete		
😼 🗘	Biohazard						Radio	active			
	Extremely	Low ter	mperature				Other	Other			
Route(s) of Entry (tick as appropriate)											
Skin Contact ✓ Ir n	halatio	✓	Eye Contact	~	Ingest	ion	✓ Other (specify)				
Use of Substance(s)											
How should the substance		 Cher pipet Cher pipet 	nicals ted and nicals ted and	used fo l titrated used fo l stained	or silicon 1 r silicon 1	extract localiza	ions are to	o be d be			
Who will be exposed to the	e substance?		Researchers	and Lal	ooratory	v technici	ans				
How much is used during	Laboratory experiment - Silicon extraction: > Dry strawberry leaves sample – 100mg (0.1mg) > Sirius (Polyether modified polysioxane, Ethanol tetraethyl silicate and Alkyloxypoly ethyleneoxy ethanol) – Less than 100ml per treatment > Nitric Acid – 1ml > Mounting fluid + Glycerine +lactic acid – 0.0% > H202-2mls > NaOH – 2mls > Boric acid – 0.075mls > Ammonium molybdate – 0.25mls > Tartaric acid – 0.25mls > Reducing solutions (1-amino-2-mapthol-4 sulfonic acid + NaHSO ₃ + Deionized water) – 0.25mls Laboratory experiment – Silicon deposition: > Vectashield mounting medium (for fluorescence) – in drops per slide (One drop per slide) > Lyso tracker Yellow HCK-123 (Fluorescence dye) - 1µM > De-ionized water – 9mls Glasshouse experiment: > Sirius- 0.017% and 0.25%										
		 Bug clear (for fruit and rapeseed oil – Maximu per crop Chicken poo (Manure) crop Tomorite (Liquid fer extracts – 1 Litre De-ionized water 	l vegetab ım numb – Less t rtilizer)	er of han a conta	ontair treatn hand	ns 17g/L nent – 3 full per seaweed					
---	--	---	---	-------------------------	--------------------------	---					
Silicon extraction (Laboratory): Leaves in incubator - 2 - 3 days Autoclave - 1 hour Titration - 2 hours Silicon deposition (Laboratory): Staining (cross-sections) - 2 hours											
What are the consequences of exposure?	See	page 6-7									
Does the substance pose additional risks certain individuals (e.g. expectant mother)	une system compromised p -term antibacterial anti splant patients etc. n microbiological and some	eople, e., biotics e chemica	g. pres treati	gnant ment,	woman, organ						
Will the substance(s) or composition change during the activity and have these changes been considered during this assessment?											
Control Measures											
Does the Substance(s) require authorisation to retain? (i.e. is it used very or does age pose additional risks)			quently	Yes		No ✓					
Can a less hazardous substance be used to	do the sa	me job? (if you do not know	, please	Yes		No					
Contact your supplier for additional informed what controls are required for this substa	mation) [i	ise the hierarchy of control.	s) Fauinmer	nt (PP	E)2 (P	✓ Eg well					
ventilated areas, not in spray/mist form, local exhaust ventilation, authorised persons only)											
Authorised persons only (i.e the researcher and laboratory technicians)											
Is any Personal Protective Equipment required, when using this substance?			No								
Hearing protection (State type required) Gloves (State type required) Ove hing		Overa hing type re	lls/clot (State equired)								
N/A	Latex gloves		La	b coat							

	Mask/respirator (State type required)	0	Other (State type required)		Eye Prote (State requi	ction ? type red)
	N/A		Cover shoes or safety shoes		Ey goggi	e les
Are any indicate	PPE combinations contra d?	N/A				
Other P	recautions and Emergency	Procedur	es			
Handling & Storage & Transport All chemicals listed are segregated and stored away from combu materials in the laboratory.			nbustible			
Have pe	rsons undertaking this activi	ty and usin	g this substance(s) been provid	led with	Yes	No
commun controls	<i>vicated to all involved with th</i>).	is activity	and that they are aware of the r	equired	~	
First Aid	1 Measures	J In free If atte If Dr Si tra add K L In Sy da plo In Ex ind Sy da plo In Ex ind Sy da Plo If Sy da Plo If If If If If If If If If If	 SIRUS: In contact with ey plenty of water for 15minut Seek medical attention. contact with skin: If inhaled: Tesh air. Seek medical attention. inhaled: Move the exposed persention. ingested: Do not induce vomitier ink 1 to 2 glasses of water. gns and symptoms of exposure of exposed in the exposed persect. NITRIC ACID, TARTARIC contact with eyes: Contact rappendicts include eye burns, mage to cornea may result. In centy of water and seek medical exposed tissues. Harmonediately flush with plenty of the exposed tissues. Harmonediately flush with plenty of exposed tissues. Harmonediately flush with plenty of exposed tissue of exposed tiss	es: Rinse es holdin Move the son to fres ng. Rinse e: Irritatin ptoms po CACID, I pidly cau watering ase of eye attention rapid corr ration of ness, infi ful if abs of water a othing an	 immediating the eyeliang the eyelia	ely with ds open. >erson to = medical oroughly. spiratory medical 2ID: damage. ermanent inse with ely. a contact. ymptoms a and/or ugh skin. minutes ing soap.

• <u>If ingested</u> : Do not induce vomiting. Severe and rapid corrosive burns of the mouth, gullet and gastrointestinal tract will result if swallowed. Wash out mouth with water and give a glass of water or milk. Get medical attention immediately.
• Signs and symptoms of exposure: burning, coking, nausea, vomiting, severe pain, coughing, wheezing, laryngitis, shortness of breath and headache.
 Is hospitalization required? Yes. For Nitric acid in eyes. If ingested: Inhalation of mists can cause corrosive action on mucous membranes. Move casualty to fresh air and keep at rest. May be fatal if inhaled, may cause delayed pulmonary edema. Get medical attention.
M. TRYPAN BLUE:In contact with eyes: flushes eyes with water as a precaution.
• In contact with skin: wash off with soap and plenty of water. Consult with a physician.
• If inhaled: IF breathed in, move person into fresh air. If not breathing, give artificial respiration. Consult a physician.
If ingested: Do not give anything to drink. Rinse out mouth with water and seek urgent medical A.D.
Signs and symptoms of exposure: Not thoroughly investigated (according to Sigma MSDS).
Persons at special risk. All persons at potential risk.
 N. BUG CLEAR (Protection during/in use Keep off skin, keep away from eyes, wash off splashes immediately. Do not breathe spray. Wash hands and exposed skin after use IF SWALLOWED: Call a POISON CENTRE or doctor/physician if you feel unwell. O. Rinse mouth; If medical advice is needed, have product container or label at hand. Dispose of content/ content to a household waste recycling centre as hazardous waste except for empty containers, which

	can be disposed of by recycling - Contact your local council		
	ior details.		
	TOMORITE (Liquid fertilizer)		
	• If inhaled; unlikely to cause a problem		
	• If in contact with skin; wash off with water		
	• If in contact with eyes; irrigate with water, seek medical		
	advice if irritation persists.		
	 If ingested, Obtain medical advice 		
	Types of fire extinguisher to be used: use water spray alcohol-resistant foam, dry chemical or carbon dioxide.		
	Nitric acid, tartaric acid and boric acid- Flammable. Use extinguishing media appropriate to the surrounding fire conditions;		
Fire Fighting Measures	Sirius - Product is not flammable. Use appropriate media for adjacent fire. Use flooding quantities of water to cool containers, keep away from common metals.		
	Emergency procedure in case of fire: Wear self-contained, approved breathing apparatus and full protective clothing, including eye protection and boots. Material can react violently with water (splattering and misting) and react with metals to produce flammable hydrogen gas.		
	Solid spill: Trypan blue- without raising any dust, mop up spill with tissue paper		
	Liquid Spill: Sirius – Absorb with inert, absorbent material. Transfer to suitable, labelled container. Clean spillage area thoroughly with plenty of water.		
Accidental Release Measures	Nitric acid- Cautiously add water to spill, taking care to avoid splashing and splattering. Neutralize diluted spill with soda ash or lime. Absorb neutralized spill with vermiculite or other inert absorbent material, then place in a suitable container for disposal. Clean surfaces thoroughly with water to remove residual contamination		
	Trypan blue- absorb spill into tissue.		
Disposal	Solids: Strawberry leaves – Segregate and collect into sealed bin bags for plants disposal.		

 Liquids: Sirius – dispose of as special waste in compliance with local and national regulations. Empty containers can be cleaned with water or deionized water and sent for disposal or recycling. Nitric acid, Tartaric acid, Boric acid- Carefully dilute with water, neutralize spill. Neutralized material may be flushed to sewer or disposed of through a licensed contractor. Users should review their operations in terms of the applicable federal/nation or local regulators and consult with appropriate regulatory agencies before discharging or disposing of waste materials. Containers, if thoroughly cleaned, preferably by rinsing three times and handling the rinse water as waste residues, may be disposed of or recycled as non-hazardous waste. Users should review their operations and consult with appropriate regulations and consult with appropriate regulatory agencies before discharging or local regulations and consult with appropriate regulatory agencies before discharging or disposing at waste material. Trypan Blue- Put in a hazardous waste bag and send out via yellow bin disposal system Solvents: Biohazard: Autoclaves microbiological wastes. As long as they do not 					
	Does this substance(s) require any user health	Yes	No		
Health Surveillance required?	surveillance?, if yes, please state, which type & level, and, who it is conducted by:		~		
Workplace Exposure Limits Ref: EH40					
Is/are the substance(s) used subject to the referenced in the latest edition of the HSE's		Yes	No		
EH40 document? (If yes, then please see workplace exposure limits guidance note)			~		
DSEAR Risk Assessment					
Is/are the substance(s) used subject to the Dangerous Substances & Explosive		Yes	No		
Atmospheres Regulations?			~		
Assessment of Risk					
Are all the controls detailed currently in place?		Yes	No		
Are an the controls detailed currently in place?		\checkmark			
If these controls are not in place or additional controls are required, please state action to be taken. (Please note – COSHH substances must NOT be used if adequate controls measures are not in place).					
Remedial Actions Required	Date for com	pletion			
Please state any further controls that have not been outlined in the assessment that may be required?					

				✓	
Are there any hazards not relating to substances that need to be considered for this activity?					
Hazard	What harm	Who	Controls		
Sirius (Polyether modified polysioxane, Ethanol tetraethyl silicate and Alkyloxypoly ethyleneoxy ethanol)	Is an irritant and may cause damage eyes if in contact.	Researchers	Q. R. S. <u>In contact</u> immediate water for 1 eyelids ope T. Seek medie	t with eyes: ly with plen 5minutes hold: en. cal attention.	Rinse ty of ing the
Nitric acid, Boric acid, Ammonium molybdate, Tartaric acid and reducing solutions (1-amino-2- mapthol-4 sulfonic acid)	Contact with combustible material may cause fire and severe burns. <u>In contact with eyes</u> Contact rapidly causes severe damage. Symptoms include eye burns, watering eyes. Permanent damage to cornea may result. <u>In contact with skin:</u> Severe and rapid corrosion from contact. Extent of damage depends on duration of contact. Symptoms include burning, itching, redness, inflammation and/or swelling of exposed tissues. Harmful if absorbed through skin.	Researchers	In contact eye contact water and s immediate In contact t Immediate water at le removing c and wash medical att	with eyes: In o t, rinse with plo seek medical at y. with skin: ly flush with plo ast 15 minutes contaminated cl using soap ention.	case of enty of tention enty of while lothing . Get
solutions Reducing	Irritating eyes, respiratory tract, nausea. If irritation or symptoms persist, seek medical advice	Researcher	If inhaled person to fi attention.	Move the extension of the matrix of the matr	xposed redical

mapthol-4 sulfonic acid + NaHSO3)			If ingested: Do not induce vomiting. Rinse mouth thoroughly. Drink 1 to 2 glasses of water.
Bug clear	Is an irritant and may cause damage eyes if in contact.	Researcher	In contact with eyes: Rinse immediately with plenty of water for 15minutes holding the eyelids open.
Vectashield mounting medium (for fluorescence)	This product has been classified as non- hazardous based on the physical and/or chemical nature and/or concentration of ingredients. (Non-hazardous substance or mixture)	Researcher	<u>Inhalation</u> : Remove to fresh air. <u>Skin contact</u> : Wash thoroughly with soap and water. <u>Eye contact</u> : Flush eyes with water as a precaution. <u>Ingestion</u> : Do not induce vomiting. Wash out mouth with water.
TOMORITE - (Liquid fertilizer)	This product has been classified as non- hazardous based on the physical and/or chemical nature and/or concentration of ingredients. (Non-hazardous substance or mixture)	Researcher	If inhaled; unlikely to cause a problemIf in contact with skin; wash off with waterIf in contact with eyes; irrigate with water, seek medical advice if irritation persists.If ingested, Obtain medical advice
Trypan blue	Irritating eyes, respiratory tract, nausea. If irritation or symptoms persist, seek medical advice	Researcher	If inhaled: Move the exposed person to fresh air. Seek medical attention.If ingested: Do not induce vomiting. Rinse mouth thoroughly. Drink 1 to 2 glasses of water.

Appendix 76 List of posters

Poster I



Poster II



Poster III





Authors: Asiana ,I ., Hall A.M., Davies K. University of Hertfordshire, College Lane, Hatfield AL10 9AB, UK

Introduction

۵ The most important disease of protected strawberries in the UK is strawberry powdery mildew caused by Podosphaera aphanis, which has to be controlled by the frequent use of fungicides. (Dodgson, Hall & Jin 2016).

All plants contain silicon but work carried ÷ out at the University of Hertfordshire has shown that the weekly use of a silicon nutrient in the fertigation tubes at a commercial strawberry farm results in reduced susceptibility to this disease. Silicon can only be taken up in a bioavailable form and the nutrient used in this experiment is a bioavailable form of silicon

- Aim
- * To examine the effect of a silicon nutrient applied through the roots of strawberry plants in reducing strawberry powdery mildew. Materials and methods The silicon fertigation field trial had 6 treatments and samples were collected every fortnightly for disease assessment. For treatments see table 2.

The silicon localization experiment in the glasshouse had 12 treated and 12 untreated strawberry plants in a glasshouse. 0.017% silicon nutrient (as in the fertigation field trial) was delivered for 8 weeks into two ways;

- a). Through the root application

 b). Hydroponically
 Cross-sections of strawberry leaves, petioles and roots were stained with a fluorescence dye (Basic amine Lyso tracker yellow HCK -123), final concentration 1μ M (Shetty et al.,2012).

*Examination of sections was conducted using a confocal microscope at x400 magnification and wavelength 450nm.

Results Figure 1 and Table 1 shows the largest epidemic took place in the untreated and the silicon nutrient reduced the epidemic even in the absence of the fungicide. The results in figures 2-7 showed that in the leaf, silicon was found in the cuticle, epidermis, palisade layer, stomata and vascular tissue. In the petiole, the silicon was found in the epidermis and xylem and in the roots, the silicon was found in the xylem only. The fluorescence intensity (Table 2) of the cross sections was quantified and this shows that the silicon was 5 times higher in the treated plants than the untreated. In addition, the silicon fertigation field trial has shown that plants with higher levels of silicon are less susceptible to the disease (Figure 1 and Table 1).



Poster IV



ARE STRAWBERRIES EVER DEFICIENT IN SILICON ?

Asiana, I., Hall A.M., Davies K.

School of Life and Medical Sciences University of Hertfordshire, College Lane, Hatfield AL10 9AB, UK

Introduction

The most important disease of protected strawberries in the UK is strawberry powdery mildew caused by *Podosphae*ra grown without silicon. *aphanis*, which has to be controlled by the **Material and methods**

Material and methods frequent use of fungicides (Dodgson,Hall & A hydroponic experiment was set up for 22 weeks (24 January – June 2018) in plastic tubs containing Hoagland's solution (Jones, 2016). All work was carried

Aim

Work carried out at the University of Hertfordshire has shown that the weekly use of silicon nutrient in the fertigation results in reduced susceptibility to this disease. Previous work done at the university has also shown that the use of silicon nutrient enhances the constitutive

university has also shown that the use of silicon nutrient enhances the constitutive defence mechanism of the plant to infection and additional crop benefits (Jin, 2015 & Jomas per tub. The nutrient used was "Sirius", a bioavailable form of silicon "Si(CPI)4" (Polyether modified Polysioxane, Ethanol, tetraethyl silicate and grown in coir which has little bioavailable silicon only. Additionally, they were topped up with Hoagland's solution twice weekly.

Results Results in table 1 found that there were significantly more leaves (P<0.05), significantly more runners (P<0.05), significantly more frunces (P<0.05), significantly more runners (P<0.05), significantly higher °Brix levels (P<0.05) and significantly more chlorophyll (P<0.05) in 'Sirius' treated strawberries compared with the untreated. See table 1.

University of Hertfordshire

Additionally, there was an increase in weight, size and biomass in the treated than the untreated plants (see table 1, figure 2 and 3). Flowering was a week earlier in the treated strawberries compared to the untreated. See table 1. (Data was analysed using Analysis of Variance (ANOVA), Regression statistics and the "Unpaired" t test respectively).

Results from statistics and table 1 show a summary of additional benefits of using silicon nutrient in growing strawberries

Table 1; Cumulative results from the hydroponic deficiency experiment January – June 2018

	Untreated	Treated
Strawberry plants	(no silicon nutrient) de-ionized water	(silicon nutrient) v/v 0.25%
Leaves	142 (Counted weekly for 22 weeks)	195 (Counted weekly for 22 weeks)
Runners	32 (Counted weekly for 22 weeks)	48 (Counted weekly for 22 weeks)
Flowering dates	22 May 2018 (First sight of flowering)	15 May 2018 (First sight of flowering)
Number of Fruits	15 (Counted during fruiting period)	32 (Counted during fruiting period)
°Brix	13 (Measurements taken after each sample)	15 (Measurements taken after each sample)
Weight of fruit (g)	12.9 (Fruits weighed after each sample)	20.6 (Fruits weighed after each sample)
Size (centimetres)	1.18cm (Fruit size was measured after sample)	1.38cm (Fruit size was measured after sample)
Chlorophyll (µmol/m²)	909 (Measured by SPAD once weekly)	1099.86 (Measured by SPAD once weekly)
Fresh weight biomass (g)	144 (At the end of experiment)	169 (At the end of experiment)





Acknowledgement Special thanks to Henry & Harriet Duncalle, Maltmas farm for the provision of the field tri Gidon Bahiri (OrionF) for providing Sirius for the silicon field trial, also my supervisors Avice M. Hall and Dr Keith Davies for their contributions towards my research journey. ors D

Avice M. Hall and Dr. Keith Davies for their contributions towards my research journey. **Beferences** Dogdson J, Hall A, Jin S. (2016). Control of strawberry powdery mildew under protection (project SF 62 & SF 62a). Fachsheet 17/02. spp. Stoneleigh, Warvickshire: AHDB horticulture Jin, X. (2015). *Epidemiology and control of powdery mildew (Podosphaera⁺ aphanis)* or "strawberry. Unpublished doctoral thesis. Hatfield: University of Hertfordshire. Jones Jr, J. B. (2016). *Hydroponics: a practical guide for the soilless grower.* CRC press. Liu B (2017). Sustainable strawberry production and management including control of strawberry powdery mildew. Unpublished doctoral thesis. Hatfield: University of Hertfordshire

Discussion and conclusion Whilst the results from the hydroponic deficiency experiment showed no deficiency symptoms, the leaves, runners, fruits and chlorophyll of silicon treated strawberries were significantly higher than the untreated. The concentration of silicon nutrient used significantly increased productivity.

The results suggests that though silicon is not an essential element, it is probably a limiting factor in strawberry productivity. It is therefore recommended that growers use silicon nutrient throughout the growing season, particularly when growing in coir.